TERAYON COMMUNICATION SYSTEMS Form DEFM14A May 24, 2007

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549 SCHEDULE 14A

Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934 (Amendment No.)

Filed by the Registrant p

Filed by a Party other than the Registrant o

- Check the appropriate box:
- <sup>o</sup> Preliminary Proxy Statement
- Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))
- b Definitive Proxy Statement
- Definitive Additional Materials
- o Soliciting Material Pursuant to §240.14a-12

Terayon Communication Systems, Inc.

(Name of Registrant as Specified In Its Charter)

#### N/A

(Name of Person(s) Filing Proxy Statement, if other than the Registrant)

Payment of Filing Fee (Check the appropriate box):

• No fee required.

- <sup>o</sup> Fee computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.
  - (1) Title of each class of securities to which transaction applies:
  - (2) Aggregate number of securities to which transaction applies:
  - (3) Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11 (set forth the amount on which the filing fee is calculated and state how it was determined):
  - (4) Proposed maximum aggregate value of transaction:
  - (5) Total fee paid:
- **b** Fee paid previously with preliminary materials.
- Check box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for which the offsetting fee was paid previously. Identify the previous filing by registration statement number, or the

Form or Schedule and the date of its filing.

- (1) Amount Previously Paid:
- (2) Form, Schedule or Registration Statement No.:

(3) Filing Party:

(4) Date Filed:

# MERGER PROPOSAL YOUR VOTE IS VERY IMPORTANT

May 24, 2007

Dear Stockholder:

You are cordially invited to attend the special meeting of stockholders of Terayon Communication Systems, Inc., to be held on June 28, 2007, at 11:00 a.m., local time, at our corporate headquarters located at 2450 Walsh Avenue, Santa Clara, California 95051.

At the special meeting, we will ask you to consider and vote upon (i) a proposal to adopt the merger agreement among our company, Motorola, Inc. and a wholly owned subsidiary of Motorola, and approve the merger of this subsidiary with us, and (ii) a proposal to approve the adjournment of the special meeting, if necessary, to solicit additional proxies if there are insufficient votes at the time of the special meeting to adopt the merger agreement and approve the merger. If the merger is completed, you will be entitled to receive \$1.80 in cash, without interest, for each share of our common stock that you own, unless you have properly exercised your appraisal rights.

Our board of directors has carefully reviewed and considered the terms and conditions of the proposed merger. Based on its review, the board of directors has determined that the merger agreement is fair, advisable to, and in the best interests of, our stockholders. ACCORDINGLY, OUR BOARD OF DIRECTORS HAS UNANIMOUSLY APPROVED AND DECLARED ADVISABLE THE MERGER AGREEMENT AND UNANIMOUSLY RECOMMENDS THAT YOU VOTE FOR THE ADOPTION OF THE MERGER AGREEMENT AND APPROVAL OF THE MERGER AND FOR THE ADJOURNMENT OF THE SPECIAL MEETING, IF NECESSARY, TO SOLICIT ADDITIONAL PROXIES.

Your vote is important. We cannot complete the merger unless the merger agreement is adopted and the merger is approved by the affirmative vote of the holders of a majority of our shares of common stock outstanding at the close of business on May 18, 2007, the record date for the special meeting. The proposal to adjourn the special meeting for the purpose of soliciting additional proxies, if necessary, will be approved if the votes cast in favor of the proposal by shares of common stock, present in person or represented by proxy and entitled to vote on the subject matter, exceed the votes cast against the proposal. The obligations of Terayon and Motorola to complete the merger are also subject to the satisfaction or waiver of certain conditions, including receiving clearance from regulatory agencies. Failure to submit a signed proxy or vote in person at the special meeting will have the same effect as a vote against the adoption of the merger and it will have no effect on the proposal to adjourn the special meeting for the purpose of soliciting additional proxies. Only stockholders who owned shares of our common stock at the close of business on May 18, 2007 will be entitled to vote at the special meeting.

**PLEASE COMPLETE, SIGN, DATE AND RETURN YOUR PROXY.** If you hold your shares in street name, you should instruct your broker how to vote in accordance with your voting instruction form.

If you sign, date and mail your proxy and do not indicate how you want to vote, your proxy will be voted FOR the adoption of the merger agreement and the approval of the merger and FOR the proposal to adjourn the special meeting, if necessary, to solicit additional proxies, provided that no proxy that is specifically marked AGAINST the proposal to adopt the merger agreement and approve the merger will be voted in favor of the adjournment proposal, unless it is specifically marked FOR the adjournment proposal.

This proxy statement explains the proposed merger and merger agreement and provides specific information concerning the special meeting. Please review this document carefully.

Sincerely,

Jerry D. Chase Chief Executive Officer

Neither the Securities and Exchange Commission nor any state securities regulatory agency has approved or disapproved the merger, passed upon the merits or fairness of the merger agreement or the transactions contemplated thereby, including the proposed merger, or passed upon the adequacy or accuracy of the information contained in this document. Any representation to the contrary is a criminal offense.

This proxy statement is dated May 24, 2007, and is first being mailed to stockholders of Terayon on or about May 24, 2007.

### NOTICE OF SPECIAL MEETING OF STOCKHOLDERS To be Held on June 28, 2007 11:00 a.m.

To the Stockholders of Terayon Communication Systems, Inc.:

Notice is hereby given that a special meeting of the stockholders of Terayon Communication Systems, Inc. will be held at our corporate headquarters located at 2450 Walsh Avenue, Santa Clara, California 95051 on June 28, 2007, at 11:00 a.m., local time, for the following purposes:

1. To consider and vote upon a proposal to adopt the merger agreement among Motorola, Inc., Motorola GTG Subsidiary VI Corp., a wholly owned subsidiary of Motorola, and us, and to approve the merger of Motorola GTG Subsidiary VI Corp. with and into us, in connection with which we will become a wholly owned subsidiary of Motorola, and each outstanding share of our common stock will be converted into the right to receive \$1.80 in cash, without interest;

2. To approve the adjournment of the special meeting, if necessary, to solicit additional proxies if there are insufficient votes at the time of the special meeting to adopt the merger agreement and approve the merger; and

3. To transact such other business that may properly come before the special meeting or any adjournments or postponements of the special meeting.

Holders of record of shares of our common stock at the close of business on May 18, 2007, the record date for the special meeting, are entitled to notice of, and to vote at, the special meeting and any adjournments or postponements of the special meeting.

# Our board of directors unanimously recommends that you vote FOR the adoption of the merger agreement and the approval of the merger and FOR the adjournment of the special meeting, if necessary, to solicit additional proxies.

We cannot complete the merger unless the merger agreement is adopted and the merger is approved by the affirmative vote of the holders of a majority of the shares of our common stock outstanding at the close of business on May 18, 2007.

Under Delaware law, holders of shares of our common stock who do not vote in favor of the adoption of the merger agreement and approval of the merger will have the right to seek appraisal of the fair value of their shares as determined by the Delaware Court of Chancery if the merger is completed, but only if they submit a written demand for an appraisal prior to the vote on the merger agreement and if they comply with the Delaware law procedures explained in the accompanying proxy statement. See The Merger Appraisal Rights beginning on page 36 and Annex C of the accompanying proxy statement, which contains the text of Section 262 of the Delaware General Corporation Law.

Whether or not you plan to attend the special meeting, please vote your shares as soon as possible. You can vote your shares prior to the special meeting (1) by mail with the enclosed proxy card, in accordance with the instructions on the proxy card, (2) by telephone by calling the toll-free number that appears on the enclosed proxy card and following the instructions given, or (3) by means of the Internet, by following the Internet voting instructions provided on the enclosed proxy card. Executed proxy cards with no instructions indicated thereon will be voted FOR the adoption of the merger agreement and approval of the merger and FOR the proposal to adjourn the special meeting to solicit

additional proxies, provided that no proxy that is specifically marked AGAINST the proposal to adopt the merger agreement and approve the merger will be voted in favor of the adjournment proposal, unless it is specifically marked

FOR the adjournment proposal. Even if you have returned your proxy, you may still vote in person if you attend the special meeting. Please note, however, that if your shares are held of record by a broker, bank or other nominee and you wish to vote at the meeting, you must obtain from the record holder a proxy issued in your name. If you fail to return your proxy or to vote in person at the special meeting, your shares will not be counted for purposes of determining whether a quorum is present at the special meeting, and will have the same effect as voting against the adoption of the merger agreement and approval of the merger and will have no effect on the proposal to adjourn the special meeting for the purpose of soliciting additional proxies.

# Please do not send any stock certificates at this time.

By order of the Board of Directors,

/s/ Zaki Rakib

Zaki Rakib Secretary

Santa Clara, California May 24, 2007

# TABLE OF CONTENTS

SUMMARY4Ibe Merger4Ibe Merger5Conditions to the Completion of the Merger5No Solicitation of Other Acquisition Proposals by Terayon7The Special Meeting8Recommendation of Our Board of Directors9Jinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Ireating Station of State Sector10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12CAUTONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS14Herayon Communication Systems. Inc.14Vatorola Inc.14Vatorola Inc.14Vatorola Inc.16Recommendation of our Directors and Executive Officers16Darpose of the Special Meeting16Record Date: States Entitled to Vote: Ouroum16Record Date: States Entitled to Vote: Ouroum16Record Date: States Entitled to Vote: Ouroum18Vote Required16Shares Owned by Our Directors and Executive Officers17Yoting of Proxies18Solicitation of Proxies18Solicitation of Proxies19States States19Directors and Executive Officers20Background of the Merger20Background of the Merger20Background of the Merger20Background of the Merger20 </th <th>SUMMARY4The Companies4The Companies5Conditions to the Completion of the Merger5No Solicitation of Other Acquisition Proposals by Terayon7Tremination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12Help In Answering Questions12CAUTTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Pater, Time and Place16Pater, States Entitled to Vote; Quorum16Voting of Proxies17Record Date; Shares Entitled to Vote; Quorum18Noting Statens, States19Atsitance19Merger20Record Date; Shares Entitled to Vote; Quorum16Yoing of Proxies18Solicitation of Proxies19Merger20Record Date; Shares Entitled to Vote; Quorum20Record Date; Shares Entitled to Vote; Quorum20Record Date;</th> <th><b>QUESTIONS AND ANSWERS ABOUT THE MERGER</b></th> <th>1</th>	SUMMARY4The Companies4The Companies5Conditions to the Completion of the Merger5No Solicitation of Other Acquisition Proposals by Terayon7Tremination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12Help In Answering Questions12CAUTTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Pater, Time and Place16Pater, States Entitled to Vote; Quorum16Voting of Proxies17Record Date; Shares Entitled to Vote; Quorum18Noting Statens, States19Atsitance19Merger20Record Date; Shares Entitled to Vote; Quorum16Yoing of Proxies18Solicitation of Proxies19Merger20Record Date; Shares Entitled to Vote; Quorum20Record Date; Shares Entitled to Vote; Quorum20Record Date;	<b>QUESTIONS AND ANSWERS ABOUT THE MERGER</b>	1	
Ihe Companies4Ihe Merger5Conditions to the Completion of the Merger5No Solicitation of Other Acquisition Proposals by Terayon7Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Dpinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paving Agent12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Hotorola, Inc.14Motorola, Inc.14Motorola, Inc.16Purpose of the Special Meeting16Record Date; Shares Entitled to Vote; Ouorum16Voting Of Proxies17Revorability of Proxies18Solicitation of Proxies18Solicitation of Proxies19Nater Surger16Revorability of Proxies18Solicitation of Proxies18Solicitation of Proxies19Assistance19Assistance19Assistance20Resorability of Proxies20Resorability of Proxies20Resorability of Proxies20Resorability of Proxies20Resorability of Proxies20Resorability of Proxies<	The Companies4The Merger5Onditions to the Completion of the Merger5No Solicitation of Other Acquisition Proposals by Terayon7Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals12Help In Answering Questions12Help In Answering Questions14Terayon Communication Systems. Inc.14Motorola. Inc.14Motorola. Inc.16Durose of the Special Meeting16Durose of the Special Meeting16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote: Quorum16Notorola. Inc.17Voing of Proxies17Record Date: Shares Entitled to Vote: Quorum16Shares Owned by Our Directors and Executive Officers17Notorola of Proxies18Solicitation of Proxies19Metral United States Federal Income Tax Consequences18Solicitation of Proxies16Record Date: Shares Entitled to Vote: Quorum16Shares Owned by Our Directors and Executive Officers17Revocability of Proxies18Solicitation of Proxies <t< td=""><td></td><td></td></t<>			
Ihe Merger5Conditions to the Completion of the Merger5Solicitations to the Completion of the Merger Agreement7Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12Paving Agent12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola, Inc.14Motorola, GTG Subsidiary VI Corp.15ITHE SPECIAL MEETING16Date, Time and Place16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote; Ourum16Voting of Proxies17Revorability of Proxies18Solicitation of Proxies18Solicitation of Proxies18Solicitation of Ore Board of Directors18Solicitation of Drectors and Executive Officers19Aksistance19Aksistance19Recommendation of Our Board of Directors20Recommendation of Our Board of Directors20Recommendation of Our Board of Directors18Solicitation of Proxies </td <td>The Merger5Conditions to the Completion of the Merger5Os Solicitation of Other Acquisition Proposals by Terayon7Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals12Help In Answering Questions12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola, Inc.14Motorola GIG Subsidiary VI Corp.15THE SPECIAL MEETING16Parce of Due Stock Officers16Parcose of the Special Meeting16Nater and Place16Parcose of the Special Meeting16States Sciel Meeting16Solicitation of Our Board of Directors16Record Date: Shares Entitled to Vote; Outrum16Voting of Proxies17Record Date: Shares Entitled to Vote; Outrum16Noting Proxies18Solicitation of Proxies19Assistance19HE MERGER20Background of the Merger20Record Date: Shares Entitled to Directors21Record Date: Shares Entitled to Directors18Solicitation of Proxies18<tr< td=""><td></td><td></td></tr<></td>	The Merger5Conditions to the Completion of the Merger5Os Solicitation of Other Acquisition Proposals by Terayon7Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals12Help In Answering Questions12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola, Inc.14Motorola GIG Subsidiary VI Corp.15THE SPECIAL MEETING16Parce of Due Stock Officers16Parcose of the Special Meeting16Nater and Place16Parcose of the Special Meeting16States Sciel Meeting16Solicitation of Our Board of Directors16Record Date: Shares Entitled to Vote; Outrum16Voting of Proxies17Record Date: Shares Entitled to Vote; Outrum16Noting Proxies18Solicitation of Proxies19Assistance19HE MERGER20Background of the Merger20Record Date: Shares Entitled to Directors21Record Date: Shares Entitled to Directors18Solicitation of Proxies18 <tr< td=""><td></td><td></td></tr<>			
Conditions5No Solicitation of Other Acquisition Proposals by Terayon7No Solicitation of Other Acquisition Proposals by Terayon7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Einancial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola, Inc.14Motorola, Inc.14Motorola, Inc.16Purpose of the Special Meeting16Purpose of the Special Meeting16States Forder of Understras17Record Date: Shares Entitled to Vote: Ourum16Votin of Proxies17Record Date: Shares Entitled to Vote: Ourum16Notin of Proxies18Solicitation of Proxies19Assistance19Assistance19Assistance19Merger Marker States20Record Date: The Merger20Record Date: The Merger20Record Date: Shares Entitled to Vote: Ourum16Record Date: Shares Entitled to Vote: Ourum16Record Date: Shares Entitled to Vote: Ourum16Record Date: Shares Entitled to Vote: Ourum16Rec	Conditions to the Completion of the Merger5No Solicitation of Other Acquisition Proposals by Terayon7The Special Meeting8Recommendation of Our Board of Directors9Jutrests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13The Special Meeting14Motorola, Inc.14Motorola, Inc.14Motorola, Inc.16Pare, Substeins, Inc.16Date, Time and Place16Purpose of the Space and Executive Officers16Purpose of the Space and Executive Officers16Coord Date: Shares Entitled to Vote; Ouorum16Voting of Proxies17Voting of Proxies17Noting of Proxies18Other Business19Assistance19Meessiness19Assistance19Meessiness19Assistance19Meessiness20Recommendation of Our Board of Directors20Record Date: Shares Entitled to Vote; Ouorum16Other Business17Meessiness19Assistance19Meessiness19Assistance20Reconnendation of Our Board of Directors2	*		
No Solicitation of Other Acquisition Proposals by Terayon7Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Drinton of Our Einancial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Baying Agent12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Ferayon Communication Systems. Inc.14Motorola. Inc.14Motorola. Inc.16Purpose of the Special Meeting16Recommedation of our Board of Directors16Record Date: Shares Entitled to Vote: Quorum16Voting of Proxies17Voting of Proxies17Voting of Proxies17Record Date: Shares Entitled to Vote: Quorum16Solicitation of Proxies17Noting of Proxies17Record Date: Shares Entitled to Vote: Quorum18Solicitation of Proxies19Phineness19Assistance19Phineness20Record Othe Merger20Reaksround of the Merger20Record Date: Time and Place16Stares Owned by Our Directors and Executive Officers17Record Date: Shares Entitled to Vote: Quorum16S	No Solicitation of Other Acquisition Proposals by Terayon7Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola Inc.14Motorola CIG Subsidiary VI Corp.15Date. Time and Place16Paroso of the Special Meeting16Paroso of the Special Meeting16Notorola CIG Subsidiary VI Corp.16Date. Time and Place16Paroso Convent bour Directors and Executive Officers17Record Date: Shares Entitled to Vote: Ourrum16Voting of Proxies17Solicitation of Proxies18Solicitation of Proxies18Solicitation of Our Directors and Executive Officers20Reaksfrance19HE MERGER20Reaksfrance19Meeting20Reaksfrance20Reaksfrance20Reaksfrance20Reaksfrance20Reaksfrance20Reaksfrance20Reaksfrance20Reaksfrance </td <td></td> <td></td>			
Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Partposes Entitled to Vote; Ouorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Solicitation of Proxies18Other Business19Assistance19Pher Business20Background of the Merger20Reasons for the Merger20Reasons for the Merger20Recommendation of Our Board of Directors18Other Business19Assistance19PHE MERGER20Reasons for the Merger <td< td=""><td>Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola (TG Subsidiary VI Corp.15THE SPECIAL MEETING16Date Special Meeting16Recommendation of our Board of Directors16Recommendation of our Board of Directors16Recommendation of UR Special Meeting16States Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies17Revocability of Proxies17Revocability of Proxies19Assistance19Metric Business19Assistance20Recornmendation of our Board of Directors20Recornmendation of Dur Directors and Executive Officers20Recornmendation of UR Board of Directors20Recornmendation of Our Directors and Executive Officers20Recornmendation of Our Board of Directors20Recornmendation of Our Board of Directors20Recornmendation of Our Board of Directors20Reco</td><td></td><td></td></td<>	Termination of the Merger Agreement7The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola (TG Subsidiary VI Corp.15THE SPECIAL MEETING16Date Special Meeting16Recommendation of our Board of Directors16Recommendation of our Board of Directors16Recommendation of UR Special Meeting16States Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies17Revocability of Proxies17Revocability of Proxies19Assistance19Metric Business19Assistance20Recornmendation of our Board of Directors20Recornmendation of Dur Directors and Executive Officers20Recornmendation of UR Board of Directors20Recornmendation of Our Directors and Executive Officers20Recornmendation of Our Board of Directors20Recornmendation of Our Board of Directors20Recornmendation of Our Board of Directors20Reco			
The Special Meeting8Recommendation of Our Board of Directors9Dpinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13CHE COMPANIES14Teravon Communication Systems, Inc.14Motorola, Inc.14Motorola, Inc.16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Koter Required16Noter Required16States Sources17Voting of Proxies17Solicitation of Proxies18Other Business19Assistance19PHE MERGER20Background of the Merger20Reasons for the Merger20Reasons for the Merger20Reasons for the Merger26Opinion of Our Board of Directors25Opinion of Our Board of Directors25Opinion of Our Board of Directors25Opinion of Our Board of Directors26Commendation of Our Board of Directors26Commendation of Our Board of Directors26Commendation of Our Board of Directors<	The Special Meeting8Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Material United States Federal Income Tax Consequences10Material United States Federal Income Tax Consequences11Regulatory Approvals11Regulatory Approvals12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola, Inc.16Purpose of the Special Meeting16Purpose of the Special Meeting16Purpose of the Special Meeting16Stares Owned by Our Directors and Executive Officers17Vote Required16Solicitation of Proxies18Solicitation of Proxies19Assistance19HE MERGER20Recorns for the Merger21Matrial United States Federal Income Tax Consequences34			
Recommendation of Our Board of Directors9Dinition of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12Help In Answering Ouestions12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13IFHE COMPANIES14Hotorola, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15IFHE SPECIAL MEETING16Parose of the Special Meeting16Record Date: Shares Entitled to Vote: Ouorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Vote Required18Other Basicass18Solicitation of Proxies18Other Business19Assistance19Assistance19Assistance19Dakesfront the Merger20Recommendation of Our Board of Directors20Revocability of Proxies18Other Business19Assistance19Dakesfront the Merger20Record Date: Shares Control of Directors20Record Of the Merger20Record Of the Merger20Record Of the Merger20Record Of the Merger24Record Of the Merger <t< td=""><td>Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Stares Entitled to Vole; Quorum16Solicitation of Proxies17Volte Required16Shares Owned by Our Directors and Executive Officers17Volte Required18Other Business19Assistance19HE MERGER20Background of the Merger20Record of Our Enancial Advisor26Opinion of Our Financial Advisor26Opinion of Our Enancial Advisor26Opinion of Our Enancial Advisor26Opinion of Our Financial Adv</td><td></td><td></td></t<>	Recommendation of Our Board of Directors9Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Stares Entitled to Vole; Quorum16Solicitation of Proxies17Volte Required16Shares Owned by Our Directors and Executive Officers17Volte Required18Other Business19Assistance19HE MERGER20Background of the Merger20Record of Our Enancial Advisor26Opinion of Our Financial Advisor26Opinion of Our Enancial Advisor26Opinion of Our Enancial Advisor26Opinion of Our Financial Adv			
Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12Edelp In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Parpose of the Special Meeting16Parpose of the Special Meeting16Record Date: Shares Entitled to Vote; Ouorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies18Oliter Business18Oliter Business19Assistance19Distances20Rescond of the Merger20Rescond of the Merger2	Opinion of Our Financial Advisor9Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Appraisal Rights11Regulatory Approvals11Paying Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Trayon Communication Systems. Inc.14Motorola Inc.14Motorola CITG Subsidiary VI Corp.15THE SPECIAL MEETING16Purpose of the Special Meeting16Recommendation of our Board of Directors16Shares Owned by Our Directors and Executive Officers17Yoting of Proxies18Solicitation of Proxies18Solicitation of Proxies19THE MERGER20Record Date: Shares Entitled to Vote; Quorum16Yoting of Proxies18Solicitation of Proxies18Solicitation of Proxies19Absistance19THE MERGER20Background of the Merger20Reasons for the Merger25Opinion of Our Board of Directors25Opinion of Our Board of Directors25Opinion of Our Board of Directors26Financial Projections31Itterests of Our Directors and Executive Officers31Itterest of Our Directors and Executive Officers31Itterest of Our Directors and Executive Officers in the Merger32			
Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals11Paying Agent12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Ferayon Communication Systems, Inc.14Motorola, Inc.14Motorola, Inc.16Date, Time and Place16Parpose of the Special Meeting16Record Date: Shares Entitled to Vote; Ourum16Voting of Proxies17Revocability of Proxies18Solicitation of Proxies19Assistance19Assistance20Background of the Merger20Background of the Merger20Recommendation of Our Board of Directors20Revocability of Proxies18Solicitation of Proxies19Assistance19Assistance20Background of the Merger20Record Date, Barnerial Advisor20Recommendation of Our Board of Directors20Recommendation of Our Board of Directors20Recommendation of Droxies20Commendation of Proxies20Commendation of Droxies20Commendation of Our Board of Directors20Recommendation of Our Board of Directo	Interests of Our Directors and Executive Officers in the Merger10Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems. Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Special Meeting16Record Date: Shares Entitled to Vote; Quorum16Voting of Proxies17Revocability of Proxies17Revocability of Proxies18Solicitation of Proxies19THE MERGER20Bakistonce19THE MERGER20Resons for the Merger20Reasons for the Merger20Reasons for the Merger20Reasons for the Merger25Opinion of Our Board of Directors25Opinion of Our Board of Directors<			
Incatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals11Paving Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems. Inc.14Motorola. Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date. Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Voting of Proxies17Voting of Proxies18Solicitation of Proxies18Solicitation of Proxies19Assistance19THE MEGER20Background of the Merger20Recommendation of Our Board of Directors20Recommendation of Our Board of Directors20Recondition of Proxies20Recondition of Proxies20Recons for the Merger24Recons for the Merger24Recons for the Merger26Chancial Projections31	Treatment of Our Stock Options10Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paying Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola, Inc.16Date, Time and Place16Purpose of the Special Meeting16Purpose of the Special Meeting16Stares Entitled to Vote; Ouorum16Voting Or Proxies17Revocability of Proxies17Revocability of Proxies19THE MERGER20Basistance19THE MERGER20Basistance20Basistance20Basistance20Datasitan of Our Board of Directors25Opinion of Our Board of Directors26Financial Projections25Opinion of Our Board of Directors26Easons for the Merger20Basistance20Basistance20Basistance20Basistance20Charled Startes Federal Income Tax Consequences31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34			
Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Regulatory Approvals12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Ferayon Communication Systems, Inc.14Motorola Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Special Meeting16Purpose of the Special Meeting16Recommendation of our Board of Directors16Shares Owned by Our Directors and Executive Officers17Voting of Proxies18Solicitation of Proxies18Solicitation of Proxies19Massinance19Massinance19IHE MERGER20Background of the Merger20Recons for the Merger20Recons for the Merger24Recons for the Merger24Recons for the Merger24Recons for the Merger24Recons for the Merger25Opinion of Our Board of Directors25Spinion of Our Financial Advisor26Financial Projections31	Material United States Federal Income Tax Consequences10Appraisal Rights11Regulatory Approvals11Paving Agent12Help In Answering Ouestions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Motorola, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Shares Owned by Our Directors and Executive Officers17Yoting of Proxies18Solicitation of Proxies19Assistance19THE MERGER20Background of the Merger20Background of the Merger20Background of the Merger20Background of Our Board of Directors20Interests of Our Directors and Executive Officers20Background of the Merger20Background of the Merger20Background of the Merger20Background of Our Board of Directors25Opinion of Our Board of Directors26Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34			
Appraisal Rights11Regulatory Approvals11Paving Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES13CHE COMPANIES14Ierayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Keeord Date: Shares Entitled to Vote: Quorum16Voting of Proxies17Voting of Proxies17Revocability of Proxies19Soliciation of Proxies19Soliciation of Proxies19PHE MERGER20Background of the Merger20Reasons for the Merger20Recons for the Merger26Enancial Projections31	Appraisal Rights11Regulatory Approvals11Paving Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola, GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Shares Owned by Our Directors and Executive Officers17Voting of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Reasons for the Merger20Background of Durectors25Option of Our Board of Directors25Option of Our Board of Directors26Financial Advisor26Financial Advisor31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34			
Regulatory Approvals11Paying Agent12Palp In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13IFHE COMPANIES14Rerayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15IFHE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Record Date; Shares Entitled to Vote; Ouorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies18Olicitation of Proxies18Olicitation of Proxies19IHE MERGER20Background of the Merger20Recons for the Merger20Recons for the Merger24Recons for the Merger24Recons for the Merger26Einancial Projections31	Regulatory Approvals11Paying Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc,14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Background of Our Board of Directors18Other Business21Interests of Our Directors and Executive Officers23Material United States Federal Income Tax Consequences34	*		
Paying Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Pargose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Note Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies18Solicitation of Proxies18Solicitation of Proxies19Her MERGER20Background of the Merger20Recond Dati Cheffer20Recond nof Our Board of Directors20Recond Directors20Revocability of Proxies20Solicitation of Proxies20Resons for the Merger20Recond nof Our Board of Directors20Recond nof Our Board of Directors20Recond nof Our Board of Directors20Resons for the Merger20Recons for the Merger24Recons for the Merger26Econnendation of Our Board of Directors26Einancial Projections31	Paving Agent12Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Shares Owned by Our Directors and Executive Officers17Yoting of Proxies18Solicitation of Proxies18Solicitation of Proxies19HE MERGER20Reasons for the Merger20Reasons for the Merger20Reasons for the Merger20Reasons for Une Board of Directors21It mender20Restrant31It mender31It mender32Material United States Federal Income Tax Consequences34			
Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Voting of Proxies17Revocability of Proxies17Revocability of Proxies18Solicitation of Proxies18Solicitation of Proxies19Assistance19THE MERGER20Background of Merger20Reasons for the Merger20Reasons for the Merger20Quincial Advisor25Opinion of Our Board of Directors25Opinion of Our Board of Directors26Einancial Projections31	Help In Answering Questions12CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola, GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote; Quorum16Yoting of Proxies17Yoting of Proxies17Solicitation of Proxies18Solicitation of Proxies19HE MERGER20Reasons for the Merger20Reasons for the Merger20Reasons for the Merger20Reasons for the Merger20Reasons for Un Board of Directors31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34			
CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13ITHE COMPANIES14Ferayon Communication Systems. Inc.14Motorola, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15ITHE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote; Quorum16Voting of Proxies17Voting of Proxies17Solicitation of Proxies18Solicitation of Proxies19Assistance19ITHE MERGER20Background of the Merger20Reasons for the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Board of Directors25Opinion of Our Board of Directors31	CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS13THE COMPANIES14Terayon Communication Systems, Inc.14Motorola. Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date. Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies19Assistance19HE REGER20Background of the Merger20Background of the Merger20Resons for the Merger20Resons for the Merger20Background of Unectors31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34			
IHE COMPANIES14Iterayon Communication Systems, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15ITHE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Solicitation of Proxies19HE MERGER20Background of the Merger20Reasons for the Merger24Record not of Our Board of Directors25Opinion of Our Board of Directors25Opinion of Our Board of Directors31	THE COMPANIES14Terayon Communication Systems, Inc.14Motorola, Inc.14Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote: Ouorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Revocability of Proxies18Solicitation of Proxies18Solicitation of Proxies19THE MERGER20Background of the Merger20Reasons for the Merger20Reasons for the Merger26Commendation of Our Encies and Executive Officers in the Merger31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34			
Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Motorola Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date. Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Recommendation of Our Board of Directors21Qino of Our Enancial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34			
Motorola, Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Motorola Inc.14Motorola GTG Subsidiary VI Corp.15THE SPECIAL MEETING16Date. Time and Place16Purpose of the Special Meeting16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Recommendation of Our Board of Directors21Qino of Our Enancial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Terayon Communication Systems, Inc.	14	
ITHE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date; Shares Entitled to Vote; Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19ITHE MERGER20Background of the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	THE SPECIAL MEETING16Date. Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Background of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34		14	
ITHE SPECIAL MEETING16Date, Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date; Shares Entitled to Vote; Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19ITHE MERGER20Background of the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	THE SPECIAL MEETING16Date. Time and Place16Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Background of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Motorola GTG Subsidiary VI Corp.	15	
Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date; Shares Entitled to Vote; Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Solicitation of Proxies19Assistance19THE MERGER20Background of the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Purpose of the Special Meeting16Recommendation of our Board of Directors16Record Date; Shares Entitled to Vote; Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34		16	
Recommendation of our Board of Directors16Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19ITE MERGER20Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Recommendation of our Board of Directors16Record Date; Shares Entitled to Vote; Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Solicitation of Proxies19Assistance19THE MERGER20Background of the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Date, Time and Place	16	
Record Date: Shares Entitled to Vote: Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19IHE MERGER20Background of the Merger20Reasons for the Merger21Recommendation of Our Board of Directors25Opinion of Our Financial Advisor31	Record Date; Shares Entitled to Vote; Quorum16Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Purpose of the Special Meeting	16	
Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19IHE MERGER20Background of the Merger20Recommendation of Our Board of Directors24Recommendation of Our Financial Advisor26Financial Projections31	Vote Required16Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Recommendation of our Board of Directors	16	
Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Solicitation of Proxies19Other Business19Assistance19ITHE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Shares Owned by Our Directors and Executive Officers17Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Record Date; Shares Entitled to Vote; Quorum	16	
Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Voting of Proxies17Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Vote Required	16	
Revocability of Proxies18Solicitation of Proxies18Solicitation of Proxies19Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Revocability of Proxies18Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Shares Owned by Our Directors and Executive Officers	17	
Solicitation of Proxies18Other Business19Assistance19IHE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Solicitation of Proxies18Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Voting of Proxies	17	
Other Business19Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Other Business19Assistance19THE MERGER20Background of the Merger20Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Revocability of Proxies	18	
Assistance19THE MERGER20Background of the Merger20Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Assistance19THE MERGER20Background of the Merger20Reasons for the Merger20Recommendation of Our Board of Directors24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Solicitation of Proxies	18	
Image: THE MERGER20Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	THE MERGER20Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Other Business	19	
Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Background of the Merger20Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Assistance	19	
Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Reasons for the Merger24Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	THE MERGER	20	
Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31	Recommendation of Our Board of Directors25Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Background of the Merger	20	
Opinion of Our Financial Advisor26Financial Projections31	Opinion of Our Financial Advisor26Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Reasons for the Merger	24	
Financial Projections 31	Financial Projections31Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Recommendation of Our Board of Directors	25	
	Interests of Our Directors and Executive Officers in the Merger32Material United States Federal Income Tax Consequences34	Opinion of Our Financial Advisor	26	
Interests of Our Directors and Executive Officers in the Merger 32	Material United States Federal Income Tax Consequences 34			
merests of our Directors and Executive Officers in the integer		Interests of Our Directors and Executive Officers in the Merger	32	
	Pagulatory Approvals 26	•		
	<u>Regulatory Approvals</u> 30	Regulatory Approvals	36	

	Page
Appraisal Rights	36
Structure of the Merger	39
Effective Time of the Merger	39
Merger Consideration	39
Treatment of Our Stock Options	40
Surrender of Stock Certificates	40
Lost Certificates	40
Unclaimed Amounts	41
THE MERGER AGREEMENT	42
Representations and Warranties	42
Covenants	44
Directors and Officers Indemnification, Advancement of Expenses, Exculpation and Insurance	46
Options and Employee Stock Purchase Plans; Employee Benefits Matters	46
Efforts to Complete the Merger	46
Conditions to the Completion of the Merger	48
No Solicitation of Other Acquisition Proposals by Terayon	50
Termination: Payment of Termination Fee	52
Fees and Expenses	54
Amendment, Extension and Waiver	54
PROPOSAL TO ADJOURN THE SPECIAL MEETING	55
The Adjournment Proposal	55
Vote Required and Board Recommendation	55
SECURITY OWNERSHIP OF MANAGEMENT AND PRINCIPAL SHAREHOLDERS	56
OTHER MATTERS	57
FUTURE STOCKHOLDER PROPOSALS	57
HOUSEHOLDING OF PROXY STATEMENT	58
WHERE YOU CAN FIND ADDITIONAL INFORMATION	58

ANNEXES		
<u>ANNEX A</u>	Agreement and Plan of Merger, dated as of April 21, 2007, by and among Motorola, Inc.,	
	Motorola GTG Subsidiary VI Corp. and Terayon Communication Systems, Inc.	A-1
<u>ANNEX B</u>	Opinion of Goldman, Sachs & Co.	B-1
<u>ANNEX C</u>	Section 262 of the Delaware General Corporation Law	C-1

# ii

# QUESTIONS AND ANSWERS ABOUT THE MERGER

The following questions and answers address briefly some questions you may have regarding the special meeting and the proposed merger. These questions and answers may not address all questions that may be important to you as a stockholder of Terayon Communication Systems, Inc. Please refer to the more detailed information contained elsewhere in this proxy statement, the annexes to this proxy statement and the documents referred to or incorporated by reference in this proxy statement. In this proxy statement, the terms Terayon, Company, we, our, ours, and to Terayon Communication Systems, Inc. and references to subsidiaries of the Company includes all wholly owned subsidiaries of the Company.

# Q: Why am I receiving this proxy statement?

A: We have agreed to be acquired by Motorola under the terms of a merger agreement that is described in this proxy statement. A copy of the merger agreement is included as Annex A to this proxy statement. In order to complete the merger, our stockholders must vote to adopt the merger agreement and approve the merger. This proxy statement contains important information about the merger and our special meeting of stockholders.

# **Q:** Where and when is the special meeting?

A: The special meeting will take place at our corporate headquarters located at 2450 Walsh Avenue, Santa Clara, California 95051, on June 28, 2007, at 11:00 a.m., local time.

# **Q:** What matters will I be asked to vote on at the special meeting?

A: You will be asked to vote on a proposal to adopt the merger agreement and approve the merger and a proposal for the adjournment of the special meeting, if necessary, to solicit additional proxies in the event that there are not sufficient votes in favor of adoption of the merger agreement and approval of the merger at the time of the special meeting.

# **Q:** What will happen to Terayon as a result of the merger?

A: If the merger is completed, we will become a wholly owned subsidiary of Motorola.

# **Q:** Who is eligible to vote at the special meeting?

A: Holders of our common stock at the close of business on May 18, 2007 are entitled to receive notice of the special meeting and to vote the shares of our common stock that they held at that time at the special meeting, or at any adjournments or postponements of the special meeting.

#### Q: What will happen to my shares of Terayon common stock after the merger?

A: At the effective time of the merger, each outstanding share of our common stock will automatically be canceled and will be converted into the right to receive a per share amount equal to \$1.80 in cash, without interest.

# **Q:** What do I need to do now?

A:

After carefully reading and considering the information contained in this proxy statement, please vote your shares as soon as possible. You can vote your shares prior to the special meeting (1) by mail with the enclosed proxy card, in accordance with the instructions on the proxy card, (2) by telephone by calling the toll-free number that appears on the enclosed proxy card and following the instructions given, or (3) by means of the Internet, by following the Internet voting instructions provided on the enclosed proxy card. If you hold your shares in street name (which means that you hold your shares through a bank, brokerage firm or other nominee), you should vote in accordance with the directions on the voting instruction card that your bank, brokerage firm or other nominee provides to you.

# **Q:** Is the merger expected to be taxable to me?

A: Generally, yes. The receipt of \$1.80 in cash for each share of our common stock pursuant to the merger will be a taxable transaction for U.S. federal income tax purposes. For U.S. federal income tax purposes, you will generally recognize gain or loss as a result of the merger measured by the difference, if any, between \$1.80 per share and your adjusted tax basis in that share.

You should read The Merger Material United States Federal Income Tax Consequences beginning on page 34 for a more complete discussion of the federal income tax consequences of the merger. Tax matters can be complicated and the tax consequences of the merger to you will depend on your particular tax situation. You should also consult your tax advisor on the tax consequences of the merger to you.

#### Q: How does the Terayon board of directors recommend that I vote?

A: Our board of directors unanimously recommends that our stockholders vote FOR the proposal to adopt the merger agreement and approve the merger, and FOR the proposal to adjourn the special meeting for the purpose of soliciting additional proxies, if necessary. You should read The Merger Reasons for the Merger beginning on page 24 for a discussion of the factors that our board of directors considered in deciding to recommend the adoption of the merger agreement and approval of the merger.

#### Q: What vote of our stockholders is required to adopt the merger agreement and approve the merger?

A: For us to complete the merger, stockholders holding at least a majority of the shares of our common stock outstanding at the close of business on May 18, 2007 must vote FOR the adoption of the merger agreement and approval of the merger.

#### **Q:** What vote of our stockholders is required to approve the adjournment of the special meeting?

A: The proposal to adjourn the special meeting for the purpose of soliciting additional proxies, if necessary, will be approved if the votes cast in favor of the proposal by shares of our common stock, present in person or represented by proxy and entitled to vote on the subject matter, exceed the votes cast against the proposal.

#### Q: Am I entitled to appraisal rights?

A: Yes. Under the General Corporation Law of the State of Delaware, holders of our common stock who do not vote in favor of adoption of the merger agreement and approval of the merger will have the right to seek appraisal of the fair value of their shares as determined by the Delaware Court of Chancery if the merger is completed, but only if they submit a written demand for an appraisal prior to the vote on the adoption of the merger agreement and approval of the merger, vote against, or abstain from voting for, the proposal to adopt the merger agreement and approve the merger, and otherwise comply with the Delaware law procedures explained in this proxy statement.

#### Q: How do I cast my vote?

A: If you are the holder of record of shares of our common stock, you can vote by any of the following methods:

you may indicate your vote on the enclosed proxy card and complete, sign, date and return the proxy card in the accompanying pre-addressed, postage paid envelope;

you may vote by telephone by calling the toll-free number that appears on the enclosed proxy card and following the instructions given;

you may vote electronically via the Internet by going to the website that appears on the enclosed proxy card and following the instructions given; or

you may attend the special meeting and vote in person.

If you sign, date and mail your proxy and do not indicate how you want to vote, your proxy will be voted FOR the adoption of the merger agreement and the approval of the merger, and FOR the proposal to adjourn the special meeting to solicit additional proxies, provided that no proxy that is specifically marked AGAINST the proposal to adopt the merger agreement and approve the merger will be voted in favor of the adjournment proposal, unless it is specifically marked FOR the adjournment proposal.

# Q: If my Terayon shares are held in street name by my broker or bank, will my broker or bank vote my shares for me?

A: Your broker or bank will vote your Terayon shares only if you provide instructions on how to vote. You should follow the directions provided by your broker or bank regarding how to instruct your broker or bank to vote your shares. Without instructions, your shares will not be voted, which will have the effect of a vote against the adoption of the merger agreement and approval of the merger.

# **Q:** What happens if I abstain from voting or fail to vote on the proposals or instruct my broker to vote on the proposals?

A: If you abstain from voting, fail to cast your vote in person, by proxy, or electronically via the Internet or by telephone, or fail to give voting instructions to your broker, bank or other nominee, it will have the same effect as a vote against the proposal to adopt the merger agreement and approve the merger, and it will have no effect on the proposal to adjourn the special meeting for the purpose of soliciting additional proxies.

#### Q: Can I change my vote after I have returned my proxy?

A: Yes. If you are a shareholder of record, you may revoke your proxy and change your vote at any time before your proxy card is voted at the special meeting. You can do this in one of four ways:

by providing a written instrument or transmission to our corporate secretary prior to the special meeting stating that you revoke your proxy;

by completing and submitting to our corporate secretary a proxy in writing via mail dated later than your original proxy relating to the same shares;

by voting via the Internet or by telephone following the date of your original proxy relating to the same shares; or

by attending the special meeting and voting in person, which will automatically cancel any proxy previously given; your attendance at the special meeting alone, however, will not revoke any proxy that you have previously given.

If you have instructed a broker, bank or other nominee to vote your shares, you must follow the directions received from your broker, bank or other nominee to change those instructions.

#### Q: Should I send in my stock certificates now?

A: No. After the merger is completed, you will receive a transmittal form with instructions for the surrender of certificates formerly representing shares of our common stock. Please do not send in your stock certificates with your proxy.

#### Q: What should I do if I receive more than one set of voting materials?

A: You may receive more than one set of voting materials, including multiple copies of this proxy statement and multiple proxy cards or voting instruction cards. For example, if you hold your shares in more than one brokerage account, you will receive a separate voting instruction card for each brokerage account in which you hold shares. If you are a holder of record and your shares are registered in more than one name, you will receive more than one proxy card. Please complete, sign, date and return each proxy card and voting instruction card that you receive.

#### **Q:** When do you expect the merger to be completed?

A: We are working to complete the merger as quickly as possible. We currently expect to complete the merger in the second or third quarter of 2007. However, we cannot predict the exact timing of the merger because the merger is

#### Table of Contents

subject to regulatory approvals and other closing conditions. While we expect to obtain all required regulatory approvals, we cannot assure you that these regulatory approvals will be obtained and, even if they are ultimately obtained, they might not be obtained for a substantial period of time following the special meeting.

#### **Q:** Who can help answer my questions?

A: If you need assistance in completing your proxy card or voting your shares or have questions regarding the special meeting, please contact MacKenzie Partners, which is assisting us in the solicitation of proxies, at 800-322-2885 (toll-free) or 212-929-5500 (collect), or write to the following address:

MacKenzie Partners 105 Madison Avenue New York, NY 10016

# SUMMARY

This summary highlights selected information from this proxy statement and may not contain all the information that is important to you. You should carefully read this entire proxy statement and the other documents to which we have referred you. See also Where You Can Find Additional Information on page 58. We have included page references parenthetically to direct you to a more complete description of the topics presented in this summary.

# The Companies (Page 14)

**Terayon Communication Systems, Inc.** 2450 Walsh Avenue

Santa Clara, California 95051 (408) 235-5500

Terayon currently develops, markets and sells digital video equipment to network operators and content aggregators who offer video services. Terayon s primary products include the Network CherryPicker line of digital video processing systems and the CP 7600 line of digital-to-analog decoders. Its products are used for multiple digital video applications, including the rate shaping of video content to maximize the bandwidth for standard definition (SD) and high definition (HD) programming, grooming customized channel line-ups, carrying local ads for local and national advertisers and branding by inserting corporate logos into programming. Terayon s products are sold primarily to cable operators, television broadcasters, telecom carriers and satellite providers in the United States, Europe and Asia.

Terayon was incorporated in California in 1993 and reincorporated in Delaware in 1998.

Motorola, Inc. 1303 East Algonquin Road Schaumburg, Illinois 60196 (847) 576-5000

Motorola builds, markets and sells products, services and applications that make simple and seamless connections to people, information and entertainment possible through broadband, embedded systems and wireless networks. Motorola s vision is to provide cutting-edge technologies that empower mobile consumers to go anywhere and do anything without sacrificing connectivity. This is seamless mobility.

#### **Business Segments**

Motorola reports financial results for the following three operating business segments:

#### Mobile Devices Segment

The Mobile Devices segment designs, manufactures, sells and services wireless handsets with integrated software and accessory products, and licenses intellectual property. In 2006, the segment s net sales represented 66% of Motorola s consolidated net sales.

#### Networks and Enterprise Segment

The Networks and Enterprise segment designs, manufactures, sells, installs and services: (i) cellular infrastructure systems and wireless broadband systems to public carriers and other wireless service providers (referred to as the public networks market), and (ii) analog and digital two-way radio, voice and data communications products and systems, as well as wireless broadband systems, to a wide range of public safety, government, utility, transportation and other worldwide enterprise markets (referred to as the private networks market). In January 2007, the segment completed the acquisition of Symbol Technologies, Inc. (Symbol), a leader in providing products and systems used in end-to-end enterprise mobility solutions. Symbol will become the cornerstone of the

#### **Table of Contents**

segment s enterprise mobility strategy. In 2006, the segment s net sales represented 26% of Motorola s consolidated net sales.

#### **Connected Home Solutions Segment**

The Connected Home Solutions segment designs, manufactures, sells and services: (i) cable television, Internet Protocol (IP) video and broadcast network set-top boxes (digital entertainment devices), (ii) end-to-end digital video system solutions, (iii) broadband access networks, and (iv) IP-based data and voice products (including modems). In 2006, the segment s net sales represented 8% of Motorola s consolidated net sales.

Motorola is a corporation organized under the laws of the State of Delaware as the successor to an Illinois corporation organized in 1928. Motorola s principal executive offices are located at 1303 East Algonquin Road, Schaumburg, Illinois 60196.

#### Motorola GTG Subsidiary VI Corp.

1303 East Algonquin Road Schaumburg, Illinois 60196 (847) 576-5000

Merger Sub is a Delaware corporation and a wholly owned subsidiary of Parent. Merger Sub was organized solely for the purpose of completing a merger. It has not conducted any activities to date other than activities incidental to its formation and in connection with the transactions contemplated by the merger agreement.

#### The Merger

*Structure of the Merger (see page 39).* This proxy statement relates to the proposed acquisition of our company by Motorola pursuant to an agreement and plan of merger, dated as of April 21, 2007, among Motorola, Merger Sub and us. We have attached a copy of this agreement, which we refer to as the merger agreement, as Annex A to this proxy statement. We encourage you to read the merger agreement in its entirety.

Under the terms of the merger agreement, Merger Sub will merge with and into Terayon, with Terayon surviving the merger as a wholly owned subsidiary of Motorola.

*Merger Consideration (see page 39).* At the effective time of the merger, each share of our common stock (other than shares owned by us, Motorola, or Merger Sub, and other than shares for which appraisal rights have been validly exercised under Delaware law) will be converted into the right to receive \$1.80 in cash, without interest. Based on the number of shares of our common stock outstanding on April 21, 2007, the aggregate consideration paid by Motorola to our stockholders will be approximately \$140 million.

*Completion.* We expect to complete the merger as soon as practicable after the adoption of the merger agreement and approval of the merger by our stockholders and after all other conditions to the merger have been satisfied or waived.

#### Conditions to the Completion of the Merger (Page 48)

The obligations of each of Motorola and Merger Sub, on the one hand, and us, on the other hand, to complete the merger depend on the satisfaction or waiver of, on or prior to the effective time of the merger, a number of conditions, including:

receipt of the required vote to adopt the merger agreement and approve the merger by our stockholders at the special meeting;

expiration or termination of the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and clearance under other applicable antitrust laws;

absence of legal restraints making illegal or prohibiting the completion of the merger or the other transactions contemplated by the merger agreement;

performance in all material respects of the other party s obligations under the merger agreement; and

the representations and warranties of the other party set forth in the merger agreement being true and correct without regard to materiality and material adverse effect qualifiers, as of the date of the merger agreement and as of the completion date (except when made as of an earlier date, in which case as of that date), except to the extent that the aggregate of all inaccuracies with respect to such representations and warranties would not have a material adverse effect on such party, and, solely with respect to us, except for capitalization, corporate authority and takeover statute representations, which must be true and accurate in all material respects.

The obligations of Motorola and Merger Sub to complete the merger are subject to the following additional conditions:

the absence of any pending or threatened governmental actions, investigations or proceedings, or issuance of a related order, challenging the merger, seeking to prohibit Motorola s ability to own or operate either of our or its businesses or assets, or those of our respective subsidiaries, or to exercise ownership rights over the stock of the surviving corporation after the merger;

the absence of any pending or threatened governmental actions, investigations or proceedings that may reasonably be expected to result in (1) criminal sanctions on us or our subsidiaries, or (2) material fines to a governmental entity or restitution to a third party, in each case resulting from the conviction of us or any of our subsidiaries of a crime, or the settlement with a governmental entity for purposes of closing an investigation, being imposed on Motorola or any of its affiliates;

all governmental authorizations, consents, approvals, declarations, notices, filings and waiting period expirations shall have been made or obtained, except those, the failure of which to make or obtain could not reasonably be expected to have a material adverse effect on us or Motorola, or provide a reasonable basis to conclude the parties or any of their affiliates would be subject to the risk of criminal sanctions or any of their representatives would be subject to the risk of criminal civil sanctions;

the continued enforceability of specified contracts of ours;

no occurrence since the date of the merger agreement that has had, or could reasonably be expected to have, a material adverse effect (as defined in the merger agreement) on us;

our receipt of all necessary certifications under Sections 302 and 906 of the Sarbanes-Oxley Act on our reports filed with the SEC;

our having filed all required documents with the SEC, and all required amendments to such documents, and the absence of any information in any amendments that is materially and adversely different from the information in the documents amended by such amendments;

the absence of any pending or threatened action, suit, investigation or proceeding challenging the validity and ownership of the intellectual property owned by us, except those that are not expected to have a material adverse effect on us;

the SEC shall not have recommended any charges or enforcement action against us or our officers or directors, authorized such recommendation or issued a Wells Notice to us or our officers or directors, in connection with the SEC investigation relating to certain accounting matters that was commenced in December 2005 and

recently discontinued, and there shall not be any pending governmental action, suit, proceeding or investigation against any of our or our subsidiaries directors or officers, or certain key employees, relating to the SEC investigation;

our having at least \$15 million of cash and cash equivalents (\$13 million if the merger is completed after July 31, 2007, and, in either case, reduced by the amount of all reasonable merger-related expenses (including payments required on or prior to, or as a result of, completion of the merger) paid after April 21, 2007), net of indebtedness;

three-quarters of a small group of key employees, and 80% of a larger group of employees, still being employed by us and performing their usual and customary duties immediately prior to completion of the merger; and

our having filed specified tax returns and other tax returns, if any, that are identified as being delinquent prior to completion of the merger.

The merger is not conditioned upon Motorola or Merger Sub obtaining financing.

#### No Solicitation of Other Acquisition Proposals by Terayon (see page 50)

The merger agreement contains restrictions on our ability to solicit or engage in discussions or negotiations with a third party with respect to a proposal to acquire a significant interest in our company. Notwithstanding these restrictions, the merger agreement provides that under specified circumstances, if prior to our stockholder vote to adopt the merger agreement and approve the merger we receive an acquisition proposal from a third party that our board of directors determines in good faith, after consultation with outside legal counsel, is a superior proposal, or is more favorable from a financial point of view than the merger and is reasonably likely to lead to a superior proposal, we may, if our board of directors also determines in good faith (after consultation with outside legal counsel) that such action is required to discharge our board of directors fiduciary duties to our stockholders, furnish nonpublic information to that third party and engage in negotiations regarding an acquisition proposal with that third party. Our board of directors may change or withdraw its recommendation of the merger agreement or approve or recommend any superior proposal, if a superior proposal is pending, our board of directors has determined in good faith, after receipt of advice of outside legal counsel, that such action is required to discharge its fiduciary duties to our stockholders, and we have given Motorola three business days to match the superior proposal. We may also terminate the merger agreement and enter into an acquisition agreement with respect to a superior proposal, as described under Termination of the Merger Agreement below.

Termination of the Merger Agreement (see page 52)

The merger agreement may be terminated at any time prior to the effective time of the merger, regardless of whether our stockholders have adopted the merger agreement and approve the merger:

- by mutual written consent of Motorola and us;
- by either Motorola or us, by written notice if:

the merger has not been completed on or before September 21, 2007, or a later date as extended by mutual written consent of Motorola and us, unless the party that seeks to terminate has breached or failed to perform in any material respect its obligations under the merger agreement in any manner that has been the principal cause of, or has primarily resulted in, the merger not being completed by that date;

our stockholders do not adopt the merger agreement and approve the merger at the special meeting, or at any adjournment or postponement thereof, unless the party that seeks to terminate has breached or failed to perform in any material respect its obligations under the merger agreement in any manner that has been the principal cause of, or has primarily resulted in, the merger not being completed by that date; or

any order permanently restraining, enjoining or otherwise prohibiting completion of the merger becomes final and nonappealable (provided that the party that seeks to terminate used commercially reasonable efforts

to have the order lifted).

#### by Motorola if:

our board fails to recommend approval of the merger agreement and the merger in this proxy statement, changes, or resolves to change, its recommendation to adopt the merger agreement and approve the merger, recommends to the stockholders a competing transaction or publicly announces that it intends to do so, or enters into any alternative acquisition agreement accepting any competing transaction;

a tender offer or exchange offer for our outstanding shares of capital stock is commenced, and our board fails to recommend to our stockholders against accepting the offer;

#### **Table of Contents**

our board, upon request of Motorola following receipt of a proposal or offer for a competing transaction, fails to reaffirm to Motorola the approval or recommendation of the merger and the merger agreement within five business days of the request;

we or any of our officers, directors, representatives or agents knowingly and materially breaches our obligations under the non-solicitation provisions or specified provisions relating to the special meeting of our stockholders in the merger agreement; or

we breach any of our representations, warranties, covenants or agreements in the merger agreement, or any of our representations or warranties becomes untrue after the date of the merger agreement, so that the related closing condition would not be satisfied, and we do not cure the breach within 20 days after we receive written notice of it from Motorola (although Motorola may not terminate the merger agreement if Motorola or Merger Sub is in material breach of the merger agreement);

by us if:

prior to the stockholder vote with respect to the adoption of the merger agreement and approval of the merger, our board of directors, in compliance with the nonsolicitation provisions of the merger agreement, approves or recommends to our stockholders a superior proposal, Motorola fails to match the superior proposal within three business days after we notify Motorola of the superior proposal, and we pay Motorola the termination fee described below; or

Motorola or Merger Sub breaches any of their representations, warranties, covenants or agreements in the merger agreement, or any of their representations or warranties becomes untrue after the date of the merger agreement, so that the related closing condition would not be satisfied, and they do not cure the breach within 20 days after they receive written notice of it from us (although we may not terminate the merger agreement if we are in material breach of the merger agreement).

*Termination Fee (see page 52).* The merger agreement provides that, in specified circumstances, we may be required to pay Motorola a termination fee of \$5.25 million.

#### **The Special Meeting**

*Date, Time and Place (see page 16).* The special meeting of our stockholders will be held at our corporate headquarters located at 2450 Walsh Avenue, Santa Clara, California 95051, at 11:00 a.m., local time, on June 28, 2007. At the special meeting, our stockholders will be asked to vote on the proposal to adopt the merger agreement and approve the merger and the proposal to adjourn the special meeting to solicit additional proxies, if necessary.

*Record Date, Voting Power (see page 16).* Our stockholders are entitled to vote at the special meeting if they owned shares of our common stock as of the close of business on May 18, 2007, the record date. On the record date, there were 77,637,177 shares of our common stock entitled to vote at the special meeting. Stockholders will have one vote at the special meeting for each share of our common stock that they owned on the record date.

Voting (see page 17). Holders of record of shares of our common stock can vote by any of the following methods:

by completing, signing, dating and returning the enclosed proxy card in the accompanying pre-addressed, postage paid envelope;

by calling the toll-free number that appears on the enclosed proxy card and following the instructions given;

by going to the Internet website that appears on the enclosed proxy card and following the instructions given; or

by appearing and voting in person by ballot at the special meeting.

Brokers or banks holding shares of our common stock in street name may vote the shares only if the underlying stockholders provide instructions on how to vote. Brokers or banks will provide stockholders with directions on how to instruct the broker or bank to vote the shares. All properly completed proxies that we receive prior to the vote at the special meeting, and that are not revoked, will be voted in accordance with the instructions

#### Table of Contents

indicated on the proxies. If no direction is indicated on a properly completed proxy returned to us, the underlying shares will be voted FOR the adoption of the merger agreement and approval of the merger and FOR the proposal to adjourn the special meeting to solicit additional proxies, if necessary, provided that no proxy that is specifically marked AGAINST the proposal to adopt the merger agreement and approve the merger will be voted in favor of the adjournment proposal, unless it is specifically marked FOR the adjournment proposal.

As of the date of this proxy statement, we know of no matters that will be presented for consideration at the special meeting other than as described in this proxy statement. If, however, other matters are brought before the special meeting, the persons named as proxies will vote in accordance with their judgment on such other matters unless otherwise indicated on the proxy.

*Revocability of Proxies (see page 18).* You may change your vote at any time before your proxy is voted at the special meeting. You can do this in one of four ways:

by providing a written instrument or transmission to our corporate secretary prior to the special meeting stating that you revoke your proxy;

by completing and submitting to our corporate secretary a proxy in writing via mail dated later than your original proxy relating to the same shares;

by voting via the Internet or by telephone following the date of your original proxy relating to the same shares; or

by attending the special meeting and voting in person, which will automatically cancel any proxy previously given; your attendance at the special meeting alone, however, will not revoke any proxy that you have previously given.

If you instructed your broker to vote your shares, you must follow directions from your broker to change these instructions.

*Vote Required (see page 16).* Approval of the proposal to adopt the merger agreement and approve the merger requires the affirmative vote of stockholders holding a majority of the shares of our common stock outstanding at the close of business on May 18, 2007. Approval of the proposal to adjourn the special meeting for the purpose of soliciting additional proxies, if necessary, requires the affirmative vote of the holders of a majority of the outstanding shares voted on such proposal.

*Shares Owned by Our Directors and Executive Officers (see page 17).* On May 18, 2007, the record date, our directors and executive officers beneficially owned and were entitled to vote approximately 7.8% percent of the shares of our common stock outstanding on that date.

#### **Recommendation of Our Board of Directors (see page 16)**

Our board of directors has unanimously determined that the merger agreement is advisable, fair to and in the best interests of, our stockholders. The board of directors unanimously recommends that our stockholders vote FOR the adoption of the merger agreement and approval of the merger, and FOR the proposal to adjourn the special meeting to solicit additional proxies, if necessary.

#### **Opinion of Our Financial Advisor (see page 26)**

Goldman, Sachs & Co. rendered its opinion to our board of directors that, as of April 21, 2007, and based upon and subject to the factors and assumptions set forth therein, the \$1.80 per share of common stock in cash to be received by the holders of shares of our common stock pursuant to the merger agreement was fair from a financial point of view to such holders.

The full text of the written opinion of Goldman Sachs, dated April 21, 2007, which sets forth the assumptions made, procedures followed, matters considered and limitations on the review undertaken in connection with the opinion, is attached as Annex B to this proxy statement. We urge you to read the opinion carefully and in its entirety. Goldman Sachs provided its opinion for the information and assistance of our board of directors in connection with its consideration of the transaction. The Goldman Sachs opinion is not a recommendation as to how any holder of

shares of our common stock should vote with respect to the transaction. Pursuant to the amended letter agreement between Goldman Sachs and us, we have agreed to pay Goldman Sachs a transaction fee of approximately \$4.0 million in connection with the transaction. \$1.0 million of the transaction fee, net of retainer fees paid by us to Goldman Sachs through the end of December 2006, became due upon the signing of the merger agreement based on our request that Goldman Sachs undertake a study in order to enable it to render its opinion as to the fairness from a financial point of view of the \$1.80 per share of common stock in cash to be received by the holders of shares of our common stock pursuant to the merger agreement. The remaining \$3.0 million of the transaction fee is due upon the consummation of the transaction. In addition, retainer fees in the amount of \$50,000 per month from January 2007 onwards have been deferred under the amended letter agreement, and will become due upon the earlier of the consummation of the transaction or the termination of Goldman Sachs is providing and has provided certain investment banking services to Motorola.

# Interests of Our Directors and Executive Officers in the Merger (see page 32)

In considering the recommendation of our board of directors to vote for the proposal to adopt the merger agreement and approve the merger, you should be aware that all of our directors and executive officers have personal interests in the merger that are, or may be, different from, or in addition to, our stockholders interests. Our executive officers (including Jerry Chase, who is also a member of our board of directors) are entitled to benefits under their employment agreements pursuant to which they will receive severance benefits if their employment is terminated following the completion of the merger under specified circumstances. Additionally, all options held by our directors and executive officers to purchase shares of our common stock granted under our 1995 Stock Option Plan, 1997 Equity Incentive Plan, 1998 Non-Employee Directors Stock Option Plan, and 1999 Non-Officer Equity Incentive Plan, including any unvested portion of such options, will be cashed out in the merger, as described under Treatment of Our Stock Options below. In addition, the terms of the merger agreement provide for the continued indemnification of our directors and officers following the effective time of the merger.

Our board of directors was aware of these interests and considered them, among other matters, when approving the merger.

#### Treatment of Our Stock Options (see page 40)

All options to purchase shares of our common stock granted under our 1995 Stock Option Plan, 1997 Equity Incentive Plan, 1998 Non-Employee Directors Stock Option Plan, and 1999 Non-Officer Equity Incentive Plan, as well as all options granted outside of any plan, including any unvested portion of such options, will be cashed out in the merger. For these purposes, cashed out in the merger means that the option, whether vested or unvested, will be cancelled in the merger in exchange for a cash payment equal to the product of (i) the excess, if any, of \$1.80 over the applicable per share option exercise price and (ii) the number of shares of our common stock subject to the option at such time, without interest and less any applicable withholding taxes. The aggregate cash-out value of all of our options that are in-the-money is approximately \$188,000.

# Material United States Federal Income Tax Consequences (see page 34)

The receipt of \$1.80 in cash in exchange for each share of our common stock pursuant to the merger will be a taxable transaction for U.S. federal income tax purposes. For U.S. federal income tax purposes, you will generally recognize gain or loss as a result of the merger measured by the difference, if any, between \$1.80 per share of common stock and your adjusted tax basis in that share.

You should read The Merger Material United States Federal Income Tax Consequences beginning on page 34 for a more complete discussion of the federal income tax consequences of the merger. Tax matters can be complicated, and the tax consequences of the merger to you will depend on your particular tax situation. We urge you to consult your tax advisor on the tax consequences of the merger to you.

# Appraisal Rights (see page 36)

Holders of record of shares of our common stock who do not wish to accept the cash consideration payable pursuant to the merger may seek, under Section 262 of the General Corporation Law of the State of Delaware, judicial appraisal of the fair value of their shares by the Delaware Court of Chancery. This value could be more or less than or the same as the merger consideration for our common stock. This right to appraisal is subject to a number of restrictions and technical requirements. Generally, in order to properly demand appraisal, among other things:

you must not vote in favor of the proposal to adopt the merger agreement and approve the merger;

you must make a written demand on us for appraisal in compliance with the General Corporation Law of the State of Delaware before the vote on the proposal to adopt the merger agreement and approve the merger at the special meeting; and

you must hold your shares of record continuously from the time of making a written demand for appraisal through the effective time of the merger.

Merely voting against adoption of the merger agreement and approval of the merger will not preserve your right to appraisal under Delaware law. Also, because a submitted proxy not marked against or abstain will be voted for the proposal to adopt the merger agreement and approve the merger, the submission of a proxy not marked against or abstain will result in the waiver of appraisal rights. If you hold shares in the name of a broker or other nominee, you must instruct your nominee to take the steps necessary to enable you to demand appraisal for your shares. If you or your nominee fails to follow all of the steps required by Section 262 of the General Corporation Law of the State of Delaware, you will lose your right of appraisal.

Annex C to this proxy statement contains the full text of Section 262 of the General Corporation Law of the State of Delaware, which relates to your right of appraisal. We encourage you to read these provisions carefully and in their entirety.

# **Regulatory Approvals (see page 36)**

The merger is subject to discretionary review by the Antitrust Division of the U.S. Department of Justice and the Federal Trade Commission to determine whether it is in compliance with applicable antitrust laws. The Hart-Scott-Rodino Antitrust Improvements Act of 1976 and the rules promulgated thereunder prohibit us from completing the merger until we have furnished certain information and materials to the Antitrust Division of the Department of Justice and the Federal Trade Commission, and the required waiting period has ended. Both Motorola and we filed the required notification and report forms on May 18, 2007. The waiting period will expire at 11:59 p.m., Eastern time, on June 18, 2007, unless extended by a request for more information or shortened by an early termination notice. In addition, filings are required to be made with, and regulatory approvals are required to be obtained from, antitrust or competition authorities in other nations, including Germany, Israel and the Ukraine. All of these filings have been made or will be made shortly. Until approvals have been received from antitrust or competition authorities in each of those jurisdictions, we are prohibited from completing the merger. Neither we nor Motorola has yet obtained any of the governmental or regulatory approvals required to complete the merger. The completion of the merger also is subject to compliance with the General Corporation Law of the State of Delaware.

While we expect to obtain all required regulatory approvals, we cannot assure you that these regulatory approvals will be obtained or that the granting of these regulatory approvals will not involve the imposition of conditions on the completion of the merger or require changes to the terms of the merger that would have a materially adverse effect on the combined company. These conditions or changes could require the grant of a complete or partial license, a divestiture or spin-off, or the holding separate of assets or businesses and could result in the conditions to Motorola s obligation to complete the merger not being satisfied.

#### **Table of Contents**

# **Paying Agent**

Computershare Shareholder Services or another comparable institution will act as the paying agent in connection with the merger.

# A COPY OF THE MERGER AGREEMENT IS INCLUDED IN THIS PROXY STATEMENT AS ANNEX A. YOU ARE STRONGLY ENCOURAGED TO READ IT CAREFULLY AND IN ITS ENTIRETY.

#### Help in Answering Questions

If you have questions about the special meeting or the merger after reading this document, please contact MacKenzie Partners, which is assisting us in the solicitation of proxies, at 800-322-2885 (toll-free) or 212-929-5500 (collect), or write to the following address:

MacKenzie Partners 105 Madison Avenue New York, NY 10016

# CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This proxy statement contains forward-looking statements within the meaning of Section 21E of the Securities Exchange Act of 1934, as amended. Statements other than statements of historical fact, including the expected timetable for completing the proposed merger between us and Motorola, and any other statements regarding our future expectations, beliefs, goals or prospects, are forward-looking statements for the purposes of federal and state securities laws. Forward-looking statements are commonly identified by words such as may, will. should. would. expects, contemplates, anticipates, believes, estimates, predicts. future. intends, anticipates and other terms with meanings. These forward-looking statements, including without limitation statements relating to our projected financial and operating results under the heading Financial Projections, reflect our management s current expectations, estimates, forecasts and projections, and are subject to a number of risks and uncertainties that may cause actual results or events to differ materially from those expressed in, or implied by, such statements. Our future financial condition and results of operations, as well as any forward-looking statements, are subject to change and to inherent risks and uncertainties. Risks and uncertainties pertaining to the following factors, among others, could cause actual results or events to differ materially from those indicated by such forward-looking statements:

we may be unable to complete the merger due to failure to obtain stockholder approval or required regulatory clearances, to satisfy other conditions to completion of the merger, or for other reasons;

the retention of certain of our key employees;

the impact or outcome of any legal proceeding that may be instituted against us and others relating to the merger, and other uncertainties relating to the proposed merger;

unexpected costs or liabilities resulting from the proposed transaction;

diversion of management s attention from the operations of the business as a result of preparations for the proposed merger;

the risk that announcement of the proposed merger may negatively affect our relationship with our customers, suppliers, and employees;

the risk that we are unable to achieve the financial projections set forth under Financial Projections below, for the reasons set forth in that section;

risks associated with any failure to obtain intellectual property licenses that may be required and to pay associated royalties, including our recent failure to do so with respect to Dolby Laboratories, as identified in our Quarterly Report on Form 10-Q for the quarter ended March 31, 2007; and

other factors described in our Annual Report on Form 10-K for the fiscal year ended December 31, 2006 and our most recent Quarterly Report on Form 10-Q filed with the SEC.

You should consider the cautionary statements contained or referred to in this section in connection with any subsequent written or oral forward-looking statements that may be issued by us or persons acting on our behalf. We do not undertake any obligation to release publicly any revisions to any forward-looking statements contained herein to reflect events or circumstances that occur after the date of this proxy statement or to reflect the occurrence of unanticipated events, except as we are required to do by law.

#### THE COMPANIES

#### **Terayon Communication Systems, Inc.** 2450 Walsh Avenue Santa Clara, California 95051 (408) 235-5500

Terayon currently develops, markets and sells digital video equipment to network operators and content aggregators who offer video services. Terayon s primary products include the Network CherryPicker line of digital video processing systems and the CP 7600 line of digital-to-analog decoders. Its products are used for multiple digital video applications, including the rate shaping of video content to maximize the bandwidth for standard definition (SD) and high definition (HD) programming, grooming customized channel line-ups, carrying local ads for local and national advertisers and branding by inserting corporate logos into programming. Terayon s products are sold primarily to cable operators, television broadcasters, telecom carriers and satellite providers in the United States, Europe and Asia.

Terayon was incorporated in California in 1993 and reincorporated in Delaware in 1998.

#### Motorola, Inc.

1303 East Algonquin Road Schaumburg, Illinois 60196 (847) 576-5000

Motorola builds, markets and sells products, services and applications that make simple and seamless connections to people, information and entertainment possible through broadband, embedded systems and wireless networks. Motorola s vision is to provide cutting-edge technologies that empower mobile consumers to go anywhere and do anything without sacrificing connectivity. This is seamless mobility.

#### **Business Segments**

Motorola reports financial results for the following three operating business segments:

#### Mobile Devices Segment

The Mobile Devices segment designs, manufactures, sells and services wireless handsets with integrated software and accessory products, and licenses intellectual property. In 2006, the segment s net sales represented 66% of Motorola s consolidated net sales.

#### Networks and Enterprise Segment

The Networks and Enterprise segment designs, manufactures, sells, installs and services: (i) cellular infrastructure systems and wireless broadband systems to public carriers and other wireless service providers (referred to as the

public networks market), and (ii) analog and digital two-way radio, voice and data communications products and systems, as well as wireless broadband systems, to a wide range of public safety, government, utility, transportation and other worldwide enterprise markets (referred to as the private networks market). In January 2007, the segment completed the acquisition of Symbol Technologies, Inc. (Symbol), a leader in providing products and systems used in end-to-end enterprise mobility solutions. Symbol will become the cornerstone of the segment s enterprise mobility strategy. In 2006, the segment s net sales represented 26% of Motorola s consolidated net sales.

### **Connected Home Solutions Segment**

The Connected Home Solutions segment designs, manufactures, sells and services: (i) cable television, Internet Protocol (IP) video and broadcast network set-top boxes (digital entertainment devices), (ii) end-to-end digital video system solutions, (iii) broadband access networks, and (iv) IP-based data and voice products (including modems). In 2006, the segment s net sales represented 8% of Motorola s consolidated net sales.

#### **Table of Contents**

Motorola is a corporation organized under the laws of the State of Delaware as the successor to an Illinois corporation organized in 1928. Motorola s principal executive offices are located at 1303 East Algonquin Road, Schaumburg, Illinois 60196.

# Motorola GTG Subsidiary VI Corp.

1303 East Algonquin Road Schaumburg, Illinois 60196 (847) 576-5000

Merger Sub is a Delaware corporation and a wholly owned subsidiary of Parent. Merger Sub was organized solely for the purpose of completing a merger. It has not conducted any activities to date other than activities incidental to its formation and in connection with the transactions contemplated by the merger agreement.

# THE SPECIAL MEETING

We are furnishing this proxy statement to our stockholders, as of May 18, 2007, the record date, as part of the solicitation of proxies by our board of directors for use at the special meeting.

#### Date, Time and Place

The special meeting of our stockholders will be held at our corporate headquarters located at 2450 Walsh Avenue, Santa Clara, California 95051, at 11:00 a.m., local time, on June 28, 2007.

#### **Purpose of the Special Meeting**

At the special meeting, we are asking holders of record of our common stock on May 18, 2007 to consider and vote on the following proposals:

- The adoption of the Agreement and Plan of Merger, dated as of April 21, 2007, by and among Terayon Communication Systems, Inc., Motorola, Inc. and Motorola GTG Subsidiary VI Corp., and the approval of the merger of Motorola GTG Subsidiary VI Corp. with and into Terayon Communication Systems, Inc., in connection with which Terayon Communication Systems, Inc. will become a wholly owned subsidiary of Motorola, Inc. and each outstanding share of the common stock of Terayon Communication Systems, Inc. will be converted into the right to receive an amount equal to \$1.80 in cash, without interest;
- 2. The approval of the adjournment of the special meeting, if necessary, to solicit additional proxies if there are insufficient votes at the time of the special meeting to adopt the merger agreement and approve the merger; and
- 3. The transaction of any other business as may properly come before the special meeting or any adjournment or postponement of the special meeting.

#### **Recommendation of our Board of Directors**

Our board of directors has unanimously approved the merger agreement and determined that it is advisable, fair to and in the best interests of, our stockholders, and unanimously recommends that our stockholders vote FOR the adoption of the merger agreement and approval of the merger and FOR any proposal to adjourn the special meeting to solicit additional proxies, if necessary.

#### **Record Date; Shares Entitled to Vote; Quorum**

Only holders of record of shares of our common stock at the close of business on May 18, 2007, the record date, are entitled to notice of and to vote at the special meeting. On the record date, 77,637,177 shares of our common stock were issued and outstanding and held by approximately 501 holders of record. A quorum is present at the special meeting if a majority of all the shares of our common stock issued and outstanding on the record date are represented at the special meeting in person or by a duly authorized and properly completed proxy. Abstentions and broker non-votes, discussed below, count as present for establishing a quorum. Holders of record of shares of our common stock on the record date are entitled to one vote per share on each matter submitted to a vote at the special meeting.

#### **Vote Required**

The adoption of the merger agreement and approval of the merger requires the affirmative vote of stockholders holding a majority of the shares of our common stock outstanding on the record date. Because the required vote of our stockholders is based upon the number of outstanding shares of our common stock, rather than upon the shares actually voted, the failure by the holder of any such shares to submit a proxy or to vote via telephone or the Internet or in person at the special meeting, including abstentions and broker non-votes (described under Voting of Proxies below), will have the same effect as a vote against the adoption of the merger agreement and approval of the merger.

The proposal to adjourn the special meeting for the purpose of soliciting additional proxies, if necessary, will be approved if the votes cast in favor of the proposal by shares of our common stock, present in person or

represented by proxy and entitled to vote on the subject matter, exceed the votes cast against the proposal. The failure to submit a proxy or to vote, including abstentions and broker non-votes, will have no effect on the proposal to adjourn the special meeting.

#### Shares Owned by Our Directors and Executive Officers

At the close of business on May 18, 2007, the record date, our directors and executive officers beneficially owned and were entitled to vote approximately 7.8% percent of the shares of our common stock outstanding on that date.

#### **Voting of Proxies**

We are offering you four methods of voting:

by completing, signing, dating and returning the enclosed proxy card in the accompanying pre-addressed, postage paid envelope;

by calling the toll-free number that appears on the enclosed proxy card and following the instructions given;

by going to the Internet website that appears on the enclosed proxy card and following the instructions given; or

by appearing and voting in person by ballot at the special meeting.

Brokers or banks who hold shares of our common stock in street name for customers who are the beneficial owners of such shares may not submit a proxy to vote those customers shares in the absence of specific instructions from those customers. The brokers and banks will provide their customers with directions on how to instruct the broker or bank to vote their shares. In addition, a large number of brokers and banks participate in the ADP Investor Communication Services online program. This program provides eligible stockholders who receive a paper copy of this proxy statement the opportunity to vote via the Internet or by telephone. If your broker or bank participates in ADP s program, your broker or bank will provide instructions.

If no instructions are given to the broker or bank holding shares, or if instructions are given to the broker or bank indicating that the broker or bank does not have authority to vote on the proposal to adopt the merger agreement and approve the merger, then, in either case, a broker non-vote will generally occur and the shares will be counted as present for purposes of determining whether a quorum exists, but will not be voted on the proposal to adopt the merger agreement and approve the merger and will therefore have the same effect as voting against the adoption of the merger agreement and approval of the merger. Similarly, broker non-votes will not be voted on the proposal to adopt to adjourn the special meeting to solicit additional proxies, but will have no effect on that proposal. Brokers and other nominees will not have discretionary authority to vote on the proposal to adopt the merger agreement and approve the merger or the proposal to adopt the merger or the proposal to adjourn the special meeting to solicit additional proxies, but will have no effect on that proposal. Brokers and other nominees will not have discretionary authority to vote on the proposal to adopt the merger agreement and approve the merger or the proposal to adjourn the special meeting to solicit additional proxies, if necessary.

All shares represented by properly completed proxies received prior to the special meeting will be voted at the special meeting in the manner specified in the proxies. Properly completed proxies that do not contain voting instructions will be voted FOR the adoption of the merger agreement and approval of the merger and FOR the proposal to adjourn the special meeting to solicit additional proxies, if necessary, provided that no proxy that is specifically marked

AGAINST the proposal to adopt the merger agreement and approve the merger will be voted in favor of the adjournment proposal, unless it is specifically marked FOR the adjournment proposal.

Shares of our common stock represented at the special meeting but not voting, including shares of our common stock for which proxies have been received but with respect to which holders have abstained, will be treated as present at the special meeting for purposes of determining whether a quorum exists, but will not be voted on the proposal to adopt the merger agreement and approve the merger, and will effectively count as votes against the adoption of the merger agreement and the approval of the merger. Similarly, shares of our common stock for which proxies have been received but with respect to which holders have abstained will not be voted on the proposal to adjourn the special meeting to solicit additional proxies, but will have no effect on that proposal.

#### Table of Contents

Although it is not currently expected, if the proposal to adjourn the special meeting to solicit additional proxies is approved, the special meeting may be adjourned for the purpose of soliciting additional proxies to approve the proposal to adopt the merger agreement and approve the merger. Other than for the purposes of adjournment to solicit additional proxies, if a quorum exists, then the chairman or a vote by a majority of the shares casting votes, excluding abstentions, at the special meeting may adjourn the meeting. Alternatively, if no quorum exists, then the chairman or a majority of the shares represented at the special meeting may adjourn the meeting.

Any adjournment may be made without notice, other than by an announcement made at the special meeting, unless the adjournment is for more than thirty days or a new record date is fixed for the adjourned meeting. Any adjournment of the special meeting for the purpose of soliciting additional proxies will allow our shareholders who have already sent in their proxies to revoke them at any time prior to their use at the special meeting as adjourned.

At any time prior to convening the special meeting, our board of directors may postpone the special meeting for any reason without the approval of our stockholders. If postponed, we will provide at least ten days notice of the new meeting date. Although it is not currently expected, our board of directors may postpone the special meeting for the purposes of soliciting additional proxies if it concludes that by the meeting date it is reasonably likely that we will not have received sufficient proxies to constitute a quorum or sufficient votes to approve the proposal to adopt the merger agreement and approve the merger. Similar to adjournments, any postponement of the special meeting for the purpose of soliciting additional proxies will allow shareholders who have already sent in their proxies to revoke them at any time prior to their use. If the special meeting is adjourned or postponed and the record date remains unchanged, unrevoked proxies will continue to be effective for purposes of voting on the new meeting date.

#### **Revocability of Proxies**

You can change your vote or revoke your proxy at any time before the proxy is voted at the special meeting. You may accomplish this in one of four ways:

by providing a written instrument or transmission to our corporate secretary at our corporate headquarters prior to the special meeting stating that you revoke your proxy;

by completing and submitting to our corporate secretary a proxy in writing via mail dated later than your original proxy relating to the same shares;

by voting via the Internet or by telephone following the date of your original proxy relating to the same shares; or

by attending the special meeting and voting in person, which will automatically cancel any proxy previously given; your attendance at the special meeting alone, however, will not revoke any proxy that you have previously given.

If you instructed your broker to vote your shares, you must follow directions from your broker to change these instructions.

#### **Solicitation of Proxies**

We will pay the costs of the solicitation of proxies from our stockholders. In addition to solicitation by mail, our directors, officers and employees may solicit proxies from stockholders by telephone or other electronic means or in person. These persons will not receive additional or special compensation for such solicitation services. We will cause brokerage houses and other custodians, nominees and fiduciaries to forward solicitation materials to the beneficial

owners of stock held of record by such persons. We will, upon request, reimburse such custodians, nominees and fiduciaries for their reasonable out-of-pocket expenses in doing so. MacKenzie Partners, Inc. will assist in our solicitation of proxies. We will pay MacKenzie Partners a fee of \$20,000 plus reimbursement of certain out-of-pocket expenses, and will indemnify MacKenzie Partners against any losses arising out of its proxy solicitation services on our behalf.

#### Table of Contents

#### **Other Business**

We do not expect that any matter other than the proposal to adopt the merger agreement and approve the merger and, if necessary, the proposal to adjourn the meeting will be brought before the special meeting. If, however, other matters are properly presented at the special meeting, the persons named as proxies will vote in accordance with their best judgment with respect to those matters unless otherwise indicated on the proxy.

#### Assistance

If you need assistance in completing your proxy card or voting your shares, or have questions regarding the special meeting, please contact MacKenzie Partners, which is assisting us in the solicitation of proxies, at 800-322-2885 (toll-free) or 212-929-5500 (collect), or write to the following address:

MacKenzie Partners 105 Madison Avenue New York, NY 10016

*YOU SHOULD NOT SEND STOCK CERTIFICATES WITH YOUR PROXIES.* If the merger is approved by our stockholders and ultimately consummated, a transmittal form with instructions for the surrender of certificates formerly representing shares of our common stock will be mailed to you shortly after the effective time of the merger.

## THE MERGER

While we believe that the following description covers the material terms of the merger and the related transactions, this summary may not contain all of the information that is important to you. You should carefully read this entire document, including the annexes, and the other documents we refer to for a more complete understanding of the merger and the related transactions.

## **Background of the Merger**

In early 2005, we were nearing completion of our transition to a video company and believed there was a large market opportunity for our video products if we successfully executed on our business plan over the following few years. However, other companies were also starting to develop technology similar to ours with remultiplexing, splicing and ad insertion capabilities. We were faced with the decision between trying to execute on our business plan as a stand-alone company in an industry that would become increasingly competitive, or seeking to maximize value for our stockholders through a strategic transaction, including a potential sale of the company. As a result, towards the middle of 2005, our board of directors began exploring the possibility of a strategic transaction, and engaged Goldman Sachs as financial advisor with respect to the process. Between May 2005 and September 2005, Goldman Sachs contacted 15 parties, including both strategic and financial parties, with respect to a potential strategic transaction, and we entered into confidentiality agreements with seven of them. While Motorola was one of the parties contacted, Motorola declined to pursue a strategic transaction with us at that time based on strategic and valuation considerations. By October 2005, we were in substantive discussions with two of the interested parties, and the other five parties with whom we entered into confidentiality agreements had declined to pursue a strategic transaction further. During the negotiation process, we became aware of potential revenue recognition issues relating to our financial statements, and on November 7, 2005, we issued a press release announcing that our Audit Committee would conduct an investigation into these matters. At a regular meeting of our board of directors held on November 30, 2005, our board decided to place the strategic transaction process on hold until the Audit Committee completed its investigation.

During the first quarter of 2006, our Audit Committee concluded its investigation. On March 1, 2006, the Audit Committee concluded that certain previously issued financial statements should no longer be relied upon and would be restated. At meetings of our board of directors in March and May 2006, after considering the advice of Goldman Sachs and our legal advisors, Latham & Watkins LLP, our board of directors instructed management not to reinitiate discussions with the two interested parties, or any other potential parties, with respect to a strategic transaction at that time.

At a special meeting of our board of directors on October 9, 2006, our board reviewed with representatives of Goldman Sachs our strategic options upon the completion of the restatement process, including remaining as a stand-alone company, and reviewed potential interested parties, including Motorola, and a revised strategy and timing for the strategic transaction process. At the special meeting, representatives of Goldman Sachs advised our board of directors that it is providing and has provided certain investment banking services to some of the potential interested parties, including Motorola. Following discussion, the board directed management to reengage in the strategic transaction process.

During October and November 2006, management and Goldman Sachs contacted the two parties with which we had been engaged in substantive discussions in October 2005, four other parties which we had contacted in 2005, including Motorola, and one additional party. Two of the parties subsequently stated they were not interested in pursuing a strategic transaction. After we had entered into any necessary confidentiality agreements, members of our

management gave presentations to the remaining five parties, including Motorola in Santa Clara, California on October 26, 2006, and including the two parties with which we were in substantive discussions in October 2005. The parties that had agreed to standstill provisions in their confidentiality agreements were also provided access to our online data room at that time.

On November 10, 2006, Goldman Sachs sent bid procedures letters on our behalf to the five remaining prospective bidders indicating that initial non-binding proposals would be due on December 1, 2006. The deadline was based on the assumption that we would have filed restated financial statements prior to that time.

#### Table of Contents

On November 29, 2006, Darren Marino, Vice President of Business Development of the Connected Home Solutions Business of Motorola, contacted representatives of Goldman Sachs to request an extension of the bid deadline and to convey a list of financial and technical due diligence questions for us. Mr. Marino was informed that an extension was possible, but that due diligence materials would not be made available unless Motorola agreed to an amendment to its confidentiality agreement with us that contained standstill and non-solicitation of employee provisions. Following negotiations, we entered into the amendment with Motorola on December 4<sup>th</sup>, and then made the requested information available to it. Given that we had not filed restated financial statements, the other potential bidders were also informed that they should not view December 1<sup>st</sup> as a hard deadline for submitting non-binding proposals. None of the parties submitted a proposal prior to the deadline.

At a December 4<sup>th</sup> board meeting, representatives from Goldman Sachs updated our board on the status of discussions with interested parties, and the fact that none of the interested parties had submitted a proposal as of that date.

During early December, Mr. Marino informed representatives of Goldman Sachs that Motorola expected to deliver a non-binding proposal around the middle of December. At the request of our management, following discussions with individual members of our board of directors, Goldman Sachs contacted three of the other four interested parties and informed them that we might be receiving a preemptive offer from one of the interested parties and requesting that they forward a non-binding proposal as soon as possible in order to be considered in the process. The fourth party was not contacted because it had communicated to Goldman Sachs by that time that it was not interested in pursuing a transaction.

On December 21, 2006, Motorola submitted, in writing, a non-binding all cash proposal to acquire us at a price per share of between \$1.60 and \$1.75, subject to due diligence, negotiation of key employee arrangements and the negotiation of a definitive merger agreement. Motorola also requested a 30-day exclusivity period. At a special meeting of our board of directors the next day, Goldman Sachs updated our board on the status of discussions with interested parties, including Motorola s proposal and the fact that none of the other bidders had submitted a proposal, as well as other strategic alternatives available to the company. With regard to the absence of other bidders, Goldman Sachs informed our board that some of the bidders had expressed concern about bidding because their valuation of the company was significantly below one that was likely to be acceptable to our board, given the current trading price of our shares, which was \$1.63 as of the close of market on December 21<sup>st</sup>. The board directed Goldman Sachs to inform Motorola that it was prepared to continue discussions with Motorola, but that if Motorola wanted exclusivity, it would have to increase its proposed transaction price above \$1.75 per share. Motorola subsequently rescinded its request for exclusivity and confirmed orally a non-binding all cash proposal to acquire us at a price per share no greater than \$1.75, subject to the conditions described above.

On December 29<sup>th</sup>, we filed our Form 10-K for the year ended December 31, 2005, which included the restatement of our financial statements for various prior periods. We filed our Forms 10-Q for each of the first three quarters of 2006 on January 10, 2007.

On December 29, 2006, representatives of Winston & Strawn LLP, Motorola s legal counsel, forwarded us an initial draft merger agreement. Management and the legal and financial advisors of the two companies were in communication regarding the terms of a definitive merger agreement from this time through the signing of the merger agreement, and conducted extensive negotiations regarding the terms through a series of telephonic conference calls and email exchanges during that period. Motorola also continued its due diligence investigation throughout this time.

During January 2007, representatives of Latham & Watkins had several calls with representatives of Winston & Strawn to discuss issues regarding the draft merger agreement. Winston & Strawn circulated a revised draft of the merger agreement on January 19, 2007. During the course of January, several due diligence meetings and conference

calls took place between members of our management and advisors and members of Motorola s deal team to discuss due diligence and transaction planning matters.

At a regularly scheduled meeting of our board of directors held on January 31, 2007, members of management and Goldman Sachs updated the board on the status of discussions with Motorola and the lack of developments with any other party. Representatives of Goldman Sachs discussed with the board financial aspects of Motorola s

proposal. Members of management also updated the board on fourth quarter financial information. The board directed management to continue negotiations with Motorola.

Throughout February, we and our advisors held a number of conference calls with Motorola and its advisors regarding due diligence matters. Further drafts of the merger agreement were exchanged towards the middle of the month and at the beginning of March.

At a special meeting of our board of directors on March 2, 2007, representatives of Goldman Sachs and members of our management updated the board on the status of negotiations with Motorola, including Motorola s due diligence review, and the expected timing of completing a strategic transaction. Representatives of Goldman Sachs discussed with the board financial aspects of the proposed transaction, the status of discussions with other parties and whether, given the passage of time since reengagement in the strategic transaction process and the possibility of changed circumstances, there might be any parties who had previously informed us they were not interested in acquiring us or whom we previously considered unlikely to have an interest, but who now might be interested in acquiring us. One of the companies with which we had been in substantive discussions in 2005 had experienced a setback with regard to a major strategic initiative, and as a result Goldman Sachs stated that in its view it was worth going back to that company to see whether it had a renewed interest in acquiring us. In addition, a company that we had not previously contacted had recently publicly expressed interest in our sector, and Goldman Sachs stated that in its view it was worth approaching that company. After discussing these views, the belief of members of the board and representatives of Goldman Sachs that a broad search for potential acquirers had already been undertaken, and the potential impact that expanding that search would have on discussions with Motorola, our board of directors directed management and Goldman Sachs to approach these two companies and one additional company with which we had been in substantive discussions in 2005.

During March, Motorola sought a number of internal approvals to move forward with the transaction. Motorola had substantially completed its due diligence by that time, although it remained focused on a few areas of potential risk to Motorola, including with respect to outstanding litigation matters that we face. We and our legal and other advisors had a few calls with Motorola and its legal and other advisors regarding these matters. In addition, Motorola was waiting to receive our financial statements as of December 31, 2006. For the foregoing reasons, negotiation of the merger agreement slowed down.

At informal meetings during early March, our board of directors received an update from management and representatives of Latham & Watkins as to the status of discussions with Motorola, including material outstanding issues, as well as our alternatives as a stand-alone company. Members of the board directed management to continue negotiations with Motorola.

During the middle of March, Jerry Chase, our Chief Executive Officer, and representatives of Goldman Sachs contacted two of the additional companies, as directed by the board, and requested that they submit an expression of interest by the middle of the following week if they were interested in pursuing a potential transaction. Representatives of Goldman Sachs contacted the third additional company, as directed by the board, requesting that it submit an expression of interest in the same timeframe. None of the companies forwarded an expression of interest.

Towards the end of the month, representatives of Latham & Watkins discussed the material outstanding issues on the merger agreement with representatives of Winston & Strawn. Among the issues that remained outstanding after the call were the merger consideration, the amount of the break-up fee, the existence of a condition relating to the maximum percentage of shares for which appraisal rights could be exercised, a condition relating to the status of a pending investigation by the SEC, various covenants and conditions relating to employees and various representations, covenants and conditions relating to intellectual property and tax matters.

At a special meeting of our board of directors held on April 5<sup>th</sup>, management and representatives of Goldman Sachs updated the board on the status of discussions with Motorola and other parties. Representatives of Goldman Sachs stated that Motorola indicated that it had received some, but not yet all, of its required internal approvals. Management and Goldman Sachs advised the board that none of the three companies contacted in March had expressed an interest in acquiring us. A representative of Latham & Watkins provided a detailed summary of the material terms of the merger agreement and material open issues, including Motorola s request for a break-up fee of 3.5% of equity value plus unlimited expense reimbursement. Following consideration by the board of information

#### Table of Contents

relating to break-up fees in other transactions, including transactions where a third party had publicly announced a higher competing offer after the transaction was announced, the board directed Latham & Watkins not to agree to a break-up fee in excess of 3.75% of equity value with no expense reimbursement, and to continue negotiating the other outstanding issues along the lines outlined. Representatives of Goldman Sachs then discussed with the board financial aspects of the proposed transaction, including the possibility of Motorola increasing its offer price above \$1.75 per share. The board directed Goldman Sachs to continue attempts to negotiate a price per share in excess of \$1.75.

Later that day, a representative of Goldman Sachs spoke with Motorola s Senior Vice President of Mergers and Acquisitions. Motorola s officer indicated that he was reviewing the proposed strategic transaction, but had not yet approved it. The Goldman Sachs representative informed him that our board of directors had expressed significant reservations as to whether it would agree to a transaction unless the price per share were above \$1.75. The parties also discussed various due diligence matters, including litigation matters, of continued concern to Motorola.

From April 10<sup>th</sup> through the signing date, the parties and their advisors held numerous calls to resolve outstanding issues. On April 17<sup>th</sup>, Mr. Marino of Motorola informed a representative of Goldman Sachs that all internal Motorola approvals had been obtained to acquire us at a price no greater than \$1.80 per share, which Motorola informed us they viewed as a significant concession given the litigation risk and other risks it would be assuming in the transaction. During the afternoon of April 19<sup>th</sup>, Motorola provided us with a list of the remaining open items with respect to the merger agreement, and outlined a proposal for resolving the remaining open issues. Among the material terms that Motorola accepted were a break-up fee equal to 3.75% of the equity value of the deal with no expense reimbursement, and the removal of a closing condition relating to appraisal rights.

From April 16th through April 19th, four of our key employees completed negotiations of their respective offer letters with Motorola, as more fully described below in Interests of Our Directors and Executive Officers in the Merger Offer Letters with Motorola.

Members of our management and representatives of Goldman Sachs and Latham & Watkins reviewed and discussed Motorola s April 19 proposal, and responded with a counterproposal for resolving the remaining open issues early in the morning of April 20<sup>th</sup>. Motorola responded to our counterproposal shortly before noon on April 20<sup>th</sup>, and significantly narrowed the list of remaining open issues. Our board of directors met at 2:00 p.m. PDT on April 20<sup>th</sup> to discuss with management and our legal and financial advisors the status of negotiations with Motorola and provide input on the remaining outstanding issues. The board directed management and Latham & Watkins to continue to negotiate various tax and employment conditions in order to minimize the pre-closing risk to us that the transaction would not be completed. Following that meeting, our legal counsel continued to work with Motorola s legal counsel throughout the evening to finalize the merger agreement with Motorola.

The parties continued their negotiations that evening and the following day. During the afternoon of April 21<sup>st</sup>, all outstanding terms of the merger agreement were agreed upon, subject in our case to board approval. Our board of directors met at 6:00 p.m. PDT on April 21<sup>st</sup> to consider the proposed transaction. At the meeting, members of management updated the board on the conclusion of negotiations with Motorola. A representative of Latham & Watkins provided the board with a detailed summary of material terms in the merger agreement. Representatives of Goldman Sachs reviewed with the board its financial analysis with respect to the proposed transaction with Motorola, and delivered its oral opinion, subsequently confirmed by delivery of its written opinion, that as of April 21, 2007, and based upon and subject to the factors and assumptions set forth in its written opinion, the \$1.80 per share of common stock in cash to be received by the holders of shares of our common stock pursuant to the merger was fair from a financial point of view to such holders. Based on these considerations and its prior deliberations, the board of directors, by unanimous vote, approved and declared advisable the merger agreement, determined that the merger agreement and the merger were fair to, and in the best interests of, us and our stockholders, directed that the adoption of the merger agreement be submitted to a vote of our stockholders and unanimously recommended that our

stockholders adopt the merger agreement and approve the merger. Shortly thereafter, the parties executed the definitive merger agreement.

Prior to the opening of the New York Stock Exchange on April 23, 2007, we issued a joint press release with Motorola publicly announcing the execution of the merger agreement.

#### **Reasons for the Merger**

Our board of directors, at a special meeting held on April 21, 2007, unanimously determined that the merger agreement is advisable, fair to and in the best interests of our stockholders and unanimously approved the merger agreement and the merger. Accordingly, the board of directors unanimously recommends that you vote FOR adoption of the merger agreement and approval of the merger at the special meeting.

In the course of determining that the merger agreement is advisable, fair to and in the best interests of our stockholders, the board of directors consulted with management, as well as its financial and legal advisors, and considered the following factors:

the familiarity of the members of the board of directors with, and information provided by management as to, our business, financial condition and results of operations, including the fact that we have not been profitable in any quarterly period other than the quarter ended June 30, 2006, our competitive position, and the nature of our business and the industry in which we compete;

the opinion of Goldman, Sachs & Co. that, as of April 21, 2007, and based upon and subject to the factors and assumptions set forth in its opinion, the \$1.80 per share of common stock in cash to be received by the holders of shares of our common stock pursuant to the merger agreement was fair from a financial point of view to such holders;

the belief of the board of directors that \$1.80 per share represented the highest consideration that Motorola was willing to pay, and the highest per share value obtainable on the date of signing;

the fact the price per share of our common stock had increased significantly on the day prior to announcement of the acquisition, and the closing price of \$1.84 per share on that day represented a 24.3% increase over the closing price per share on the second trading day prior to announcement;

the familiarity of the members of the board of directors with, and information provided by management and Goldman, Sachs & Co., our financial advisor, as to, economic and market conditions, on both a historical and a prospective basis;

the execution risks associated with our management s financial projections provided to the board of directors, as described in Financial Projections, including the fact that the market and earnings per share growth targets were not reflective of our most recent historical results;

our prior discussions with respect to possible business combinations and the scope and outcome of the process conducted to identify parties having a potential interest and ability to acquire us, as described in Background of the Merger;

the impact on us, our business and employees of our restatement and delisting from Nasdaq;

other strategic alternatives available to us, including the possibility of remaining an independent company if we were unable to sell ourselves;

the ability of our stockholders to exercise appraisal rights in connection with the merger, as described in Appraisal Rights;

the fact that the merger consideration is all cash, which provides certainty of value to our stockholders, compared to a transaction involving all stock consideration or a mixture of stock and cash; and

the terms and conditions of the merger agreement, including:

that the merger agreement does not include a financing condition and our board s conclusion, after consultation with Goldman, Sachs & Co., our financial advisor, that Motorola has the financial ability to complete the merger;

our ability to furnish information to and conduct negotiations with third parties, and to terminate the merger agreement if a third party makes a superior proposal for a business combination or acquisition,

#### **Table of Contents**

as described in The Merger Agreement No Solicitation of Other Acquisition Proposals by Terayon; and

the nature and scope of the closing conditions, including the likelihood that the merger would receive the necessary regulatory approvals and the anticipated timing of those approvals.

Our board of directors also considered potentially negative factors in its deliberations concerning the merger, including:

that we will no longer exist as an independent company and our stockholders will no longer participate in our growth as an independent company;

the fact that the cash merger consideration of \$1.80 per share is lower than the closing price of \$1.84 per share of our common stock on April 20, 2007, the last trading day immediately prior to the public announcement that we signed the merger agreement, and represents a discount of approximately 2.2 percent;

that, under the terms of the merger agreement, we will be unable to solicit other acquisition proposals;

that, under the terms of the merger agreement, we will be required to pay Motorola a termination fee if we terminate the merger agreement to accept a superior proposal for a business combination or acquisition of us, and under a number of other circumstances associated with proposals by third parties to acquire us, and that our obligation to pay the termination fee might discourage other parties from proposing a business combination with, or an acquisition of, us (see The Merger Agreement Termination; Payment of Termination Fee );

that gains from an all-cash transaction will be taxable to our stockholders for U.S. federal income tax purposes;

the interests of our executive officers and directors in the merger described under Interests of Our Directors and Executive Officers in the Merger;

that, while the merger is expected to be completed, there can be no assurance that all conditions to the parties obligations to complete the merger will be satisfied, and as a result, it is possible that the merger may not be completed even if approved by our stockholders (see The Merger Agreement Conditions to the Completion of the Merger );

the restrictions on the conduct of our business prior to the completion of the merger, requiring us to conduct our business in the ordinary course, subject to specific limitations, which may delay or prevent us from undertaking business opportunities that may arise pending completion of the merger; and

the possibility of disruption to our operations following announcement of the merger, and the resulting effect on us if the merger does not close, including the diversion of management and employee attention, potential employee attrition and the potential effect on business and customer relationships.

While the board of directors considered potentially negative and potentially positive factors, the board of directors concluded that overall, the potentially positive factors outweighed the potentially negative factors.

The foregoing discussion of the factors considered by our board of directors is not intended to be exhaustive, but rather includes the material information and factors considered by our board of directors in its consideration of the merger. The board of directors collectively reached the unanimous decision to approve the merger agreement in light of the factors described above and other factors that each member of the board of directors felt were appropriate. In view of the variety of factors, many of which are qualitative or difficult to quantify, and the quality and amount of

information considered, the board of directors did not find it practicable to and did not make specific assessments of, quantify or otherwise assign relative weights to the specific factors considered in reaching its determination. Individual members of the board of directors may have given different weight to different factors.

#### **Recommendation of Our Board of Directors**

After careful consideration, our board of directors, by unanimous vote, has determined that the merger, the merger agreement and the transactions contemplated by the merger agreement are advisable, fair to and in the best

interests of Terayon and our stockholders, has approved the merger, the merger agreement and the transactions contemplated by the merger agreement, and recommends that Terayon s stockholders vote FOR the adoption of the merger agreement and approval of the merger.

#### **Opinion of Our Financial Advisor**

Goldman Sachs rendered its opinion to our board of directors that, as of April 21, 2007, and based upon and subject to the factors and assumptions set forth therein, the \$1.80 per share of common stock in cash to be received by the holders of shares of our common stock in the transaction pursuant to the merger agreement was fair from a financial point of view to such holders.

The full text of the written opinion of Goldman Sachs, dated April 21, 2007, which sets forth assumptions made, procedures followed, matters considered and limitations on the review undertaken in connection with the opinion, is attached as Annex B to this proxy statement. Goldman Sachs provided its opinion for the information of and assistance of our board of directors in connection with its consideration of the transaction. The Goldman Sachs opinion is not a recommendation as to how any holder of shares of our common stock should vote with respect to the transaction.

In connection with rendering the opinion described above and performing its related financial analyses, Goldman Sachs reviewed, among other things:

the merger agreement;

our Annual Reports on Form 10-K for the five years ended December 31, 2006 and our amended Annual Report on Form 10-K/A for the year ended December 31, 2004;

certain of our interim reports to stockholders and Quarterly Reports on Form 10-Q;

certain other communications from us to our stockholders; and

certain internal financial analyses and forecasts for us prepared by our management. See Financial Projections, below.

Goldman Sachs also held discussions with members of our senior management regarding their assessment of our past and current business operations, financial condition and future prospects. In addition, Goldman Sachs reviewed the reported price and trading activity for the shares of our common stock, compared certain financial and stock market information for us with similar information for certain other companies the securities of which are publicly traded, reviewed the financial terms of certain recent business combinations in the communications technology industry specifically and in other industries generally and performed such other studies and analyses, and considered such other factors, as it considered appropriate.

Goldman Sachs relied upon the accuracy and completeness of all of the financial, accounting, legal, tax and other information discussed with or reviewed by it and assumed such accuracy and completeness for purposes of rendering the opinion described above. In that regard, Goldman Sachs assumed with the consent of our board of directors that the internal financial forecasts prepared by our management had been reasonably prepared on a basis reflecting our best currently available estimates and judgments. In addition, Goldman Sachs did not make an independent evaluation or appraisal of the assets and liabilities (including any contingent, derivative or off-balance-sheet assets and liabilities) of us or any of our subsidiaries, nor was any such evaluation or appraisal furnished to Goldman Sachs.

Goldman Sachs opinion does not address our underlying business decision to engage in the transactions contemplated by the merger agreement. Goldman Sachs opinion is necessarily based on economic, monetary, market and other conditions as in effect on, and the information made available to Goldman Sachs as of, the date of its opinion.

The following is a summary of the material financial analyses delivered by Goldman Sachs to our board of directors in connection with rendering the opinion described above. The following summary, however, does not purport to be a complete description of the financial analyses performed by Goldman Sachs, nor does the order of analyses described represent relative importance or weight given to those analyses by Goldman Sachs. Some of the

#### Table of Contents

summaries of the financial analyses include information presented in tabular format. The tables must be read together with the full text of each summary and are alone not a complete description of Goldman Sachs financial analyses. Except as otherwise noted, the following quantitative information, to the extent that it is based on market data, is based on market data as it existed on or before April 21, 2007, and is not necessarily indicative of current market conditions.

*Discounted Cash Flow Analysis.* Goldman Sachs performed a discounted cash flow analysis based on forecasts by our management to determine a range of implied present values per share of our common stock. All unlevered cash flows were discounted to March 31, 2007, and illustrative terminal values were based upon perpetuity growth rates ranging from 3.0% to 7.0% for unlevered free cash flows for years 2012 and beyond. In performing the discounted cash flow analysis, Goldman Sachs applied discount rates ranging from 13.0% to 17.0% to our projected unlevered free cash flows for calendar years 2007 to 2011. This analysis resulted in a range of implied present values of \$0.64 to \$1.26 per share of our common stock.

*Selected Companies Analysis.* Goldman Sachs reviewed and compared certain financial information for us to corresponding financial information, ratios and public market multiples for the following publicly traded corporations in the large capitalization communications technology infrastructure, encoding and software driven video technology and hardware driven video technology industries.

Large Capitalization	
Communications	<b>Encoding and Software Driven</b>
Technology Infrastructure	Video Technology

Harmonic, Inc. Kudelski SA Macrovision Corporation

## Hardware Driven Video Technology

ARRIS Group, Inc. C-COR Incorporated Pace Micro Technology plc Westell Technologies, Inc.

Although none of the selected companies is directly comparable to us, the companies included were chosen because they are publicly traded companies with operations that for purposes of analysis may be considered similar to certain of our operations.

Goldman Sachs calculated and compared the enterprise value to estimated 2007 and 2008 revenue and the price per share to estimated 2007 earnings per share and estimated 2008 earnings per share of us and the selected companies listed above based on financial data and shares prices as of April 20, 2007, information obtained from SEC filings, estimates provided by the Institutional Brokers Estimate System (a data service that compiles estimates issued by securities analysts), or IBES, for the selected companies and us and information and forecasts for us provided by our management.

The results of these analyses are summarized as follows:

#### **Company**

Estimated

2008

Alcatel-Lucent

Motorola, Inc. Thomson

Cisco Systems, Inc.

Harris Corporation

Juniper Networks, Inc.

	Enterprise Value/Revenue Multiples	Enterprise Value/Revenue Multiples	Estimated 2007 Price/Earnings Multiples	Estimated 2008 Price/Earnings Multiples
Terayon (Management)	1.9x(2)	1.8x(2)	<b>NM</b> (1)	105.6x
Terayon (IBES Median)	2.3(2)	1.7(2)	<b>NM</b> (1)	30.7
	27			

# Large Capitalization Communications Technology Infrastructure

	Estimated 2007	Estimated 2008		
Selected Company	Enterprise Value/Revenue Multiples	Enterprise Value/Revenue Multiples	Estimated 2007 Price/Earnings Multiples	Estimated 2008 Price/Earnings Multiples
Alcatel-Lucent	1.1x	1.1x	20.2x	13.1x
Cisco Systems, Inc.	4.4	3.8	19.0	16.4
Harris Corporation	1.7	1.5	17.0	15.3
Juniper Networks, Inc.	3.7	3.2	25.3	20.9
Motorola, Inc.	1.0	0.9	43.4	18.8
Thomson	0.9	0.8	14.4	11.3
Mean	2.1x	1.9x	23.2x	16.0x
Median	1.4	1.3	19.6	15.9

# **Encoding and Software Driven Video Technology**

	Estimated Estimated 2007 2008					
Selected Company	Enterprise Value/Revenue Multiples	Enterprise Value/Revenue Multiples	Estimated 2007 Price/Earnings Multiples	Estimated 2008 Price/Earnings Multiples		
Harmonic, Inc.	2.5x	2.4x	29.1x	20.3x		
Kudelski SA	2.2	1.9	23.2	19.0		
Macrovision Corporation	4.6	4.0	19.9	16.3		
Mean	3.1x	2.8x	24.1x	18.5x		
Median	2.5	2.4	23.2	19.0		

# Hardware Driven Video Technology

	Estimated 2007	Estimated 2008		
Selected Company	Enterprise Value/Revenue Multiples	ue/Revenue Value/Revenue		Estimated 2008 Price/Earnings Multiples
ARRIS Group, Inc.	1.4x	1.3x	18.0x	15.8x
C-COR Incorporated	2.0	1.7	22.5	18.6
Pace Micro Technology plc	0.5	0.5	<b>NM</b> (1)	17.2
Westell Technologies, Inc.	0.5	0.5	25.8	25.3

Mean	1.1x	1.0x	22.1x	19.2x
Median	1.0	0.9	22.5	17.9

- (1) Not meaningful (projected earnings negative for period).
- (2) Terayon IBES Median and Management 1-year forward revenue estimates represent video revenue estimates only.

*Illustrative Present Value of Hypothetical Future Share Price Analysis.* Goldman Sachs performed an illustrative present value of hypothetical future share price analysis, which is designed to provide an indication of the present value of a theoretical future value of a company s equity as a function of such company s (i) assumed future enterprise value to revenue multiple and (ii) assumed price to future earnings per share multiple. For this analysis, Goldman Sachs used the financial projections prepared by our management for 2009 and 2010 and used assumed enterprise value to revenue multiples and price to future earnings per share multiples based on 2008 enterprise value to revenue multiples and 2008 price to future earnings per share multiples of the selected companies listed above.

Goldman Sachs first calculated the implied values per share of common stock for each of 2009 and 2010 by applying illustrative enterprise value to revenue multiples of 1.0x to 2.0x to estimated revenue for each of 2009 and

#### **Table of Contents**

2010, adding our net cash balance as of March 31, 2007, discounting the 2009 and 2010 values back to March 31, 2007, using an illustrative discount rate of 15%. This analysis resulted in a range of implied present values of \$1.04 to \$2.00 per share of our common stock.

Goldman Sachs then calculated the implied values per share of common stock for each of 2009 and 2010 by applying illustrative price to earnings per share multiples of 15.0x to 25.0x to earnings per share estimates for each of 2009 and 2010 and discounting the 2009 and 2010 values back to March 31, 2007, using an illustrative discount rate of 15%. This analysis resulted in a range of implied present values of \$0.91 to \$1.86 per share of our common stock.

*Selected Transactions Analysis.* Goldman Sachs reviewed certain publicly available information relating to the following selected transactions in the communications technology industry:

Announcement Date	Acquiror/Target
Eshmany 26, 2007	LM Ericsson Telephone Company / Tandberg Television ASA
February 26, 2007	
January 15, 2007	ARRIS Group, Inc. / Tandberg Television ASA
December 22, 2006	Motorola, Inc. / Tut Systems, Inc.
August 22, 2006	Harmonic, Inc. / Entone Technologies, Inc. (Video Networking Software Business)
August 21, 2006	Cisco Systems, Inc. / Arroyo Video Solutions, Inc.
July 25, 2006	Motorola, Inc. / Broadbus Technologies, Inc.
February 8, 2006	Tandberg Television ASA / SkyStream Networks, Inc.
November 18, 2005	Cisco Systems, Inc. / Scientific-Atlanta, Inc.
October 17, 2005	Tandberg Television ASA / GoldPocket Interactive, Inc.
August 31, 2005	Harris Corporation / Leitch Technology Corporation
July 28, 2005	Thomson / PRN Corporation
January 6, 2005	Tandberg Television ASA / N2 Broadband, Inc.
October 20, 2004	C-COR Incorporated / nCUBE Corporation

While none of the companies that participated in the selected transactions are directly comparable to us, the companies that participated in the selected transactions are companies with operations that, for the purposes of analysis, may be considered similar to certain of our operations.

For each of the selected transactions, Goldman Sachs calculated and compared (i) total enterprise value as a multiple of the target company s IBES estimated one year forward revenue; and (ii) equity value as a multiple of the target company s IBES estimated one year forward net income.

The following table presents the results of this analysis:

	Mean	Median
Enterprise Value/1-Year Forward Revenue	2.3x	2.5x
Equity Value/1-Year Forward Net Income	21.9x	22.7x

#### Multiple

#### **Terayon (IBES Median)**

Enterprise Value/1-Year Forward Revenue	2.4x(2)
Equity Value/1-Year Forward Net Income	NM(1)
Equity Value/2-Year Forward Net Income	30.0x
<b>Terayon (Management)</b> Enterprise Value/1-Year Forward Revenue Equity Value/1-Year Forward Net Income Equity Value/2-Year Forward Net Income	2.0x(2) NM(1) 103.3x

- (1) Not meaningful (projected net income negative for period).
- (2) Terayon IBES Median and Management 1-year forward revenue estimates represent video revenue estimates only.

The preparation of a fairness opinion is a complex process and is not necessarily susceptible to partial analysis or summary description. Selecting portions of the analyses or of the summary set forth above, without considering the analyses as a whole, could create an incomplete view of the processes underlying Goldman Sachs opinion. In arriving at its fairness determination, Goldman Sachs considered the results of all of its analyses and did not attribute any particular weight to any factor or analysis considered by it. Rather, Goldman Sachs made its determination as to fairness on the basis of its experience and professional judgment after considering the results of all of its analyses. No company or transaction used in the above analyses as a comparison is directly comparable to us or the contemplated merger.

Goldman Sachs prepared these analyses for purposes of Goldman Sachs providing its opinion to our board of directors as to the fairness from a financial point of view to the holders of the outstanding shares of common stock of the \$1.80 per share of common stock in cash to be received by holders of shares of our common stock in the transaction pursuant to the merger agreement. These analyses do not purport to be appraisals nor do they necessarily reflect the prices at which businesses or securities actually may be sold. Analyses based upon forecasts of future results are not necessarily indicative of actual future results, which may be significantly more or less favorable than suggested by these analyses. Because these analyses are inherently subject to uncertainty, being based upon numerous factors or events beyond the control of the parties or their respective advisors, none of us, Goldman Sachs or any other person assumes responsibility if future results are materially different from those forecast.

The merger consideration was determined through arms -length negotiations between us and Motorola and was approved by our board of directors. Goldman Sachs provided advice to us during these negotiations. Goldman Sachs did not, however, recommend any specific amount of consideration to us or our board of directors or that any specific amount of consideration for the transaction.

As described above, Goldman Sachs opinion to our board of directors was one of many factors taken into consideration by our board of directors in making its determination to approve the merger agreement. The foregoing summary does not purport to be a complete description of the analyses performed by Goldman Sachs in connection with the fairness opinion and is qualified in its entirety by reference to the written opinion of Goldman Sachs attached as Annex B to this proxy statement.

Goldman Sachs and its affiliates, as part of their investment banking business, are continually engaged in performing financial analyses with respect to businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, competitive biddings, secondary distributions of listed and unlisted securities, private placements and other transactions as well as for estate, corporate and other purposes. Goldman Sachs has acted as financial advisor to us in connection with, and has participated in certain of the negotiations leading to, the transaction contemplated by the merger agreement. Goldman Sachs is providing and has provided certain investment banking services to Motorola from time to time, including having acted as remarketing agent in connection with the remarketing of \$1,200,000,000 in principal amount of Motorola s 4.608% notes due 2007 in August 2004, as dealer-manager in connection with the sale of Motorola s 16.4% stake in Pantech Co. in November 2004, as dealer-manager in September 2005 in connection with the tender offer for up to a total of \$1,000,000,000 in principal amount of Motorola s 6.5% notes due 2008, Motorola s 5.8% notes due 2008, Motorola s 7.625% notes due 2010 and Motorola s 8.0% notes due 2011, as exclusive financial advisor in connection with Motorola s investment in Clearwire Corporation and associated acquisition of NextNet Wireless, Inc.

announced in July 2006, as exclusive financial advisor in connection with Motorola s acquisition of Symbol Technologies, Inc. announced in September 2006 and as exclusive financial advisor in connection with a proxy contest relating to Motorola s 2007 Annual Meeting of Stockholders. Goldman Sachs has also acted for Motorola in various stock repurchase activities from time to time. In addition, Goldman Sachs is providing and has provided certain investment banking services to Freescale Semiconductor, Inc. (Freescale), a former subsidiary of Motorola, including having acted as global coordinator and joint bookrunning manager in connection with the initial public offering of 121,621,622 shares of Freescale s class A common stock in July 2004, as lead manager in connection with the offering of \$400,000,000 in principal amount of Freescale s floating rate

#### Table of Contents

notes due 2009 in July 2004, and as co-manager in connection with the offering of \$350,000,000 of Freescale s 6.875% notes due 2011 and \$500,000,000 in principal amount of Freescale s 7.125% notes due July 2014 in July 2004.

Goldman Sachs also may provide investment banking services to us and Motorola in the future. In connection with the above-described investment banking services Goldman Sachs has received, and may receive, compensation.

Goldman, Sachs & Co. is a full service securities firm engaged, either directly or through its affiliates, in securities trading, investment management, financial planning and benefits counseling, risk management, hedging, financing and brokerage activities for both companies and individuals. In the ordinary course of these activities, Goldman, Sachs & Co. and its affiliates may provide such service to us, Motorola and their respective affiliates, may actively trade the debt and equity securities (or related derivative securities) of us and Motorola for their own account and for the accounts of their customers and may at any time hold long and short positions of such securities.

Our board of directors selected Goldman Sachs as its financial advisor because it is an internationally recognized investment banking firm that has substantial experience in transactions similar to the transaction. Pursuant to a letter agreement dated May 30, 2005, as amended by a letter agreement dated April 19, 2007, we engaged Goldman Sachs to act as our financial advisor in connection with the contemplated transactions. Pursuant to the amended letter agreement between Goldman Sachs and us, we have agreed to pay Goldman Sachs a transaction fee of approximately \$4.0 million in connection with the transaction. \$1.0 million of the transaction fee, net of retainer fees paid by us to Goldman Sachs through the end of December 2006, became due upon the signing of the merger agreement based on our request that Goldman Sachs undertake a study in order to enable it to render its opinion as to the fairness from a financial point of view of the \$1.80 per share of common stock in cash to be received by the holders of shares of our common stock pursuant to the merger agreement. The remaining \$3.0 million of the transaction fee is due upon the consummation of the transaction. In addition, retainer fees in the amount of \$50,000 per month from January 2007 onwards have been deferred under the amended letter agreement, and will become due upon the earlier of the consummation of the transaction or the termination of Goldman Sachs engagement under the amended letter agreement. In addition, we have agreed to reimburse Goldman Sachs for its expenses, including attorneys fees and disbursements, and to indemnify Goldman Sachs and related persons against various liabilities, including certain liabilities under the federal securities laws.

#### **Financial Projections**

Our management does not as a matter of course make public projections as to future performance or earnings beyond the current fiscal year and is especially wary of making projections for extended periods due to the unpredictability of the underlying assumptions and estimates. However, certain financial projections prepared by management were made available to Motorola, the other potential acquirors that signed confidentiality agreements with us, our board of directors and Goldman Sachs in connection with a potential transaction. We have included below the material financial projections (on a consolidated basis) to provide our stockholders access to certain nonpublic information that was provided to Motorola, the other potential acquirors that signed confidentiality agreements with us, our board of directors and Goldman Sachs in connection with a potential transaction. The inclusion of this information should not be regarded as an indication that Motorola, the other potential acquirors, our board of directors, Goldman Sachs or any other recipient of this information considered, or now considers, it to be a reliable prediction of future results. Our board of directors considered the execution risks associated with the financial projections below in considering and evaluating the merger, including the fact that the market and earnings per share growth targets were not reflective of our recent historical results.

The financial projections reflect numerous estimates and assumptions with respect to industry performance, general business, economic, regulatory, market and financial conditions, as well as matters specific to our business, all of which are difficult to predict and many of which are beyond our control. As a result, there can be no assurance that the

projected results will be realized or that actual results will not be significantly higher or lower than projected. The financial projections cover multiple years and such information by its nature becomes less reliable with each successive year. The financial projections were prepared solely for internal use and for the use of Motorola, the other potential acquirors that signed confidentiality agreements with us, our board of directors and Goldman Sachs in connection with a potential transaction and not with a view toward public disclosure or toward complying with GAAP, the published guidelines of the SEC regarding projections or the guidelines established by

#### Table of Contents

the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information. For example, the first assumption listed in the second footnote below relating to calculation of gross profit indicates that gross profit was not calculated in accordance with GAAP. The financial projections included below were prepared by, and are the responsibility of, our management. Neither our independent registered public accounting firm, nor any other independent accountants, have compiled, examined or performed any procedures with respect to the prospective financial information contained herein, nor have they expressed any opinion or any other form of assurance on such information or its achievability, and assume no responsibility for, and disclaim any association with, the prospective financial information. The financial projections do not take into account any circumstances or events occurring after the date they were prepared.

The inclusion of financial projections in this proxy statement should not be regarded as an indication that such targets will be an accurate prediction of future events, and they should not be relied on as such. Except as required by applicable securities laws, we do not intend to update, or otherwise revise the financial projections to reflect circumstances existing after the date when made or to reflect the occurrence of future events, even in the event that any or all of the assumptions are shown to be in error. See Cautionary Statement Regarding Forward-Looking Statements.

	Fiscal Year End(1) Estimated Estimated Estimate 12/31/2007(2) 12/31/2008(2) 12/31/2009			timated	Estimated ) 12/31/2010(2)			
Revenue Cost of goods sold	\$	63.5 19.7	\$	69.3 21.6	\$	76.9 23.4	\$	83.2 26.0
Gross profit		43.8		47.7		53.3		57.2
Operating Expenses		43.4		47.1		49.0		50.8
Operating Income		0.3		0.6		4.3		6.4
Other Income		0.7		1.0		1.2		1.3
Income before taxes Income tax expense		1.0 0.4		1.6 0.2		5.5 0.2		7.7 0.2
Net Income		0.6		1.4		5.3		7.5
Diluted EPS	\$	0.01	\$	0.02	\$	0.07	\$	0.09

<sup>(1)</sup> Financial projections for all years were provided to our board of directors and Goldman Sachs. Financial projections for fiscal year 2007 were provided to Motorola and the other potential acquirors that signed confidentiality agreements with us. For purposes of the discounted cash flow analysis performed by Goldman Sachs, financial projections for fiscal year 2011 were derived assuming projected video revenue growth for fiscal year 2011 equal to average projected video revenue growth for fiscal years 2008 through 2010 and projected operating margins for fiscal year 2011 equal to projected operating margins for fiscal year 2010.

(2)

All dollars in the above chart are in millions, except per share data. The principal assumptions of our management preparing the projections are as follows:

Gross profit equals revenue for products shipped plus service recognized during the period less cost of goods sold during the period, and excludes hardware revenue and cost of goods sold from prior periods;

Revenue projections are based on introduction and market acceptance of new video technology products;

Net cash is assumed to be \$21.6 million as of March 31, 2007;

Diluted earnings per share assumes 78 million shares outstanding; and

Effective tax rate is not meaningful, given availability of federal and state net operating losses.

## Interests of Our Directors and Executive Officers in the Merger

In considering the recommendation of our board of directors with respect to the merger, you should be aware that all of our directors and executive officers have personal interests in the merger that are, or may be, different

from, or in addition to, your interests. Our board of directors was aware of the interests described below and considered them, among other matters, when approving the merger agreement and the merger.

*Employment Agreements.* In July 2005, we entered into employment agreements with Jerry Chase, our Chief Executive Officer and a member of our board of directors; Mark Richman, our Chief Financial Officer and Vice President of Finance and Administration; and Matthew Aden, our Vice President of Global Sales and Customer Support, who constitute our executive officers and whom we sometimes refer to as our named executive officers. The employment agreements for each of these executive officers provide for certain severance payments and benefits if their employment terminates following a change in control under certain circumstances. The completion of the merger will constitute a change in control for purposes of the employment agreements. In addition, Motorola has informed us that it intends at this time to terminate the employment of Mr. Chase, Mr. Richman and Mr. Aden around the time the merger is completed.

The employment agreements, as amended as discussed below, provide for the following severance benefits in the event that the employment of these officers is involuntarily terminated (as defined in each employment agreement and set forth below), within twelve months after a change in control (as defined in each employment agreement), provided the executive officer has executed a general form of release:

Mr. Chase and Mr. Richman are entitled to a lump-sum cash payment of 2.5 times and two times, respectively, the sum of (1) the executive officer s base salary (at the greater of the rate in effect at the time of termination or the rate in effect immediately prior to the change in control) and (2) the greater of the executive officer s annual performance bonus for the most recent calendar year prior to the date of termination or the executive officer s target performance bonus in effect for the year in which the executive officer s employment is terminated;

Mr. Aden is entitled to a lump-sum cash payment of two times the sum of (1) his base salary (at the greater of the rate in effect at the time of termination or the rate in effect immediately prior to the change in control) and (2) the greater of his annual sales commission payment for the most recent calendar year prior to the date of termination or his target sales commission payment in effect for the year in which his employment is terminated;

Mr. Chase, Mr. Richman and Mr. Aden are entitled to continuation for up to 24 months (30 months in the case of Mr. Chase), but no longer than the expiration of the applicable continuation coverage period under COBRA, at our cost of the health care benefits that were being provided by us to each executive officer and his dependents immediately prior to the termination of employment;

Mr. Chase, Mr. Richman and Mr. Aden are entitled to full option vesting acceleration; and

In the event any payment or distribution to or for the benefit of Mr. Chase or Mr. Richman under the employment agreements or otherwise is deemed to constitute an excess parachute payment within the meaning of Section 280G of the Internal Revenue Code, and such payments and benefits will cause Mr. Chase and Mr. Richman to incur an excise tax under Section 4999 of the Internal Revenue Code, then the executive officer will be responsible for the first \$200,000 in excise tax imposed and, if the amount of excise tax exceeds \$200,000, then the executive officer will be entitled to receive from the Company an additional payment equal to the amount of any excise tax in excess of \$200,000.

In general involuntary termination (as defined in the employment agreements) means:

The involuntary discharge of the employee by the Company for reasons other than cause or permanent disability (each as defined in the employment agreements); or

the voluntary resignation of the employee within 30 days following:

a material adverse change to the employee s title, authority or responsibilities with the Company;

a material reduction in the employee s base salary; or

receipt of notice that the employee s principal workplace will be relocated by more than 25 miles.

In addition to the severance payments and benefits that Mr. Chase, Mr. Richman and Mr. Aden are entitled to under their employment agreements, we will pay for the cost of employment outplacement counseling services

upon the involuntary termination of their employment under the circumstances described for purposes of their employment agreements.

Our other non-executive officers and certain of our key employees are also entitled to severance benefits similar to the types of severance benefits to which our named executive officers are entitled to receive (as described above) in the event that their employment is involuntarily terminated within twelve months after a change in control, including the completion of the merger.

On April 21, 2007, our board of directors approved amendments to the employment agreements of certain of our employees, including our named executive officers, revising the formulation that will be applied to determine the bonus amount used for calculating the severance amount payable under the employment agreements. As previously written, the formula was based on the greater of the annual performance bonus paid in the prior year or the current year s bonus target. As no bonus targets have been established for 2007 due to our decision to delay the implementation of our 2007 corporate bonus plans, the formulation was amended to provide that the bonus amount would be equal to the greater of (i) the annual performance bonus (or, for Mr. Aden, the annual sales commission payments) paid in 2006 or (ii) a specified percentage of the employee s base salary in 2007. The specified percentages are the same as the bonus target percentages currently set forth in the respective employment agreements for such employees. The amendments to the employment agreements are effective immediately prior to, and are conditioned upon, the completion of the merger.

*Retention Transaction Plan.* In May 2006, we adopted the Retention Transaction Plan, which provides for the payment of transaction bonuses to certain non-executive officers and key employees who remain employed with us following a change in control (as the term is defined in the plan document). The completion of the merger will constitute a change in control for purposes of the Retention Transaction Plan. Under the Retention Transaction Plan, a participant is eligible to receive a transaction bonus equal to a specified percentage of the bonus pool. The bonus pool is equal to 2% of the total merger consideration less \$5,000,000 and less certain incurred expenses and other reductions, and subject to certain limitations, as contemplated in the plan document. None of our named executive officers participates in the Retention Transaction Plan.

*Options.* All options, including options held by our executive officers and directors, to purchase shares of our common stock granted under our 1995 Stock Option Plan, 1997 Equity Incentive Plan, 1998 Non-Employee Directors Stock Option Plan, and 1999 Non-Officer Equity Incentive Plan, as well as all options granted outside of any plan, including any unvested portion of such options, will be cashed out in the merger. For these purposes, cashed out in the merger means that the option, whether vested or unvested, will be cancelled in the merger in exchange for a cash payment equal to the product of (i) the excess, if any, of \$1.80 over the applicable per share option exercise price and (ii) the number of shares of our common stock subject to the option at such time, without interest and less any applicable withholding taxes. The aggregate cash-out value of all of our options that are in-the-money is approximately \$188,000.

*Offer Letters with Motorola.* Contemporaneously with the execution of the merger agreement, four of our key employees who are not executive officers entered into employment offer letters with Motorola that become effective and are conditioned upon the completion of the merger. These employment offer letters, which do not provide for employment with Motorola for any specified period, supersede any employment and compensation arrangements in effect between the employee and us, except for our Retention Transaction Plan, which is described above, and in the case of one of the key employees, an additional performance plan applicable only to him.

Motorola has informed us that it intends to enter into additional employment offer letters with other key employees prior to the completion of the merger.

*Indemnification.* The terms of the merger agreement provide for the continued indemnification of our directors and officers and provision of insurance for their benefit as more fully described below under The Merger Agreement Directors and Officers Indemnification, Advancement of Expenses, Exculpation and Insurance.

### Material United States Federal Income Tax Consequences

The following discussion is a summary of certain material U.S. federal income tax consequences of the merger to United States Holders (as defined below) of our common stock whose shares are converted into the right to

### **Table of Contents**

receive cash at the effective time of the merger. This summary is based on the Internal Revenue Code of 1986, as amended (the Code ), applicable Treasury Regulations, and administrative and judicial interpretations thereof, each as in effect as of the date hereof, all of which may change, possibly with retroactive effect.

This summary is limited to United States Holders who hold shares of our common stock as capital assets. This summary also does not address tax considerations applicable to a holder s particular circumstances or to holders who may be subject to special tax rules, including, without limitation:

banks, insurance companies or other financial institutions;

broker-dealers;

traders;

expatriates;

tax-exempt organizations;

persons subject to alternative minimum tax;

partnerships, S-corporations or other pass-through entities;

persons who hold their shares of common stock as a position in a hedging transaction, straddle, conversion transaction, or other risk reduction transaction;

persons deemed to sell their shares of common stock under the constructive sale provisions of the Code;

persons who have a functional currency other than the United States dollar; or

persons who acquired their shares of our common stock upon the exercise of stock options or otherwise in connection with the performance of services.

In addition, this summary does not address any U.S. federal estate or gift tax consequences, nor any state, local or foreign tax consequences, of the merger, and this summary does not address the tax consequences to holders of our common stock who exercise appraisal rights under Delaware law.

THIS SUMMARY DOES NOT PURPORT TO BE A COMPLETE ANALYSIS OF THE POTENTIAL TAX CONSIDERATIONS RELATING TO THE MERGER, AND IS NOT TAX ADVICE. THEREFORE, YOU ARE URGED TO CONSULT YOUR TAX ADVISOR WITH RESPECT TO THE APPLICATION OF THE U.S. FEDERAL INCOME TAX LAWS TO YOUR PARTICULAR SITUATION, AS WELL AS ANY TAX CONSEQUENCES OF THE MERGER ARISING UNDER THE FEDERAL ESTATE OR GIFT TAX RULES OR UNDER THE LAWS OF ANY STATE, LOCAL OR FOREIGN TAXING JURISDICTION OR UNDER ANY APPLICABLE TAX TREATY.

For purposes of this discussion, a United States Holder means a holder that is:

an individual citizen or resident of the United States;

a corporation (or an entity treated as a corporation for U.S. federal income tax purposes) created or organized in or under the laws of the United States, any state thereof or the District of Columbia;

an estate the income of which is subject to U.S. federal income taxation regardless of its source; or

a trust (a) the administration over which a U.S. court can exercise primary supervision and all of the substantial decisions of which one or more U.S. persons have the authority to control and (b) certain other trusts considered United States Holders for U.S. federal income tax purposes.

## Consequences of the Merger

The receipt of cash in exchange for shares of our common stock pursuant to the merger will be a taxable transaction for U.S. federal income tax purposes. In general, a United States Holder who receives cash in exchange for shares of our common stock pursuant to the merger will recognize capital gain or loss for U.S. federal income tax purposes equal to the difference, if any, between the amount of cash received and the holder s adjusted tax basis in

### Table of Contents

the shares of our common stock exchanged for that cash. Any such gain or loss would be long-term capital gain or loss if the holding period for the shares of our common stock exceeded one year. Long-term capital gains of noncorporate taxpayers are generally taxable at reduced rates. The deductibility of capital losses is subject to limitations. Gain or loss must be calculated separately for each block of common stock (i.e., shares acquired at the same cost in a single transaction) exchanged for cash in the merger.

## **Backup Withholding**

Backup withholding may apply to payments made in connection with the merger. Backup withholding will not apply, however, to a holder who (1) furnishes the holder s correct taxpayer identification number and certifies that the holder is not subject to backup withholding on the substitute Form W-9 or successor form included in the letter of transmittal to be delivered to holders of our common stock prior to completion of the merger, or (2) is otherwise exempt from backup withholding. Backup withholding is not an additional tax, but rather, is a method of tax collection. Any amounts withheld under the backup withholding rules may be allowed as a refund or a credit against such holder s U.S. federal income tax liability provided the required information is furnished to the Internal Revenue Service.

### **Regulatory Approvals**

Under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 and the rules promulgated thereunder by the Federal Trade Commission, certain transactions, including the merger, may not be completed unless waiting period requirements have been satisfied. We and Motorola filed notification and report forms with the Antitrust Division of the Department of Justice and the Federal Trade Commission on May 18, 2007. The waiting period will expire at 11:59 p.m., Eastern time, on June 18, 2007, unless extended by a request for more information or shortened by an early termination notice. In addition, filings are required to be made with, and regulatory approvals are required to be obtained from, antitrust or competition authorities in other nations, including Germany, Israel and the Ukraine. All of these filings have been made or will be made shortly. Until approvals have been received from antitrust or competition authorities, or required waiting periods have ended, in each of those jurisdictions, we are prohibited from completing the merger. Neither we nor Motorola has yet obtained any of the governmental or regulatory approvals required to complete the merger.

While we expect to obtain all required regulatory approvals, we cannot assure you that these regulatory approvals will be obtained or that the granting of these regulatory approvals will not involve the imposition of conditions on the completion of the merger or require changes to the terms of the merger that would have a material adverse effect on the combined company. These conditions or changes could require the grant of a complete or partial license, a divestiture or spin-off, or the holding separate of assets or businesses and could result in the conditions to Motorola s obligation to complete the merger not being satisfied.

In addition, at any time before or after the effective time of the merger, the Antitrust Division, the Federal Trade Commission or others could take action under the antitrust laws, including seeking to prevent the merger, to rescind the merger or to conditionally approve the merger upon the divestiture by us or Motorola of substantial assets. In addition, in some jurisdictions, a competitor, customer or other third party could initiate a private action under the antitrust laws challenging or seeking to enjoin the merger, before or after it is completed. There can be no assurance that a challenge to the merger on antitrust grounds will not be made or, if such a challenge is made, that it would not be successful.

# **Appraisal Rights**

Holders of record of shares of our common stock who do not vote in favor of the adoption of the merger agreement and approval of the merger and who properly demand appraisal of their shares will be entitled to appraisal rights in

### Table of Contents

connection with the merger under Section 262 of the General Corporation Law of the State of Delaware.

The following discussion is not a complete statement of the law pertaining to appraisal rights under Section 262 and is qualified in its entirety by the full text of Section 262 which is attached to this proxy statement as Annex C. All references in Section 262 and in this summary to a stockholder are to the record

holder of the shares of our common stock as to which appraisal rights are asserted. A person having a beneficial interest in shares of our common stock held of record in the name of another person, such as a broker, fiduciary, depositary or other nominee, must act promptly to cause the record holder to follow the steps summarized below properly and in a timely manner to perfect appraisal rights.

Under Section 262, stockholders who follow the procedures set forth in Section 262 will be entitled to have their shares appraised by the Delaware Court of Chancery and to receive payment of the fair value of the shares, exclusive of any element of value arising from the accomplishment or expectation of the merger, together with a fair rate of interest, if any, as determined by the court.

Under Section 262, where a merger is to be submitted for approval at a meeting of stockholders, as in the case of the adoption of the merger agreement and approval of the merger by our stockholders, the corporation, not less than 20 days prior to the meeting, must notify each of its stockholders entitled to appraisal rights that appraisal rights are available and include in the notice a copy of Section 262. This proxy statement shall constitute the notice, and the full text of Section 262 is attached to this proxy statement as Annex C. Any stockholder who wishes to exercise appraisal rights or who wishes to preserve such stockholder s right to do so, should review the following discussion and Annex C carefully because failure to timely and properly comply with the procedures specified will result in the loss of appraisal rights.

Any stockholder wishing to exercise appraisal rights must deliver to us, before the vote on the adoption of the merger agreement and approval of the merger at the special meeting, a written demand for the appraisal of the stockholder s shares, and must not vote in favor of the adoption of the merger agreement and approval of the merger. A stockholder wishing to exercise appraisal rights must hold of record the shares on the date the written demand for appraisal is made and must continue to hold the shares of record through the effective time of the merger. A vote against the adoption of the merger agreement and approval of the merger agreement and approval of the merger agreement and approval of the merger will not in and of itself constitute a written demand for appraisal satisfying the requirements of Section 262. The written demand for appraisal must be in addition to and separate from any proxy or vote. The demand must reasonably inform us of the identity of the stockholder as well as the intention of the stockholder to demand an appraisal of the fair value of the shares held by the stockholder. A stockholder s failure to make the written demand prior to the taking of the vote on the adoption of the merger agreement and approval of the merger at the special meeting of stockholders will constitute a waiver of appraisal rights.

Only a holder of record of shares of our common stock is entitled to assert appraisal rights for the shares registered in that stockholder s name. A demand for appraisal should be executed by or on behalf of the holder of record, fully and correctly, as the stockholder s name appears on the stockholder s stock certificates, and must state that the person intends thereby to demand appraisal of the stockholder s shares in connection with the merger. If the shares are owned of record in a fiduciary capacity, such as by a trustee, guardian or custodian, execution of the demand should be made in that capacity, and if the shares are owned of record by more than one person, as in a joint tenancy and tenancy in common, the demand should be executed by or on behalf of all joint owners. An authorized agent, including an agent for two or more joint owners, may execute a demand for appraisal on behalf of a holder of record; however, the agent must identify the record owner or owners and expressly disclose the fact that, in executing the demand, the agent is acting as agent for the record owner or owners. A record holder such as a broker who holds shares as nominee for several beneficial owners may exercise appraisal rights with respect to the shares held for one or more beneficial owners while not exercising the rights with respect to the shares held for other beneficial owners; in such case, however, the written demand should set forth the number of shares as to which appraisal is sought and where no number of shares is expressly mentioned the demand will be presumed to cover all shares of our common stock held in the name of the record owner. Stockholders who hold their shares in brokerage accounts or other nominee forms and who wish to exercise appraisal rights are urged to promptly consult with their brokers or other nominees to determine the appropriate procedures for the making of a demand for appraisal by such broker or

### nominee.

All written demands for appraisal pursuant to Section 262 should be sent or delivered to us at Terayon Communication Systems, Inc., 2450 Walsh Avenue, Santa Clara, California 95051, Attention: Secretary.

Within ten days after the effective time of the merger, the surviving corporation must notify each stockholder who has complied with Section 262 and who has not voted in favor of the adoption of the merger agreement and

### Table of Contents

approval of the merger that the merger has become effective. Within 120 days after the effective time of the merger, but not thereafter, the surviving corporation or any stockholder who has so complied with Section 262 and is entitled to appraisal rights under Section 262 may file a petition in the Delaware Court of Chancery demanding a determination of the fair value of the stockholder s shares. The surviving corporation is under no obligation to and has no present intention to file a petition. Accordingly, it is the obligation of our stockholders to initiate all necessary action to perfect their appraisal rights within the time prescribed in Section 262.

Within 120 days after the effective time of the merger, any stockholder who has complied with the requirements for exercise of appraisal rights will be entitled, upon written request, to receive from the surviving corporation a statement setting forth the aggregate number of shares not voted in favor of the adoption of the merger agreement and approval of the merger and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. The statement must be mailed within ten days after a written request therefor has been received by the surviving corporation or within ten days after the expiration of the period for delivery of demands for appraisal, whichever is later.

If a petition for an appraisal is timely filed and a copy thereof is served upon the surviving corporation, the surviving corporation will then be obligated within 20 days to file with the Delaware Register in Chancery a duly verified list containing the names and addresses of all stockholders who have demanded an appraisal of their shares and with whom agreements as to the value of their shares have not been reached. After notice to the stockholders as required by the Court, the Delaware Court of Chancery is empowered to conduct a hearing on the petition to determine those stockholders who have complied with Section 262 and who have become entitled to appraisal rights thereunder. The Delaware Court of Chancery may require the stockholders who demanded payment for their shares to submit their stock certificates to the Register in Chancery for notation thereon of the pendency of the appraisal proceeding; and if any stockholder fails to comply with the direction, the Court of Chancery may dismiss the proceedings as to the stockholder.

After determining the stockholders entitled to appraisal, the Delaware Court of Chancery will appraise the fair value of their shares, exclusive of any element of value arising from the accomplishment or expectation of the merger, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value. Stockholders considering seeking appraisal should be aware that the fair value of their shares as so determined could be more than, the same as or less than the consideration they would receive pursuant to the merger if they did not seek appraisal of their shares and that an investment banking opinion as to fairness from a financial point of view is not necessarily an opinion as to fair value under Section 262. We do not anticipate offering more than \$1.80 per share to any stockholder exercising appraisal rights and we reserve the right to assert, in any appraisal proceeding, that for purposes of Section 262, the fair value of a share of our common stock is less than \$1.80. The Delaware Supreme Court has stated that proof of value by any techniques or methods which are generally considered acceptable in the financial community and otherwise admissible in court should be considered in the appraisal proceedings. In addition, Delaware courts have decided that the statutory appraisal remedy, depending on factual circumstances, may or may not be a dissenter s exclusive remedy. The Delaware Court of Chancery will also determine the amount of interest, if any, to be paid upon the amounts to be received by persons whose shares of our common stock have been appraised. The costs of the action may be determined by the Court and taxed upon the parties as the Court deems equitable. The Court may also order that all or a portion of the expenses incurred by any stockholder in connection with an appraisal, including, without limitation, reasonable attorneys fees and the fees and expenses of experts utilized in the appraisal proceeding, be charged pro rata against the value of all the shares entitled to be appraised.

Any stockholder who has duly demanded an appraisal in compliance with Section 262 will not, after the effective time of the merger, be entitled to vote the shares subject to the demand for any purpose or be entitled to the payment of dividends or other distributions on those shares (except dividends or other distributions payable to holders of record of our common stock as of a record date prior to the effective time of the merger).

If any stockholder who demands appraisal under Section 262 fails to perfect, or effectively withdraws or loses, such holder s right to appraisal, the stockholder s shares will be deemed to have been converted at the effective time of the merger into the right to receive \$1.80 in cash per share. A stockholder will fail to perfect, or effectively lose or withdraw, the holder s right to appraisal if no petition for appraisal is filed within 120 days after the effective time of

#### **Table of Contents**

the merger, or if the stockholder delivers to the surviving corporation a written withdrawal of the holder s demand for appraisal and an acceptance of the merger, except that any attempt to withdraw made more than 60 days after the effective time of the merger will require the written approval of the surviving corporation and, once a petition for appraisal is filed, the appraisal proceeding may not be dismissed as to any holder absent court approval.

Stockholders are urged to read Section 262, which is included in this proxy statement as Annex C, carefully and in its entirety. Failure to follow the steps required by Section 262 for perfecting appraisal rights may result in the loss of such rights.

#### Structure of the Merger

The merger agreement provides that, upon the terms and subject to the conditions of the merger agreement, Merger Sub will be merged with and into us and the separate corporate existence of Merger Sub will thereupon cease, and we will be the surviving corporation of the merger and will be a wholly owned subsidiary of Motorola. At the effective time of the merger, all of our property, rights, privileges, powers and franchises and those of the Merger Sub will vest in the surviving corporation, and all of our debts, obligations, claims, liabilities and duties and those of the Merger Sub will continue as the surviving corporation and will be unaffected by the merger, except that all of our then outstanding common stock will be owned by Motorola.

#### **Effective Time of the Merger**

The merger will become effective when a certificate of merger is duly filed with the Secretary of State of the State of Delaware or at such later time as the parties agree to and specify in the certificate of merger, which we refer to as the effective time of the merger. Such filing will be made no later than the second business day after the satisfaction or waiver of the conditions to the completion of the merger described in the merger agreement, unless otherwise agreed by Motorola and us. See The Merger Agreement Conditions to the Completion of the Merger.

The merger agreement also provides that:

our restated certificate of incorporation will be amended and restated in its entirety to be in the form of Exhibit A to the merger agreement and, as so amended, will be the restated certificate of incorporation of the surviving corporation until later amended;

after the effective time of the merger, the authorized capital stock of the surviving corporation will consist of 1,000 shares of common stock, par value \$0.01 per share;

our bylaws will be amended and restated in their entirety to be in the form of the bylaws of Merger Sub at the effective time of the merger until later amended;

the directors of Merger Sub immediately prior to the effective time of the merger will be the directors of the surviving corporation; and

the officers of Merger Sub immediately prior to the effective time of the merger will be the officers of the surviving corporation.

#### **Merger Consideration**

Each share of our common stock issued and outstanding immediately before the effective time of the merger, other than:

treasury shares or shares held by Motorola or Merger Sub or any direct or indirect subsidiary of Motorola, all of which will be cancelled in the merger, and

shares held by stockholders who comply with all of the provisions of the General Corporation Law of the State of Delaware concerning the right to require appraisal of their shares, whom we refer to as dissenting stockholders,

#### Table of Contents

will be automatically converted into and represent the right to receive \$1.80 in cash, without interest. After the merger is effective, stockholders will no longer have any rights with respect to their shares, except for the right to receive the \$1.80 per share merger consideration. No interest will accrue or be paid with respect to the merger consideration. At the effective time of the merger, shares of our common stock will no longer be outstanding, will automatically be cancelled and will cease to exist. Stockholders will not receive the merger consideration until they surrender their stock certificates to the paying agent for exchange (as described below).

After the merger is effective, each dissenting stockholder will also no longer have any rights as a stockholder, except for the right to receive payment of the judicially-determined fair value of his or her shares pursuant to Delaware law if the stockholder has validly perfected and not withdrawn or lost such right.

## YOU SHOULD NOT RETURN STOCK CERTIFICATES WITH THE ENCLOSED PROXY.

### **Treatment of Our Stock Options**

If we complete the merger, at the effective time of the merger, each outstanding stock option to purchase our common stock, whether vested or unvested, will be cancelled, and holders of in-the-money options will be entitled to receive a cash payment for each option equal to the product of (a) the excess of \$1.80 over the exercise price per share of such option, multiplied by (b) the total number of shares of our common stock underlying such option at the effective time of the merger. The cash payment will be reduced by applicable withholding taxes. The aggregate cash-out value of all of our options that are in-the-money is approximately \$188,000.

### Surrender of Stock Certificates

At or prior to the effective time of the merger, Motorola will deposit or cause to be deposited, for the benefit of our stockholders, the aggregate merger consideration into an exchange fund with Computershare Shareholder Services or another comparable institution, as paying agent.

As soon as practicable (but within five business days) after the effective time of the merger, the paying agent will mail to each holder of record of shares of our common stock a letter of transmittal and instructions for effecting the surrender of the certificates to the paying agent in exchange for the merger consideration payable in respect of the shares formerly represented by such certificates. After the paying agent has received a stockholder s certificate(s), together with a properly completed letter of transmittal, the paying agent will deliver to the stockholder the \$1.80 per share merger consideration for shares of common stock multiplied by the number of shares formerly represented by the stockholder, subject to any required withholding taxes. In the event a transfer of ownership of our common stock is not registered in our transfer records, payment of the merger consideration will be conditioned upon:

the certificate being properly endorsed with signature guaranteed, or otherwise in proper form for transfer; and

the person requesting payment of the merger consideration paying any transfer or other taxes required as a result of the transfer or establishing to the satisfaction of Motorola that such taxes have been paid or are not applicable.

### **Lost Certificates**

If any certificate representing shares of our common stock is lost, stolen or destroyed, the paying agent will deliver the merger consideration for the shares if:

### Table of Contents

the stockholder asserting the claim of a lost, stolen or destroyed certificate makes an affidavit of that fact; and

if required by Motorola, the stockholder posts a bond in a reasonable amount, determined by Motorola, as indemnity against any claim that may be made against Motorola with respect to the lost certificate.

### **Unclaimed Amounts**

Any portion of the exchange fund which remains undistributed to our stockholders on the date that is 180 days after the effective time of the merger will be delivered by the paying agent to Motorola, and any of our stockholders who have not previously surrendered their stock certificates will be entitled to receive, upon delivery of such certificates to Motorola, payment of the merger consideration from Motorola. Subject to the other terms of the merger agreement, Motorola will remain liable for the payment of the merger consideration to these stockholders.

# THE MERGER AGREEMENT

The following description summarizes the material provisions of the merger agreement and is qualified in its entirety by reference to the complete text of the merger agreement. The merger agreement included in this proxy statement as Annex A contains the complete terms of that agreement and stockholders should read it carefully and in its entirety.

### **Representations and Warranties**

The merger agreement contains representations and warranties made by us to Motorola and Merger Sub, including those relating to:

due organization, existence, corporate power and standing and other corporate matters of us and our subsidiaries;

the capital structure of us and of our subsidiaries;

the authorization, execution, delivery and enforceability of the merger agreement and the applicable stockholder voting requirements;

our receipt of a fairness opinion from Goldman Sachs;

required governmental consents, approvals, authorizations, notices and filings, the absence of conflicts under our certificate of incorporation or bylaws and the absence of breaches, violations, or defaults under any of our contracts, arrangements, permits or licenses as a result of the execution, delivery and performance of the merger agreement;

company material contracts, including contracts

relating to employment, service or consulting arrangements,

with major customers,

containing exclusivity rights, most favored customer pricing provisions, or penalties for late delivery, repairs, returns or quality performance,

with major or sole source suppliers,

relating to specified design, manufacturing and development arrangements or containing volume commitments, rights of first refusal or similar provisions,

limiting our rights to engage in business, make, sell or distribute products or services, or use or transfer intellectual property rights,

relating to the disposition or acquisition of assets, provision of source code, indebtedness or settlements,

outlicenses,

relating to the assignment of intellectual property,

relating to future amounts in excess of \$100,000,

with affiliates and advisors,

providing for indemnification by us,

with vendors, resellers and distributors, and

other contracts material to us and our subsidiaries;

compliance with the reporting and filing requirements of the SEC, the accuracy of information, including financial information, contained in documents filed with the SEC and the absence of liabilities other than those set forth on the Company s December 31, 2006 balance sheet and other specified liabilities;

#### **Table of Contents**

the accuracy of the information supplied by us for inclusion in this proxy statement;

our disclosure controls and procedures and internal controls over financial reporting and compliance with provisions of the Sarbanes-Oxley Act;

the absence of material changes or events since December 31, 2006 concerning us or our subsidiaries;

pending or threatened litigation, investigations or proceedings;

indebtedness and indemnification obligations with respect to our and our subsidiaries directors and officers;

employment and employee benefits matters, matters affecting us relating to the Employee Retirement Income Security Act of 1974, as amended, and other compliance, compensation and employment matters (including employee benefit plans, collective bargaining or other labor union agreements and excess parachute payments under Section 280G of the Internal Revenue Code of 1986, as amended);

regulatory compliance matters, compliance with applicable labor, environmental, import and export control, anti-bribery and other applicable laws, and material consent decrees;

completion and accuracy of our tax filings, payment or accrual of our tax obligations, absence of other tax liabilities and other tax matters;

insurance matters;

matters relating to our intellectual property;

our title to and interest in material tangible assets and properties, our rights to leased real property, and compliance with the terms of, and absence of disputes or defaults under, our leases and subleases;

contracts with governmental entities;

product liability and recalls;

the inapplicability of anti-takeover statutes with respect to the proposed merger;

amendment and termination of our rights plan; and

our engagement and payment of fees for financial advisors in connection with the merger.

The merger agreement also contains representations and warranties made by Motorola and Merger Sub to us, including those relating to:

due organization, existence, corporate power and standing and other corporate matters of Motorola and Merger Sub;

the authorization, execution, delivery and enforceability of the merger agreement and the corporate power and authority to consummate the merger;

required governmental consents, approvals, authorizations, notices and filings, the absence of conflicts under governing documents, the absence of breaches, violations, or defaults under any of Motorola s or Merger Sub s contracts, arrangements, permits or licenses and the absence of a required vote of Motorola s securityholders as a result of the execution, delivery and performance of the merger agreement;

the accuracy of the information supplied by Motorola and Merger Sub for inclusion in this proxy statement;

Merger Sub s operations;

the absence of brokers and finders fees for which we would be liable in connection with the merger; and

the immediate availability of Motorola funds to pay the merger consideration.

The representations and warranties of each of the parties to the merger agreement described above will expire upon completion of the merger. The representations included in the merger agreement were made by each of us and Motorola to each other. These representations and warranties were made as of specific dates, may (along with the conduct of business covenants, described below) be subject to important qualifications and limitations agreed to by

### Table of Contents

us and Motorola in connection with negotiating the terms of the merger agreement, and may have been included in the merger agreement for the purpose of allocating risk between us and Motorola rather than to establish matters of fact. This description of the representations and warranties, and their reproduction in the copy of the merger agreement attached to this document as Annex A, are included solely to provide investors with information regarding the terms of the merger agreement, and not to provide any other factual information regarding us or our business. Accordingly, the representations and warranties and other provisions of the merger agreement should not be read alone, but instead should be read in conjunction with the information provide elsewhere in this document and the information we file with the SEC, as described under Where You Can Find Additional Information, below.

### Covenants

Under the merger agreement, we have agreed that, prior to the completion of the merger (unless Motorola provides its consent in writing, which will not be unreasonably withheld or delayed), subject to specified exceptions, we and our subsidiaries will:

carry on our business in the ordinary and usual course; and

use commercially reasonable efforts to preserve our respective current business organizations and maintain our respective existing relations and goodwill with customers, suppliers, distributors, strategic partners, creditors, lessors, employees and business associates.

In addition, we have agreed that, among other things and subject to specified exceptions, neither we nor any of our subsidiaries may, without Motorola s prior written consent, which will not be unreasonably withheld or delayed, take any of the following actions:

issue, sell, pledge, dispose of or otherwise encumber any of the capital stock of our subsidiaries;

amend our certificate of incorporation or bylaws or equivalent governing instruments of our subsidiaries;

split, combine or reclassify outstanding shares of any of our capital stock;

declare, set aside or pay any dividend in respect of our capital stock (other than dividends from our direct or indirect wholly owned subsidiaries);

purchase, redeem or otherwise acquire any shares of our or our subsidiaries capital stock or securities convertible into or exchangeable or exercisable for any shares of our or our subsidiaries capital stock (other than the acquisition of shares of our common stock from our optionholders in payment of the exercise price payable by these optionholders pursuant to their exercise of options);

authorize, issue, sell, pledge, dispose of or encumber our or any of our subsidiaries capital stock, securities convertible into or exchangeable for, or rights of any kind to acquire, any shares of our or our subsidiaries capital stock, securities having the right to vote, or other property or assets (other than shares of our common stock and associated rights issuable pursuant to options and stock-based awards outstanding on the date of the merger agreement);

transfer, lease, license, guarantee, sell, mortgage, pledge, dispose of, abandon, cancel, surrender or allow to lapse or expire or encumber any material property or material assets (other than the sale of inventory in the ordinary course of business);

restructure, recapitalize, reorganize, liquidate, adopt a plan of liquidation, or otherwise enter into any agreement that imposes material changes or restrictions on the operation of our or any of our subsidiaries assets or businesses, or adopt any resolutions for such actions;

acquire by merging or consolidating with, or by purchasing all or a substantial portion of the assets or securities of, any business or entity or any material assets, except for purchases of inventory and raw material in the ordinary course of business;

incur or guarantee any indebtedness for borrowed money, issue, sell or amend any debt securities or rights to acquire debt securities, or enter into any keep well or similar agreement;

#### **Table of Contents**

other than accounts receivable in the ordinary course of business, make any loans, advances (other than routine advances to employees in the ordinary course of business), or capital contributions to, or investments in, any other person (other than us or any of our wholly owned subsidiaries), or enter into any hedging agreements or similar arrangements other than in the ordinary course of business;

make any capital expenditures or other expenditures with respect to property, plant or equipment in excess of \$200,000 in the aggregate for us and our subsidiaries, other than as set forth in our budget for capital expenditures;

make any material changes in accounting methods, principles or practices, or change any assumption underlying, or method calculating, any bad debt, contingency or other reserve, except as required by a change in GAAP or applicable law;

other than in the ordinary course of business consistent with past practice, enter into, renew, modify, amend, terminate, waive, delay the exercise of, release or assign any material rights or claims under, any material contract or lease;

except to comply with law or existing agreements and arrangements,

take any action with respect to, adopt, enter into, terminate or amend any employment, severance, change in control, retirement, retention, welfare, incentive or similar agreement, arrangement or benefit plan for the benefit or welfare of any current or former director, officer, employee or consultant or any collective bargaining agreement;

increase the compensation or fringe benefits of, or pay bonuses to, any director, officer, employee or consultant;

amend or accelerate the payment, right to payment or vesting of compensation or benefits;

pay any benefit not provided for under any company benefit plan as of the date of the merger agreement;

grant any awards under any bonus, incentive, performance or other compensation plan or benefit plan or arrangement, including the grant of any equity awards, or the removal of existing restrictions in any benefit plans or agreements or awards made under the aforementioned plans or arrangements; or

take any action to fund or in any other way secure the payment of compensation or benefits under any company benefit plan;

initiate, settle or compromise any intellectual property litigation or material litigation, claim, grievance, charge or proceeding;

make or rescind any material tax election, amend any material tax return or permit any insurance policy naming us as a beneficiary to be cancelled or terminated, except in a manner consistent with past practice or as required by law; or

authorize, commit, resolve or agree to do any action described above.

In addition, we have agreed that we and our subsidiaries will timely file all material tax returns required to be filed by us and such subsidiaries, as the case may be, and timely pay or accrue all material taxes due and payable by us and such subsidiaries, except with respect to matters contested in good faith.

We and Motorola have agreed to notify each other promptly after learning of the following:

the occurrence of various claims, communications or investigations relating to the merger or initiated by or against the notifying party or threatened against the notifying party, or its officers, directors, employees or stockholders;

changes or events that are reasonably likely to cause the conditions to completion of the merger not to be satisfied;

any claim or written inquiry by a taxing authority regarding a material deficiency to pay taxes payable by us;

#### Table of Contents

matters relating to our disclosure controls and procedures, internal controls and other corporate governance representations; and

offers received by us to settle or compromise claims or proceedings.

We and Motorola have agreed to use commercially reasonable efforts to notify each other promptly of matters that relate to the accuracy of our representations and warranties or compliance with our covenants and agreements.

In addition, prior to the completion of the merger, we have agreed to provide Motorola with reasonable access to our properties, books, contracts and records, and such information concerning our business, properties and personnel as may reasonably be required by Motorola and that we are legally able to provide.

### Directors and Officers Indemnification, Advancement of Expenses, Exculpation and Insurance

Motorola has agreed to cause the surviving corporation to indemnify and hold harmless, through the sixth anniversary of the merger, to the fullest extent permitted under Delaware law, each current and former director or officer of us or our subsidiaries, against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, arising out of or incurred by any of them in connection with any claim, action, suit, proceeding, demand or investigation arising out of or pertaining to their service as an officer or director of us or any of our subsidiaries. Each director or officer will be entitled to advancement of expenses incurred in the defense of any such claim, action, suit, proceeding or investigation from the surviving corporation, provided that the third party to whom expenses are advanced provides an undertaking to repay such advances if it is ultimately determined that such third party is not entitled to indemnification.

Motorola has also agreed to cause the surviving corporation to purchase tail coverage under our current officers and directors liability insurance for a period of six years after the completion of the merger in respect of matters existing or occurring at or prior to the merger. In the event that the premium for tail coverage for a period of six years exceeds a specified maximum amount, Motorola will cause the surviving corporation to obtain the longest tail coverage that can be obtained for a premium not in excess of that specified amount.

#### **Options and Employee Stock Purchase Plans; Employee Benefits Matters**

All options to purchase shares of our common stock granted under our 1995 Stock Option Plan, 1997 Equity Incentive Plan, 1998 Non-Employee Directors Stock Option Plan, and 1999 Non-Officer Equity Incentive Plan, as well as options granted outside of any plan, including any unvested portion of such options, will be cashed out in the merger. For these purposes, cashed out in the merger means that the option, whether vested or unvested, will be cancelled in the merger in exchange for a cash payment equal to the product of (i) the excess, if any, of \$1.80 over the applicable per share option exercise price and (ii) the number of shares of our common stock subject to the option at such time, without interest and less any applicable withholding taxes. The aggregate cash-out value of all of our options that are in-the-money is approximately \$188,000.

Immediately prior to the completion of the merger, we have agreed to terminate our 1998 Employee Stock Purchase Plan and our 1998 Employee Stock Purchase Plan for Foreign Employees (collectively, our ESP Plans). Our ESP Plans are currently suspended and no offering is currently outstanding under our ESP Plans.

Under the merger agreement, we have agreed that, if requested by Motorola in writing, we will terminate any or all existing employee benefit programs effective immediately prior to the completion of the merger, except for certain specified arrangements.

### **Efforts to Complete the Merger**

Upon the terms and subject to the conditions set forth in the merger agreement, each party has agreed to use its commercially reasonable efforts to take all actions and do all things necessary, proper or advisable, to complete the merger as promptly as practicable, including using commercially reasonable efforts to accomplish the following:

as promptly as practicable, obtain any consents, licenses, permits, waivers, approvals, authorizations or orders that are either required or reasonably requested by Motorola in connection with the merger agreement and the merger;

as promptly as practicable, and on a mutually agreed date, make all necessary filings, and thereafter make any other required submissions, with respect to the merger agreement and the merger required under applicable law;

execute and deliver any additional instruments necessary to complete the merger and to carry out fully the purposes of the merger agreement; and

obtain any government clearances or approvals required to complete the merger under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and other applicable antitrust laws, and contest and resist any action, and have any decree or order vacated or overturned, that restricts or prohibits the merger; however, the foregoing shall not require Motorola or any of its affiliates to transfer, encumber or hold separate any assets or businesses or agree to any restriction on Motorola s ability to own or operate any assets or businesses or exercise ownership rights with respect to the stock of the surviving corporation, or to take any other action if a governmental entity seeks a preliminary injunction or restraining order to enjoin the merger.

We and Motorola have agreed to consult and cooperate with each other in connection with obtaining all required consents, licenses, permits, waivers, approvals, authorizations or orders and to use commercially reasonable efforts to furnish one another with all information required for any application or filing made pursuant to any applicable law.

We have agreed to cooperate with Motorola to help Motorola assess the desirability of obtaining additional insurance coverage relating to specified litigation matters. In addition, if requested by Motorola at least 10 business days before completion of the merger, we have agreed to use our reasonable best efforts to apply for and to purchase, immediately prior to or concurrent with the merger, a litigation insurance policy with respect to these litigation matters, naming us and our subsidiaries as the named insureds and Motorola and its affiliates as additional insureds. In the event that Motorola unilaterally terminates the merger agreement in accordance with its terms, Motorola has agreed to reimburse us for the premium paid by us for the policy.

We have also agreed to provide to Motorola our tax apportionment schedule for 2005 and similar available apportionment information for other years as reasonably requested by Motorola, which will include jurisdictions in which we had sales, owned property, employed individuals or filed income tax returns for such relevant year. Our Chief Financial Officer must deliver a certificate to Motorola certifying that, to the best of his knowledge, the apportionment information delivered to Motorola for 2002 and each taxable year thereafter is true, correct and complete, and the apportionment information delivered to Motorola for all other years is true, correct and, to the extent available, complete. We must supplement the apportionment information if we become unable to deliver the certificate referenced below. We have also agreed to cooperate with Motorola as necessary to assist Motorola in reviewing the apportionment information. Motorola may notify us in writing within 20 days after receipt of the certificate, or five days after receipt of the supplemental information, if Motorola reasonably believes that we have been delinquent in filing any income tax return not disclosed to Motorola. To the extent that a tax return identified by Motorola is considered delinquent under the merger agreement, we are required to file that tax return unless we and Motorola agree that it does not need to be filed. If Motorola provides us with such a notice within 30 days of the outside termination date, September 21, 2007, and the delinquent tax return is the only condition to closing that is not satisfied as of the outside termination date, then the outside termination date of the merger agreement will be extended to the 30<sup>th</sup> day after we receive the notice from Motorola. At least five days before the merger, our Chief Financial Officer must deliver an additional certificate to Motorola certifying that the apportionment information previously provided to Motorola (as supplemented) remains, to the best of his knowledge, true, correct and complete, and that we have filed all U.S. federal income tax returns and all other delinquent income tax returns.

### Conditions to the Completion of the Merger

The obligations of each of Motorola and Merger Sub, on the one hand, and us, on the other hand, to complete the merger depend on the satisfaction or waiver, on or prior to the effective time of the merger, of a number of conditions, including the following:

the adoption of the merger agreement and the approval of the merger by the affirmative vote of stockholders holding a majority of the shares of our common stock outstanding at the close of business on the record date for the special meeting;

expiration or termination of the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the making of filings, receipt of approvals, and expiration or termination of applicable waiting periods under any other applicable antitrust laws;

no order, stay, decree, judgment or injunction or statute, rule or regulation by any governmental entity prohibits or makes illegal the completion of the merger or the other transactions contemplated by the merger agreement;

each of the obligations of the other party required to be performed on or prior to the closing date of the merger have been performed in all material respects; and

the representations and warranties of the other party set forth in the merger agreement being true and correct without regard to materiality and material adverse effect qualifiers, as of the date of the merger agreement and as of the completion date (except when made as of an earlier date, in which case as of that date), except to the extent that the aggregate of all inaccuracies with respect to such representations and warranties would not have a material adverse effect on such party, and solely with respect to us, except for capitalization, corporate authority and takeover statute representations, which must be true and accurate in all material respects.

The obligations of each of Motorola and Merger Sub to complete the merger depend on the satisfaction or waiver of, on or prior to the effective time of the merger, a number of additional conditions, including:

no action, investigation, proceeding or litigation has been instituted, commenced, pending or threatened:

or related order enacted or entered in which a governmental entity challenges or seeks to restrain or prohibit the completion of the merger or to prohibit or impair Motorola s ability to own or operate any of our businesses and assets after the merger or any of the businesses or assets of Motorola or its subsidiaries, including through any divestiture, licensing, lease or hold separate arrangement, or to limit Motorola s ability to exercise ownership rights with respect to the stock of the surviving corporation, or

which may reasonably be expected to result in the imposition of (1) criminal sanctions on us or our subsidiaries, or (2) material penalties or fines to a governmental entity, or restitution to a third party, in each case as a result of any conviction of us or our subsidiaries of a crime, or settlement with a governmental entity for the purpose of closing an investigation, being imposed on Motorola or the surviving corporation or any of their respective affiliates;

all authorizations, consents, orders or approvals of, or declarations, notices or filings with, or expirations of waiting periods imposed by, any governmental entity in connection with the merger having been made or obtained, except those that the failure to make or obtain could not have a material adverse effect on Motorola

or a material adverse effect (as defined below) on us or provide a reasonable basis to conclude that the parties to the merger agreement or any of their affiliates would be subject to risk of criminal sanctions or any of their representatives would be subject to the risk of criminal or material civil sanctions;

the continued enforceability of specified contracts of ours;

since the date of the merger agreement, no occurrence of any change, event, circumstance or development that has had, or could reasonably be expected to have, a material adverse effect (as defined below) on us;

neither our principal executive officer nor our principal financial officer has failed to provide the necessary certifications required under Section 302 and Section 906 of the Sarbanes-Oxley Act on any of our reports filed with the SEC since the date of the merger agreement;

#### **Table of Contents**

we have filed all registration statements, forms, reports and other documents required to be filed by us with the SEC on or prior to the closing date, and all required amendments, and the amendments shall not contain any information that is materially and adversely different from the information in the documents amended by such amendments;

since the date of the merger agreement, no action has been instituted or threatened challenging the validity and ownership of our intellectual property, other than actions the outcomes of which are not reasonably expected to result in a material adverse effect (as defined below) on us;

the SEC shall not have recommended any charges or enforcement action against us or our officers or directors, authorized such recommendation, or issued a Wells Notice to us or our officers or directors in connection with the SEC investigation relating to certain accounting matters that was commenced in December 2005 and recently discontinued, and there shall not be any pending governmental action, suit, proceeding or investigation against any of our or our subsidiaries directors, officers or certain key employees relating to the SEC investigation;

we must have at least \$15 million of cash and cash equivalents (\$13 million if the merger is completed after July 31, 2007, and, in either case, reduced by the amount of all reasonable merger-related expenses (including payments required on or prior to, or as a result of, completion of the merger) paid after April 21, 2007), net of indebtedness;

three-quarters of a small group of key employees, each of whom entered into a retention agreement with Motorola on April 21, 2007, and 80% of a larger group of employees, still being employed by us and performing their usual and customary duties immediately prior to completion of the merger; and

we must have filed specified tax returns and other tax returns, if any, that are identified as being delinquent prior to completion of the merger.

The merger agreement provides that a material adverse effect on us means any change or effect on us that is reasonably likely to prevent us from completing the merger and other transactions contemplated by the merger agreement, or any material adverse change in, or materially adverse effect on, either individually or in the aggregate with all such other adverse changes in or effects on, the condition (financial or otherwise), results of operations, operations, business, assets (including intangible assets) or liabilities of us and our subsidiaries taken as a whole.

In determining whether a material adverse effect has occurred, none of the following, in and of themselves, will constitute a material adverse effect on us:

changes or effects that are primarily the result of general economic or business conditions, or conditions in financial or securities markets, in the United States;

changes or effects that are primarily the result of factors generally affecting the industries or markets in which we operate;

changes or effects that result from the public announcement or pendency of the merger on our employees or the public announcement by Motorola with respect to our or our subsidiaries businesses;

changes resulting from or arising out of actions taken pursuant to (and/or required by) the merger agreement or at the request of Motorola, or the failure to take any actions due to restrictions set forth in the merger

agreement (provided that only if we (i) reasonably believe that taking any action required by the merger agreement or at the request of Motorola, or failing to take action prohibited by the agreement, could reasonably be expected to result in a material adverse effect on us, and (ii) provide timely prior written notification to Motorola of such belief and Motorola does not provide timely relief from the provisions of the merger agreement or its request, will the changes or effects resulting from such action or inaction be deemed not to constitute a material adverse effect on us);

changes in the price or trading volume of our common stock (provided that this exception will not prevent or otherwise affect a determination that any change, effect, circumstance or development underlying these changes has or has not resulted in, or contributed to, a material adverse effect on us, and no changes will be

used as evidence that some other change, effect, circumstance or development has had or has not had a material adverse effect on us); or

adverse changes arising from or relating to any change in GAAP or any change in applicable laws, in each case proposed, adopted or enacted after the date of the merger agreement.

The merger is not conditioned upon Motorola or Merger Sub obtaining financing. The total amount of funds necessary to pay the merger consideration will be approximately \$140 million. Motorola expects to fund the cash requirements for the transaction primarily from cash on hand.

### No Solicitation of Other Acquisition Proposals by Terayon

The merger agreement provides that we will not, nor will we permit or authorize any of our subsidiaries or any officer, director, employee, accountant, counsel, financial advisor, agent or other representative of us or any of our subsidiaries, directly or indirectly, or authorize or direct any of our representatives, to:

solicit, initiate, facilitate, respond to or encourage any inquiries regarding or relating to, or the submission of, any takeover proposal, as defined below;

participate in any discussions or negotiations, furnish to any person any information or data relating to us or our subsidiaries, provide access to our properties, books, records or employees or take any other action, in each case regarding or to facilitate the making of any proposal that constitutes, or may be reasonably be expected to lead to, any takeover proposal;

enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or other similar agreement or commitment with respect to any takeover proposal, or agree to, approve, endorse or resolve to recommend any takeover proposal;

grant any waiver or release under any standstill or similar agreement by any third party who has made a takeover proposal; or

take any action to exempt any third party from the restrictions on business combinations contained in Section 203 of the Delaware General Corporation Law or otherwise cause such restrictions not to apply.

Notwithstanding any of these restrictions, we may make certain disclosures to our stockholders required under applicable law, including to comply with our board of directors fiduciary duties, and we may provide notice of these restrictions to any person.

The merger agreement provides that we will, and will cause or representatives and our subsidiaries to, immediately terminate any existing activities, discussions, solicitations or negotiations with any person conducted up to the date of the merger agreement with respect to a takeover proposal.

However, if at any time before our stockholders vote to adopt the merger agreement and approve the merger:

we receive a bona fide written takeover proposal after the date of the merger agreement that our board of directors determines, after consultation with outside counsel, constitutes a superior proposal (as defined below) or is more favorable to our stockholders from a financial point of view than the merger and is reasonably likely to lead to a superior proposal; and

the takeover proposal was made and not solicited after the date of the merger agreement, and did not otherwise result from our breach of our nonsolicitation obligations (other than unintentional breaches that did not result in the takeover proposal and had only an immaterial impact on Motorola s rights under the nonsolicitation provisions);

#### Table of Contents

we may, if our board of directors determines in good faith, after receiving advice from outside counsel, that such action is required to discharge our board of directors fiduciary duties to our stockholders and, subject to providing prompt oral and written notice of our decision to do so to Motorola:

furnish, under a confidentiality agreement substantially similar to the one signed by Motorola, information about us and our subsidiaries that is not of greater scope, area, or detail than was provided to Motorola to the person making such takeover proposal and its representatives; and

negotiate and participate in discussions with the party making the bona fide, written takeover proposal.

The merger agreement provides that:

the term takeover proposal means any inquiry, proposal, offer or indication of interest from any person other than Motorola or any of its affiliates that constitutes, or that would reasonably be expected to lead to, any transaction to acquire beneficial ownership of (1) assets that constitute 15% or more of our and our subsidiaries consolidated revenues, net income or assets, or (2) 15% or more of any class of our or of any of our subsidiaries equity securities pursuant to a transaction or series of transactions, including:

a merger, consolidation, share exchange, or other business combination involving us or our subsidiaries;

a sale, issuance, exchange, transfer or other disposition of our or our subsidiaries shares of capital stock;

a sale, lease, license, exchange, transfer or other disposition of our or our subsidiaries assets; or

a tender offer or exchange offer with respect to either us or any of our subsidiaries, which is structured to permit another person to acquire beneficial ownership of assets that constitute 15% or more of our and our subsidiaries consolidated revenues, net income or assets or 15% or more of the equity interest in either us or any of our subsidiaries; and

the term superior proposal means an unsolicited written proposal or offer made by a third party to acquire, directly or indirectly, greater than 50% of the shares of our outstanding common stock (or would result in our stockholders beneficially owning less than 50% of the voting power of the combined or ongoing entity) or to acquire all or substantially all of our assets, in each case which our board of directors determines in good faith (after consultation with its financial advisors) to be more favorable to our stockholders from a financial point of view than the merger, taking into account all relevant terms and conditions of such proposal or offer as it deems relevant, and which, in the good faith reasonable judgment of our board of directors, is reasonably likely to be completed.

In the event that we receive any takeover proposal or a communication reasonably likely to lead to one, we must promptly (within one business day) notify Motorola orally and in writing of the terms and conditions of the proposal and the identity of the person making it, and provide Motorola with a copy of the written proposal. We must also promptly (within 24 hours) notify Motorola of any change to the material terms of any takeover proposal or of any determination by our board of directors that a takeover proposal constitutes a superior proposal.

The merger agreement further provides that our board of directors will not:

withdraw or modify, or propose to withdraw or modify, in any manner adverse to Motorola or Merger Sub, the board s recommendation that our stockholders vote in favor of adoption of the merger agreement and approval

of the merger, or approve, recommend or propose to approve or recommend any takeover proposal; or

enter into, any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement, option agreement or other similar agreement or commitment with respect to any takeover proposal.

However, at any time when a superior proposal is pending and prior to our stockholders vote on the merger, if our board of directors determines in good faith (after receiving advice from outside counsel) that such action is required to discharge its fiduciary duties to our stockholders, our board of directors may change or withdraw its

#### Table of Contents

recommendation in a manner adverse to Motorola or Merger Sub and/or approve or recommend a superior proposal, and we may enter into a letter of intent, memorandum of understanding, agreement in principle, acquisition or similar agreement with respect to the superior proposal. However, before doing so, our board of directors is required to give Motorola three business days prior written notice (which notice must state the material terms and conditions of the superior proposal and the identity of the person making it), and a further three business days notice in the event of any material amendment to the financial terms or any other material term of the superior proposal, and during the three business days, Motorola must not have made a matching offer. A matching offer is one that is at least as favorable to our stockholders as the superior proposal, as concluded by our board of directors in its good faith judgment, after consultation with its financial advisors and receipt of advice from outside counsel.

For a discussion of the circumstances under which we must pay a \$5.25 million termination fee, see Termination; Payment of Termination Fee below.

#### **Termination; Payment of Termination Fee**

The merger agreement may be terminated at any time prior to the effective time of the merger, regardless of whether our stockholders have adopted the merger agreement:

by mutual written consent of Motorola and us;

by either Motorola or us, by written notice if:

the merger has not been completed on or before September 21, 2007, or a later date as extended by mutual written consent of Motorola and us, unless the party that seeks to terminate has breached or failed to perform in any material respect its obligations under the merger agreement in any manner that has been the principal cause of, or has primarily resulted in, the merger not being completed by that date;

our stockholders do not adopt the merger agreement and approve the merger at the special meeting, or at any adjournment or postponement thereof, unless the party that seeks to terminate has breached or failed to perform in any material respect its obligations under the merger agreement in any manner that has been the principal cause of, or has primarily resulted in, the merger not being completed by that date; or

any order permanently restraining, enjoining or otherwise prohibiting completion of the merger becomes final and nonappealable (provided that the party that seeks to terminate has used commercially reasonable efforts to have the order lifted);

#### by Motorola if:

our board fails to recommend approval of the merger agreement and the merger in this proxy statement, changes, or resolves to change, its recommendation to adopt the merger agreement and approve the merger, recommends to our stockholders a competing transaction or publicly announces that it intends to do so, or enters into any alternative acquisition agreement accepting any competing transaction;

a tender offer or exchange offer for our outstanding shares of capital stock is commenced, and our board fails to recommend to our stockholders against accepting the offer;

our board, upon request of Motorola following receipt of a proposal or offer for a competing transaction, fails to reaffirm to Motorola the approval or recommendation of the merger and the merger agreement within five business days of the request;

we or any of our officers, directors, representatives or agents knowingly and materially breaches our obligations under the non-solicitation provisions or specified provisions relating to the special meeting of our stockholders in the merger agreement; or

we have breached any of our representations, warranties, covenants or agreements in the merger agreement, or any of our representations or warranties becomes untrue after the date of the merger agreement, so that the related closing condition would not be satisfied, and we do not cure the breach within 20 days after we receive written notice of it from Motorola (although Motorola may not terminate the merger agreement if Motorola or Merger Sub is in material breach of the merger agreement);

52

#### by us if:

prior to the stockholder vote with respect to the adoption of the merger agreement and approval of the merger, our board of directors, in compliance with the nonsolicitation provisions of the merger agreement, has approved or recommended to our stockholders a superior proposal, Motorola has failed to match the superior proposal within three business days after we notify Motorola of the superior proposal, and we have paid Motorola the termination fee described below; or

Motorola or Merger Sub has breached any of their representations, warranties, covenants or agreements in the merger agreement, or any of their representations or warranties becomes untrue after the date of the merger agreement, so that the relating closing condition would not be satisfied, and they do not cure the breach within 20 days after they receive written notice of it from us (although we may not terminate the merger agreement if we are in material breach of the merger agreement).

If the merger agreement is terminated pursuant to the above provisions, the merger agreement will become void and of no effect, without any liability or obligation on the part of Motorola, Merger Sub or us, except with respect to specified provisions of the merger agreement, including those regarding confidentiality, fees and expenses, and liability for fraud or willful or intentional breach of the merger agreement.

We will be required to pay Motorola a \$5.25 million termination fee if:

the merger agreement is terminated by either Motorola or us because either the merger has not been completed prior to September 21, 2007, or our stockholders have not adopted the merger agreement at the special meeting, and

after the date of the merger agreement but prior to the termination date (in the case of termination for failure to complete the merger prior to September 21, 2007) or the special meeting date (in the case of termination for failure to obtain the requisite stockholder vote), a third party makes, or publicly discloses or announces an intention to make, a takeover proposal involving 35% or more of our and our subsidiaries consolidated revenues, net income or assets, or 35% or more of any class of our or of any of our subsidiaries equity securities; and

within 12 months of such termination of the merger agreement, we or any of our subsidiaries enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or other similar agreement or commitment to consummate, or approves or recommends to our stockholders or otherwise does not oppose, a transaction fitting the above description with the third party.

the merger agreement is terminated by Motorola as a result of the occurrence of any of the following, or by us, if our stockholders fail to adopt the merger agreement at the special meeting or any adjournment or postponement thereof and, prior to the date of the special meeting:

our board fails to recommend adoption of the merger agreement and approval of the merger in this proxy statement, changes, or resolves to change, its recommendation of the merger, recommends to the stockholders a competing transaction or publicly announces that it intends to do so, or enters into any alternative acquisition agreement accepting any competing transaction;

a tender offer or exchange offer for our outstanding shares of capital stock is commenced, and our board fails to recommend against our stockholders accepting it;

our board, upon request of Motorola following receipt of a proposal or offer for a competing transaction, fails to reaffirm to Motorola the approval or recommendation of the merger and the merger agreement within five business days of such request; or

we or any of our officers, directors, representatives or agents knowingly and materially breaches our obligations under the non-solicitation provisions or specified provisions relating to the stockholders meeting in the merger agreement.

#### **Table of Contents**

the merger agreement is terminated by us prior to the stockholder vote in connection with our board of directors, in compliance with the nonsolicitation provisions of the merger agreement, approving or recommending to our stockholders a superior proposal, and Motorola failing to match the superior proposal within three business days after we notify Motorola of it.

If the termination fee is paid, it is Motorola s and Merger Sub s sole and exclusive remedy upon termination of the merger agreement.

#### **Fees and Expenses**

Except for circumstances in which we are required to pay the termination fee described above, the merger agreement generally provides that each party will pay its own fees and expenses in connection with the merger agreement, whether or not the merger is completed.

#### Amendment, Extension and Waiver

At any time prior to the completion of the merger, the merger agreement may be amended, modified or supplemented by mutual consent of the parties in writing. However, after the merger agreement has been adopted by our stockholders, there may not be any amendment, modification or supplement of the merger agreement without our stockholders further approval, if their approval to the amendment, modification or supplement is required by law. At any time prior to the completion of the merger, the parties may, to the extent permitted by law, waive compliance with any condition in the merger agreement.

54

### PROPOSAL TO ADJOURN THE SPECIAL MEETING

#### **The Adjournment Proposal**

If at the special meeting of stockholders the number of shares of our common stock represented and voting in favor of adoption of the merger agreement and approval of the merger is insufficient to adopt that proposal under the Delaware General Corporation Law, we intend to move to adjourn the special meeting in order to enable our board of directors to solicit additional proxies in respect of such proposal. In that event, we will ask our stockholders to vote only upon the adjournment proposal, and not the proposal regarding adoption of the merger agreement and approval of the merger.

In this proposal, we are asking you to authorize the holder of any proxy solicited by our board of directors to vote in favor of granting discretionary authority to the proxy or attorney-in-fact to adjourn the special meeting to another time and place for the purpose of soliciting additional proxies. If the stockholders approve the adjournment proposal, we could adjourn any adjourned session of the special meeting and use the additional time to solicit additional proxies, including the solicitation of proxies from stockholders that have previously voted. Among other things, approval of the adjournment proposal could mean that, even if we had received proxies representing a sufficient number of votes against adoption of the merger agreement and approval of the merger to defeat that proposal, we could adjourn the special meeting without a vote on the merger agreement and seek to convince the holders of those shares to change their votes to votes in favor of adoption of the merger agreement and approval of the merger.

#### **Vote Required and Board Recommendation**

The proposal to adjourn the special meeting for the purpose of soliciting additional proxies, if necessary, will be approved if the votes cast in favor of the proposal by shares of common stock, present in person or represented by proxy and entitled to vote on the subject matter, exceed the votes cast against the proposal. No proxy that is specifically marked AGAINST adoption of the merger agreement and approval of the merger will be voted in favor of the adjournment proposal, unless it is specifically marked FOR the adjournment proposal.

Our board of directors recommends that you vote FOR the adjournment proposal.

55

### SECURITY OWNERSHIP OF MANAGEMENT AND PRINCIPAL SHAREHOLDERS

The following table sets forth certain information regarding the ownership of our common stock as of May 18, 2007 by: (i) each director; (ii) each of our named executive officers; (iii) all of our named executive officers and directors as a group; and (iv) all those known by us to be beneficial owners of more than five percent of our common stock. All shares of our common stock subject to options currently exercisable or exercisable within 60 days of May 18, 2007, are deemed to be outstanding for the purpose of computing the percentage of ownership of the person holding such options, but are not deemed to be outstanding for computing the percentage of ownership of any other person. This table is based upon information supplied by our officers, directors and principal stockholders and Schedules 13D and 13G filed with the SEC. Unless otherwise indicated in the footnotes to this table and subject to community property laws where applicable, we believe that each of the stockholders named in this table has sole voting and investment power with respect to the shares indicated as beneficially owned. Applicable percentages are based on 77,637,177 shares outstanding on May 18, 2007, adjusted as required by rules promulgated by the SEC. Unless otherwise indicated in the table, the address of each party listed in the table is 2450 Walsh Avenue, Santa Clara, California 95051.

	Beneficial Ownership Number of Percentage	
Beneficial Owner	Shares	Ownership
Kern Capital Management, LLC(1)	11,538,200	14.9%
114 West 47th Street, Suite 1926		
New York, New York 10036		
Zaki Rakib(2)	8,604,080	11.1%
Shlomo Rakib(3)	8,604,080	11.1%
Jerry D. Chase(4)	566,666	*
Lewis Solomon(5)	358,859	*
Mark A. Richman(6)	322,916	*
Matthew J. Aden(7)	239,583	*
David M. Woodrow(8)	134,969	*
Howard W. Speaks, Jr.(9)	77,723	*
Matthew Miller(10)	67,294	*
All executive officers and directors as a group (9 persons)(11)	10,372,090	13.4%

- Kern Capital Management, LLC filed an amendment to Schedule 13G, dated as of February 14, 2007, with the SEC. Kern Capital Management, LLC reported beneficial ownership of 11,538,200 shares of our common stock.
- (2) Shares beneficially owned by Dr. Zaki Rakib include 2,762,040 shares of our common stock (representing 3.6% of the shares outstanding on May 18, 2007), 240,000 shares of our common stock held by the Shlomo Rakib Children s Trust of which Dr. and Mrs. Rakib are trustees, and 1,300,000 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007. Shares beneficially owned also include 2,762,040 shares of our common stock held by Shlomo Rakib, 1,300,000 shares of our common stock underlying stock options that are exercisable by Shlomo Rakib, 1,300,000 shares of our common stock underlying stock options that are exercisable by Shlomo Rakib within 60 days of May 18, 2007, and 240,000 shares of our common stock held by the Zaki Rakib Children s Trust. Zaki Rakib disclaims beneficial

ownership of shares of our common stock, including shares underlying stock options, held by each of Shlomo Rakib and the Zaki Rakib Children s Trust. Zaki Rakib has informed us that he may transfer by gift up to 500,000 shares of common stock to a university.

(3) Shares beneficially owned by Shlomo Rakib include 2,762,040 shares of our common stock (representing 3.6% of the shares outstanding on May 18, 2007), 240,000 shares of our common stock held by the Zaki Rakib Children s Trust of which Mr. and Mrs. Rakib are trustees, and 1,300,000 shares of our common stock underlying stock options, which are exercisable within 60 days of May 18, 2007. Shares beneficially owned also include 2,762,040 shares of our common stock held by Zaki Rakib, 1,300,000 shares of our common stock underlying stock options that are exercisable by Zaki Rakib within 60 days of May 18, 2007, and

56

240,000 shares of our common stock held by the Shlomo Rakib Children s Trust. Shlomo Rakib disclaims beneficial ownership of shares of our common stock, including shares underlying stock options, held by each of Zaki Rakib and the Shlomo Rakib Children s Trust. Shlomo Rakib has informed us that he may transfer by gift up to 400,000 shares of our common stock to a university.

- (4) Shares beneficially owned include 566,666 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007.
- (5) Shares beneficially owned include 298,859 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007, as well as 60,000 shares of our common stock.
- (6) Shares beneficially owned include 322,916 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007.
- (7) Shares beneficially owned include 239,583 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007.
- (8) Shares beneficially owned include 134,969 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007.
- (9) Shares beneficially owned include 77,723 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007.
- (10) Shares beneficially owned include 67,294 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007.
- (11) Shares beneficially owned by our current directors and executive officers as a group include 4,308,010 shares of our common stock underlying stock options that are exercisable within 60 days of May 18, 2007.

### **OTHER MATTERS**

As of the date of this proxy statement, we know of no matters that will be presented for consideration at the special meeting other than as described in this proxy statement. If, however, other matters are brought before the special meeting, the persons named as proxies will vote in accordance with their judgment on such other matters unless otherwise indicated on the proxy.

#### FUTURE STOCKHOLDER PROPOSALS

Pursuant to Rule 14a-8 under the Securities Exchange Act of 1934, a stockholder who wishes to submit a proper proposal for inclusion in our proxy statement for our annual meeting of stockholders to be held later this year must submit the proposal in writing in a timely manner to our Secretary at our principal executive offices. In order to be included, the proposal must be received by us a reasonable time before we begin to print and mail our annual proxy materials. The proposal must also otherwise comply with Rule 14a-8, including the requirement that the proponent must have continuously held at least \$2,000 in market value or 1% of our common stock for at least one year prior to the date the proposal is submitted.

In addition, under our bylaws, a stockholder who wishes to otherwise bring a proposal for consideration at our 2007 annual meeting of stockholders must have submitted the proposal in writing to our Secretary not earlier than the close of business on the ninetieth (90th) day prior to the annual meeting and not later than the close of business on the

sixtieth (60th) day prior to the annual meeting. In the event the public announcement of the date of the annual meeting is first made by us fewer than seventy (70) days prior to the date of the annual meeting, a stockholder who wishes to otherwise bring a proposal for consideration at our 2007 annual meeting of stockholders must have submitted the proposal in writing to our Secretary by the close of business on the tenth (10th) day following the day on which public announcement of the date of the meeting is first made by us. The proposal must also otherwise comply with the requirements set forth in our bylaws.

### HOUSEHOLDING OF PROXY STATEMENT

The SEC has adopted rules that permit companies and intermediaries, such as brokers, to satisfy delivery requirements for proxy statements and annual reports with respect to two or more stockholders sharing the same address by delivering a single proxy statement and annual report addressed to those stockholders. This process, known as householding, potentially means extra convenience for stockholders and cost savings for companies. This year, a number of brokers with customers who are our stockholders will be householding our proxy materials unless contrary instructions have been received from the customers. We will promptly deliver, upon oral or written request, a separate copy of this proxy statement to any stockholder sharing an address to which only one copy was mailed. Requests for additional copies should be directed to Mark A. Richman, Chief Financial Officer, c/o Terayon Communication Systems, Inc., 2450 Walsh Avenue, Santa Clara, CA 95051, or by telephone at (408) 235-5500.

Once a stockholder has received notice from his or her broker that the broker will be householding communications to the stockholder s address, householding will continue until the broker is notified otherwise or until the stockholder revokes his or her consent. If, at any time, a stockholder no longer wishes to participate in householding and would prefer to receive separate copies of this proxy statement, the stockholder should so notify his or her broker. Any stockholder who currently receives multiple copies of a proxy statement and annual report at his or her address and would like to request householding of communications should contact his or her broker or, if shares are registered in the stockholder s name, us at the address or telephone number provided above.

### WHERE YOU CAN FIND ADDITIONAL INFORMATION

We file annual, quarterly and current reports, proxy statements and other information with the Securities and Exchange Commission. You may read and copy any document we file with the SEC at the SEC s public reference room located at 100 F. Street, N.E., Washington, D.C., 20549. Please call the SEC at 1-800-SEC-0330 for further information on the SEC s public reference rooms. Our SEC filings also are available to the public at the SEC s website at http://www.sec.gov.

Any person, including any beneficial owner, to whom this proxy statement is delivered may request copies of reports, proxy statements or other information concerning us, without charge, by written request, directed to Mark A. Richman, Chief Financial Officer, c/o Terayon Communication Systems, Inc., 2450 Walsh Avenue, Santa Clara, CA 95051, or by calling (408) 235-5500. If you would like to request documents, please do so by June 14, 2007 in order to receive them before the special meeting.

THIS PROXY STATEMENT DOES NOT CONSTITUTE THE SOLICITATION OF A PROXY IN ANY JURISDICTION TO OR FROM ANY PERSON TO WHOM OR FROM WHOM IT IS UNLAWFUL TO MAKE SUCH PROXY SOLICITATION IN SUCH JURISDICTION. YOU SHOULD RELY ONLY ON THE INFORMATION CONTAINED IN THIS PROXY STATEMENT TO VOTE YOUR SHARES AT THE SPECIAL MEETING. WE HAVE NOT AUTHORIZED ANYONE TO PROVIDE YOU WITH INFORMATION THAT IS DIFFERENT FROM WHAT IS CONTAINED IN THIS PROXY STATEMENT. THIS PROXY STATEMENT IS DATED MAY 24, 2007. YOU SHOULD NOT ASSUME THAT THE INFORMATION CONTAINED IN THIS PROXY STATEMENT IS ACCURATE AS OF ANY DATE OTHER THAN THAT DATE, AND THE MAILING OF THIS PROXY STATEMENT TO STOCKHOLDERS DOES NOT CREATE ANY IMPLICATION TO THE CONTRARY.

Annex A

**EXECUTION VERSION** 

### AGREEMENT AND PLAN OF MERGER AMONG MOTOROLA, INC., MOTOROLA GTG SUBSIDIARY VI CORP. AND TERAYON COMMUNICATION SYSTEMS, INC. DATED AS OF APRIL 21, 2007

# **Table of Contents**

		Page
ARTICI	LE I THE MERGER	A-1
1.1	The Merger	A-1
1.2	Effective Time; Closing	A-1
1.3	Effect of the Merger	A-1
	LE II CERTIFICATE OF INCORPORATION AND BY-LAWS OF THE SURVIVING	
CORPO	DRATION	A-2
2.1	The Certificate of Incorporation	A-2
2.2	The By-Laws	A-2
ARTICI	LE III OFFICERS AND DIRECTORS OF THE SURVIVING CORPORATION	A-2
3.1	Directors	A-2
3.2	Officers	A-2
5.2		
ARTICI	LE IV CONVERSION OF SECURITIES	A-2
4.1	Conversion of Capital Stock	A-2
4.2	Exchange of Certificates	A-3
4.3	Company Options	A-4
4.4	Employee Stock Purchase Plans	A-4
4.5	Actions by the Company	A-5
4.6	Dissenting Shares	A-5
ARTICI	LE V REPRESENTATIONS AND WARRANTIES OF THE COMPANY	A-5
5.1	Organization, Good Standing and Qualification; Subsidiaries.	A-5
5.2	Capital Structure	A-7
5.3	Corporate Authority; Approval and Fairness	A-9
5.4	Governmental Filings; No Violations; Certain Contracts, Etc.	A-9
5.5	Contracts	A-10
5.6	SEC Filings; Financial Statements; Information Provided.	A-12
5.7	Absence of Certain Changes	A-14
5.8	Litigation	A-14
5.9	Employee Benefits	A-15
5.10	Compliance with Laws; Permits	A-17
5.11	Environmental Matters	A-18
5.12	Taxes	A-18
5.13	Employees; Independent Contractors	A-19
5.14	Insurance	A-21
5.15	Intellectual Property	A-21
5.16	Owned and Leased Properties	A-25
	-	

Government Contracts	A-27
Import and Export Control Laws	A-27
Foreign Corrupt Practices Act	A-28
Consent Decrees	A-28
Product Liability and Recalls	A-28
Takeover Statutes	A-29
	Import and Export Control Laws Foreign Corrupt Practices Act Consent Decrees Product Liability and Recalls

		Page
5.23	Change of Control	A-29
5.23	Vote Required	A-29
5.25	Company Rights Plan	A-29
5.26	Brokers and Finders	A-29 A-29
5.20	Blokers and Finders	A-29
ARTICL	E VI REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	A-29
6.1	Organization, Good Standing and Qualification	A-30
6.2	Authority; No Conflict; Required Filings and Consents	A-30
6.3	Information Provided	A-31
6.4	Operations of Merger Sub	A-31
6.5	Brokers and Finders	A-31
6.6	Financing	A-31
ΛΡΤΙΟΙ	E VII COVENANTS	A-31
7.1	Interim Operations	A-31 A-31
7.1	No Solicitation	A-31 A-33
7.2	Proxy Statement	A-35 A-35
7.4	Company Meeting	A-35 A-36
7.5	Filings; Other Actions; Notification	A-30 A-36
7.6	Access	A-30 A-38
7.7	Notice of Certain Matters	A-38
7.8	Removal of Company Common Stock From Pink Sheets	A-38
7.9	Publicity	A-39
7.10	Company and Parent Benefit Plans	A-39 A-39
7.10		A-39 A-39
7.11	Loans to Company Employees, Officers and Directors Indemnification; Directors and Officers Insurance	A-39 A-39
7.12	Takeover Statute	A-39 A-40
7.13	Section 16 Matters	A-40 A-40
7.14		A-40 A-40
7.15	Litigation Insurance Policy Tax Returns	A-40 A-40
ARTICL	E VIII CONDITIONS	A-41
8.1	Conditions to Each Party s Obligation to Effect the Merger	A-41
8.2	Conditions to Obligations of Parent and Merger Sub	A-42
8.3	Conditions to Obligation of the Company	A-44
	ΕΙΥ ΤΕΡΜΙΝΙΑΤΙΟΝΙ	A / A
	E IX TERMINATION	A-44
9.1	Termination by Mutual Consent	A-44
9.2	Termination by Either Parent or the Company	A-44
9.3	Termination by the Company	A-45
9.4	Termination by Parent	A-45
9.5	Effect of Termination and Abandonment	A-46

# Table of Contents

ARTICL	LE X MISCELLANEOUS AND GENERAL	A-47
10.1	Survival	A-47
10.2	Modification or Amendment	A-47
10.3	Waiver of Conditions	A-47

		Page
10.4	Counterparts	A-47
10.5	GOVERNING LAW AND VENUE; WAIVER OF JURY TRIAL	A-47
10.6	Notices	A-48
10.7	Entire Agreement	A-49
10.8	No Third Party Beneficiaries	A-49
10.9	Obligations of Parent and of the Company	A-49
10.10	Definitions	A-49
10.11	Severability	A-49
10.12	Interpretation; Construction	A-49
10.13	Assignment	A-49
10.14	Expenses	A-50

	٠
A-11	1

# **DEFINED TERMS**

Term	Section
1995 Plan	5.2(a)
1997 Plan	5.2(a)
1998 ESPP	4.4(a)
1998 Plan	5.2(a)
1999 Plan	5.2(a)
Actions	5.8(a)
Adverse Recommendation Notice	7.2(c)(i)
Affiliate	5.2(d)
Agreement	Preamble
Alternative Acquisition Agreement	7.2(a)
Antitrust Laws	7.5(b)
Apportionment Schedule	7.16(a)
Bid	5.17
Burdensome Condition	8.2(c)(i)
Business Day	1.2
By-Laws	2.2
Certificate	4.2(b)
Certificate of Merger	1.2
Change in Company Recommendation	7.2(b)
Charter	2.1
Closing	1.2
Closing Date	1.2
Code	4.2(f)
Company	Preamble
Company Approvals	5.4(a)
Company Benefit Plans	5.9(a)
Company Board	5.1(a)
Company Board Recommendation	5.3(b)
Company Common Stock	4.1(b)
Company Disclosure Schedule	Article V
Company ERISA Plans	5.9(b)
Company ESP Plans	4.4(a)
Company Government Contract	5.17
Company Government Subcontract	5.17
Company Lease	5.16(b)
Company Material Adverse Effect	5.1(d)
Company Material Contract	5.5(a)
Company Meeting	7.4
Company Net Cash	5.8(c)
Company Non-U.S. Benefit Plan	5.9(f)
Company Options Plans	5.2(a)
Company Permit	5.9(b)
Company Permit	5.10
Company Representatives	7.2(a)

Company Rights Agreement	5.2(c)
Company SEC Reports	5.6(a)

Term	Section
Company Software	5.15(h)
Company Stock Option	5.2(a)
Company Stock Plans	5.2(a)
Company Triggering Event	9.4(b)
Competing Transaction	7.2(c)(ii)
Confidentiality Agreement	10.7
Constituent Corporations	Preamble
Contracts	5.4(b)
Copyrights	5.15(q)(i)
Costs	7.12(a)
Delaware Law	Recitals
Dissenting Shares	4.6(a)
Effective Time	1.2
Employees	5.13(a)
Environmental Law	5.11(a)
ERISA	5.9(a)
ERISA Affiliate	5.9(a)
Exchange Act	5.4(a)
Exchange Agent	4.2(a)
Exchange Fund	4.2(a)
Expenses	10.14
Export Approvals	5.18(a)
FCPA	5.19
GAAP	5.2(e)
Governmental Entity	5.4(a)
Hazardous Substance	5.11(a)
HSR Act	5.1(d)
Indebtedness	5.8(c)
Indemnified Parties	7.12(a)
Intellectual Property	5.15(q)(ii)
Investments	5.1(c)
IRS	5.9(b)
Key Employee	5.13(c)
knowledge	10.12
Laws	5.10
Leased Real Property	5.16(b)
Liens	5.1(d)
Limited License	5.15(o)
Litigation Insurance Policy	7.15
Material Environmental Reports	5.11(a)
Major Customer	5.5(a)(iii)
Major Customer Contract	5.5(a)(iii)
Major Supplier	5.5(a)(v)
Maximum Premium	7.12(b)
Merger	1.1
Merger Consideration	4.1(c)

Merger Sub

Preamble

Term	Section
NASD	5.6(f)
Option Holder	4.3
Option Payment	4.3
Order	8.1(c)
Outside Date	9.2(a)
Owned Intellectual Property	5.15(q)(iii)
Parent	Preamble
Parent Material Adverse Effect	6.1
Patents	5.15(q)(iv)
Permitted Liens	5.16(e)
Person	4.2(b)
Pink Sheets	5.2(a)
Preferred Shares	5.2(a)
Proxy Statement	5.6(d)
Restatement	8.2(h)
Rights	5.2(c)
Sarbanes-Oxley Act	5.6(a)
SEC	5.2(e)
SEC Investigation	8.2(j)
Section 7.16(a) Notice	7.16(a)
Securities Act	5.2(e)
Shareholder Approval	5.3(a)
Software	5.15(q)(v)
Stockholder Agreement	Recitals
Subsidiary	5.1(d)
Superior Proposal	7.2(c)(iii)
Surviving Corporation	1.1
Takeover Proposal	7.2(c)(iv)
Takeover Statute	5.22
Target Net Cash Amount	8.2(k)
Tax	5.12(b)
Tax Return	5.12(b)
Tenant	5.16(c)
Termination Fee	9.5(b)
Third Party	7.2(c)(v)
Third Party Embedded Software	5.15(c)
Third Party IP Licenses	5.15(d)
Third Party Licenses	5.15(d)
Third Party Software Licenses	5.15(c)
Trademarks	5.15(q)(vi)
Voting Debt	5.2(c)
Waiting Period	9.3(a)

#### Table of Contents

#### AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this <u>Agreement</u>) is dated as of April 21, 2007, by and among Terayon Communication Systems, Inc., a Delaware corporation (the <u>Company</u>), Motorola, Inc., a Delaware corporation (<u>Parent</u>), and Motorola GTG Subsidiary VI Corp., a Delaware corporation and a wholly-owned subsidiary of Parent (<u>Merger Sub</u>). The Company and Merger Sub are sometimes collectively referred to herein as the <u>Constituent Corporation</u>s.

#### RECITALS

WHEREAS, Parent and the respective boards of directors of Merger Sub and the Company have deemed it advisable and in the best interests of their respective corporations and stockholders that Parent and the Company consummate the business combination and other transactions provided for in this Agreement;

WHEREAS, the respective boards of directors of Merger Sub and the Company have approved, in accordance with the Delaware General Corporation Law (<u>Delaware Law</u>), this Agreement and the transactions contemplated hereby, including the Merger (as defined below);

WHEREAS, the board of directors of the Company has resolved to recommend to its stockholders approval and adoption of this Agreement and approval of the Merger;

WHEREAS, Parent, as the sole stockholder of Merger Sub, has approved and adopted this Agreement and approved the Merger pursuant to the terms and subject to the conditions set forth in this Agreement; and

WHEREAS, Parent, Merger Sub and the Company desire to make certain representations, warranties and agreements in connection with the Merger and also to prescribe certain conditions to the Merger;

NOW, THEREFORE, in consideration of the promises, representations, warranties, covenants and agreements contained in this Agreement and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, the parties agree as follows:

#### ARTICLE I

#### THE MERGER

1.1 <u>The Merger</u>. At the Effective Time (as defined below) and subject to and upon the terms and conditions of this Agreement and the applicable provisions of Delaware Law, Merger Sub will be merged with and into the Company (the <u>Merger</u>), the separate corporate existence of Merger Sub will cease and the Company will continue as the surviving corporation and as a wholly-owned subsidiary of Parent. The surviving corporation after the Merger is sometimes referred to as the <u>Surviving Corporation</u>.

1.2 Effective Time; Closing. Subject to the provisions of this Agreement, the parties shall cause the Merger to be consummated by filing a Certificate of Merger with the Secretary of State of the State of Delaware in accordance with the relevant provisions of Delaware Law (the <u>Certificate of Merger</u>) (the time of such filing with the Secretary of State of the State of Delaware or such later time as may be agreed in writing by the Company and Parent and specified in the Certificate of Merger is referred to as the <u>Effective Time</u>) on the Closing Date. The closing of the Merger (the <u>Closing</u>) shall take place at the offices of Winston & Strawn LLP, 35 West Wacker Drive, Chicago, Illinois, at a time and date to be specified by the parties, which shall be no later than the second Business Day after the satisfaction or waiver of the conditions set forth in <u>Article VIII</u> (other than those that by their terms are to be satisfied or waived at

the Closing), or at such other time, date and location as the parties agree in writing. The date on which the Closing occurs is referred to as the <u>Closing Date</u>. <u>Business</u> Day means each day that is not a Saturday, Sunday or other day on which Parent is closed for business or banking institutions located in Chicago, Illinois are authorized or obligated by law or executive order to close.

1.3 <u>Effect of the Merger</u>. At the Effective Time, the effect of the Merger will be as provided in this Agreement and the applicable provisions of Delaware Law. Without limiting the generality of the foregoing, at the Effective Time all the property, rights, privileges, powers and franchises of the Company and Merger Sub will vest

#### **Table of Contents**

in the Surviving Corporation, and all debts, obligations, claims, liabilities and duties of the Company and Merger Sub will become the debts, obligations, claims, liabilities and duties of the Surviving Corporation.

#### ARTICLE II

#### <u>CERTIFICATE OF INCORPORATION AND BY-LAWS</u> <u>OF THE SURVIVING CORPORATION</u>

2.1 <u>The Certificate of Incorporation</u>. At the Effective Time, the certificate of incorporation of the Company in effect immediately prior to the Effective Time shall be amended and restated in its entirety to be identical to the certificate of incorporation of the Merger Sub (the <u>Charter</u>) attached hereto <u>as Exhibit</u> A, until later amended as provided in the Charter or by applicable Law; <u>provided</u>, <u>however</u>, that at the Effective Time, Article I of the certificate of incorporation of the Surviving Corporation will be amended and restated in its entirety to read as follows: The name of the corporation is Terayon Communication Systems, Inc. . After the Effective Time, the authorized capital stock of the Surviving Corporation shall consist of 1,000 shares of common stock, par value \$0.01 per share.

2.2 <u>The By-Laws</u>. At the Effective Time, the by-laws of the Company in effect at the Effective Time will be amended and restated in their entirety to be identical to the by-laws of Merger Sub, as in effect immediately prior to the Effective Time (the <u>By-Laws</u>), until later amended as provided in the By-Laws or by applicable Law.

#### ARTICLE III

### OFFICERS AND DIRECTORS OF THE SURVIVING CORPORATION

3.1 <u>Directors</u>. The directors of Merger Sub at the Effective Time will, from and after the Effective Time, be the directors of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Charter and the By-Laws, and the board of directors of the Company shall take all such actions as may be necessary or appropriate to give effect to the foregoing.

3.2 <u>Officers</u>. The officers of Merger Sub at the Effective Time will, from and after the Effective Time, be the officers of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Charter and the By-Laws.

#### ARTICLE IV

#### **CONVERSION OF SECURITIES**

4.1 <u>Conversion of Capital Stock</u>. As of the Effective Time, by virtue of the Merger and without any action on the part of Merger Sub, the Company or any holder of shares of the capital stock of the Company or capital stock of Merger Sub, the following will occur:

(a) <u>Capital Stock of Merger Sub</u>. Each share of common stock, par value \$0.01 per share, of Merger Sub issued and outstanding immediately prior to the Effective Time will be converted into and become one fully paid and nonassessable share of common stock, \$0.01 par value per share, of the Surviving Corporation.

(b) <u>Cancellation of Treasury Stock and Parent-Owned Stock</u>. All shares of common stock, par value \$0.001 per share, of the Company (<u>Company Common Stock</u>) that are owned by the Company as treasury stock and any shares of Company Common Stock owned by Parent or Merger Sub or any direct or indirect Subsidiaries of Parent immediately prior to the Effective Time will be cancelled and will cease to exist and no payment will be made with respect thereto.

(c) <u>Merger Consideration for Company Common Stock</u>. Subject to <u>Section 4.2</u>, each share of Company Common Stock (other than shares to be cancelled in accordance with <u>Section 4.1(b)</u> and Dissenting Shares (as defined below)) issued and outstanding immediately prior to the Effective Time will be automatically converted into the right to receive \$1.80 in cash per share, without interest (the <u>Merger Consideration</u>). As of the Effective Time,

#### Table of Contents

all shares of Company Common Stock will no longer be outstanding and will automatically be cancelled and cease to exist, and each holder of a certificate representing any such shares of Company Common Stock will cease to have any rights with respect thereto, except the right to receive the Merger Consideration pursuant to this Section 4.1(c) upon the surrender of such certificate in accordance with Section 4.2, without interest (or in the case of Dissenting Shares, the rights contemplated by Section 4.6).

(d) <u>Adjustments to Prevent Dilution</u>. In the event that the Company changes the number of shares of Company Common Stock or securities convertible or exchangeable into or exercisable for shares of Company Common Stock issued and outstanding prior to the Effective Time as a result of a reclassification, stock split (including a reverse stock split), stock dividend or distribution, recapitalization, merger, subdivision, issuer tender or exchange offer, or other similar transaction, the Merger Consideration will be equitably adjusted; <u>provided</u>, <u>however</u>, that no such adjustment will be made for issuances of shares of Company Common Stock (or securities convertible or exchangeable into or exercisable for shares of Company Common Stock) that occur in the ordinary course of the Company s business pursuant to the conversion, exchange or exercise of any outstanding securities which are in existence as of the date of this Agreement or permitted by the terms hereof to be issued after the date hereof.

4.2 <u>Exchange of Certificates</u>. The procedures for exchanging outstanding shares of Company Common Stock for the Merger Consideration pursuant to the Merger are as follows:

(a) Exchange Agent. At or prior to the Effective Time, Parent shall deposit, or cause to be deposited, with an exchange agent appointed by Parent and reasonably approved by the Company prior to the date of this Agreement (the Exchange Agent ), for the benefit of the holders of shares of Company Common Stock, for payment through the Exchange Agent in accordance with this Section 4.2, cash in an amount equal to the product of the Merger Consideration and the number of shares of Company Common Stock issued and outstanding immediately prior to the Effective Time (exclusive of any shares to be cancelled pursuant to <u>Section 4.1(b)</u>) (the <u>Exchange Fund</u>). Pending distribution of the cash deposited with the Exchange Agent, such cash will be held in trust for the benefit of the holders of Company Common Stock entitled to receive the Merger Consideration and will not be used for any other purposes; provided, however, any interest and other income resulting from such investment shall become a part of the Exchange Fund, and any amounts in excess of the amounts payable under <u>Section 4.1(c)</u> will be promptly returned to Parent. The Exchange Agent shall invest the Exchange Fund as directed by Parent provided that (i) such investments will be in obligations of or guaranteed by the United States of America, in commercial paper obligations rated A-1 or P-1 or better by Moody s Investors Service, Inc. or Standard & Poor s Corporation, respectively, or in certificates of deposit, bank repurchase agreements or banker s acceptances of commercial banks with capital exceeding \$1 billion, and (ii) no such investments will have maturities that could prevent or delay payments to be made pursuant to this Article IV.

(b) Exchange Procedures. Promptly (and in any event within five (5) Business Days) after the Effective Time, Parent shall cause the Exchange Agent to mail to each holder of record of a certificate which immediately prior to the Effective Time represented outstanding shares of Company Common Stock (each, a <u>Certificate</u>) (i) a letter of transmittal in customary form and as reasonably approved by the Company and (ii) instructions for effecting the surrender of the Certificates in exchange for the Merger Consideration payable with respect thereto. Upon surrender of a Certificate (or effective affidavit of loss required by Section 4.2(g)) for cancellation to the Exchange Agent, together with a duly executed letter of transmittal, the holder of such Certificate will be entitled to receive in exchange therefor the Merger Consideration that such holder has the right to receive pursuant to the provisions of this <u>Article IV</u>, subject to any required withholding Taxes pursuant to <u>Section 4.2(f)</u>, and the surrendered Certificate will immediately be cancelled. No interest will be paid or accrued on the cash payable upon the surrender of such Certificates. In the event a transfer of ownership of Company Common Stock is not registered in the transfer records of the Company, it will be a condition of payment of the Merger Consideration that (A) the surrendered Certificate be properly endorsed, with signatures guaranteed, or otherwise in proper form for transfer, and (B) the Person requesting payment (I) pay any

transfer or other Taxes required by reason of the payment to a Person other than the registered holder of the surrendered Certificate, or (II) establish to the satisfaction of Parent that such Taxes have been paid or are not applicable. Until surrendered as contemplated by this <u>Section 4.2</u>, each Certificate (or effective affidavit of loss required by <u>Section 4.2(g)</u>) will be deemed at any time after the Effective Time to represent only the right to receive upon such surrender the Merger Consideration as contemplated by this

<u>Section 4.2</u>. The term <u>Person</u> means an individual, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization or other entity.

(c) <u>No Further Ownership Rights in Company Common Stock</u>. From and after the Effective Time, there will be no further registration of transfers on the stock transfer books of the Surviving Corporation of the shares of Company Common Stock which were outstanding immediately prior to the Effective Time and holders of Certificates will cease to have any rights as stockholders of the Surviving Corporation other than the right to receive the Merger Consideration upon surrender of such Certificates in accordance with <u>Section 4.2(b)</u> and <u>Section 4.2(g)</u> (or in the case of Dissenting Shares, the rights contemplated by <u>Section 4.6</u>) and any dividend or distribution with respect to shares of Company Common Stock evidenced by such Certificates with a record date prior to the Closing Date. If, after the Effective Time, Certificates are presented to the Surviving Corporation or the Exchange Agent for any reason, they will be cancelled and exchanged as provided in this <u>Article IV</u>.

(d) <u>Termination of Exchange Fund</u>. Any portion of the Exchange Fund which remains undistributed to the holders of Company Common Stock on the date that is 180 days after the Effective Time will be delivered to Parent, and any former holder of Company Common Stock who has not previously complied with this <u>Section 4.2</u> will be entitled to receive, upon demand, only from Parent payment of its claim for the Merger Consideration, without interest.

(e) <u>No Liability</u>. To the extent permitted by applicable Law, none of Parent, Merger Sub, the Company, the Surviving Corporation or the Exchange Agent will be liable to any holder of shares of Company Common Stock delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law.

(f) <u>Withholding Rights</u>. Each of the Exchange Agent, Parent and the Surviving Corporation will be entitled to deduct and withhold from the Merger Consideration otherwise payable pursuant to this Agreement to any holder of shares of Company Common Stock such amounts as it is required to deduct and withhold with respect to the making of such payment under the Internal Revenue Code of 1986, as amended (the <u>Code</u>), or any other applicable state, local or foreign Law related to Taxes. To the extent that amounts are so withheld by the Surviving Corporation or Parent, as the case may be, such withheld amounts (i) will be remitted by Parent or the Surviving Corporation, as the case may be, to the applicable Governmental Entity, and (ii) will be treated for all purposes of this Agreement as having been paid to the holder of the shares of Company Common Stock in respect of which such deduction and withholding was made by the Surviving Corporation or Parent, as the case may be, surviving Corporation or Parent, as the case may be surviving Corporation or Parent, as having been paid to the holder of the shares of Company Common Stock in respect of which such deduction and withholding was made by the Surviving Corporation or Parent, as the case may be.

(g) <u>Lost Certificates</u>. If any Certificate has been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed, and, if required by Parent, the posting by such Person of a bond in such reasonable amount as Parent may direct as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent shall pay, in exchange for such lost, stolen or destroyed Certificate, the Merger Consideration to be paid pursuant to this Agreement in respect of the shares of Company Common Stock formerly represented by such Certificate.

4.3 <u>Company Options</u>. Each Company Stock Option (as defined in <u>Section 5.2(a)</u>), to the extent outstanding and unexercised as of the Effective Time, shall be cancelled and shall thereafter no longer be exercisable except that the holder thereof (the <u>Option Holder</u>) shall be entitled to a payment in cash (the <u>Option Payment</u>), as of the Effective Time, in an amount (if any) equal to (i) the product of (x) the number of shares of Company Common Stock subject to such Company Stock Option held by such Option Holder, whether or not then vested or exercisable, and (y) the excess, if any, of the Merger Consideration over the exercise price per share of Company Common Stock subject to such Company Stock Option, <u>minus</u> (ii) all applicable federal, state and local Taxes required to be withheld by the Company. At or prior to the Effective Time, Parent shall deposit, or cause to be deposited, with the Company, for the benefit of the Option Holders, cash in an amount equal the aggregate amount of all Option Payments. Each Option Payment shall be paid by the Company or its agent to the applicable Option Holder as promptly as reasonably

practicable after the Closing Date. The Company Board (or an appropriate committee thereof) agrees to adopt resolutions to amend the Company Stock Plans to approve and effectuate the foregoing.

4.4 <u>Employee Stock Purchase Plans</u>. The Company shall take all actions with respect to the 1998 Employee Stock Purchase Plan, as amended (the <u>1998 ESP</u>P ) and the 1998 Employee Stock Purchase Plan

#### Table of Contents

for Foreign Employees (the <u>Foreign Employees ESP</u>P , and together with the 1998 ESPP, the <u>Company ESP Plans</u>) as are necessary to assure that (i) the Company ESP Plans shall continue to be suspended, (ii) there shall not be any additional Offering Period (as defined in the 1998 ESPP) or additional Purchase Period (as defined in the Foreign Employees ESPP), as the case may be, commencing following the date of this Agreement, and (iii) immediately prior to the Effective Time, the Company ESP Plans are terminated.

4.5 <u>Actions by the Company</u>. Except as contemplated by <u>Section 4.3</u>, the Company shall take all actions necessary to ensure that from and after the Effective Time the Surviving Corporation is not bound by any options, warrants, rights, awards, convertible debt securities, other convertible securities or similar arrangements to which the Company is a party which would entitle any Person (other than Parent) to beneficially own shares of the Surviving Corporation or Parent or receive any payments (other than as set forth in <u>Section 4.3</u>) in respect of such options, warrants, rights, awards, convertible debt securities, other convertible securities or similar arrangements.

#### 4.6 Dissenting Shares.

(a) Notwithstanding any other provisions of this Agreement to the contrary, any shares of Company Common Stock held by a holder who is entitled to demand and properly demands (and has not effectively withdrawn or lost such demand) appraisal rights under Section 262 of Delaware Law (collectively, the <u>Dissenting Shares</u>), will not be converted into or represent a right to receive the Merger Consideration, but the holder of Dissenting Shares will only be entitled to such rights as are provided by Delaware Law, including the right to receive payment of the fair value of such holder s Dissenting Shares in accordance with the provisions of Section 262 of Delaware Law.

(b) Notwithstanding the provisions of <u>Section 4.6(a)</u>, if any holder of Dissenting Shares effectively withdraws or loses (through failure to perfect or otherwise) such holder s appraisal rights under Delaware Law, then, as of the later of the Effective Time and the occurrence of such event, such holder s shares will automatically be converted into and represent only the right to receive the Merger Consideration, without interest thereon, upon compliance with the exchange procedures (including the surrender of the Certificate representing such shares) set forth in <u>Section 4.2</u>.

(c) The Company shall give Parent (i) prompt written notice of any written demand for appraisal received by the Company pursuant to the applicable provisions of Delaware Law, and (ii) the opportunity to participate in any negotiations and proceedings with respect to such demands. The Company shall not, except with the prior written consent of Parent, negotiate with any holder of Company Common Stock the terms of any payment, or make any payment, with respect to any such demands or offer to settle or settle any such demands, and the Company shall not communicate with any holder of Company Common Stock with respect to such demands, without prior consultation with Parent, except for communications directed to the Company stockholders generally or as required by Law.

### ARTICLE V

#### **REPRESENTATIONS AND WARRANTIES OF THE COMPANY**

The Company represents and warrants to Parent and Merger Sub that the statements contained in this <u>Article V</u> are true and correct, except as set forth in the disclosure schedule delivered by the Company to Parent and Merger Sub prior to the execution of this Agreement (the <u>Company Disclosure Schedule</u>). The Company Disclosure Schedule is arranged in sections and paragraphs corresponding to the numbered and lettered sections and paragraphs contained in this <u>Article V</u>, and the disclosure in any section or paragraph qualifies (a) the corresponding section or paragraph in this <u>Article V</u> and (b) the other sections and paragraphs in this <u>Article V</u> to the extent that it is reasonably apparent from a reading of such disclosure that it also qualifies or applies to such other sections and paragraphs.

5.1 Organization, Good Standing and Qualification; Subsidiaries.

### Table of Contents

(a) Each of the Company and its Subsidiaries is a corporation duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization, has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted, and, except as set forth on Section 5.1(a) of the Company Disclosure Schedule, is duly qualified to do

business and, where applicable as a legal concept, is in good standing as a foreign corporation in each jurisdiction where the ownership or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in good standing, or to have such power or authority, when taken together with all other such failures, has not had, and is not reasonably expected to have, a Company Material Adverse Effect (as defined below). The Company has made available to Parent a complete and correct copy of the certificate of incorporation and by-laws (or equivalent governing instruments) of the Company and each of its Subsidiaries and all amendments to such instruments. The certificate of incorporation and by-laws (or equivalent governing instruments) of the Company has made available to Parent correct and complete copies of the minutes of all meetings of the stockholders, the board of directors of the Company (the <u>Company Board</u>), the board of directors of each Subsidiaries held between January 1, 2002 and April 20, 2007. As used in this Agreement made available means that the subject documents were posted for secure external viewing on the Company s virtual data room in connection with negotiating this Agreement, or otherwise made available to Parent in writing.

(b) Section 5.1(b) of the Company Disclosure Schedule contains a complete and accurate list of (x) each of the Company s Subsidiaries and the ownership interest of the Company in each such Subsidiary, as well as the ownership interest of any other Person or Persons in each such Subsidiary and (y) each jurisdiction where the Company and each of its Subsidiaries is organized and qualified to do business.

(c) Section 5.1(c) of the Company Disclosure Schedule contains a complete and accurate list of any and all Persons, not constituting Subsidiaries of the Company, of which the Company directly or indirectly owns an equity or similar interest, or an interest convertible into or exchangeable or exercisable for an equity or similar interest (collectively, the <u>Investments</u>).

(d) The Company or a Subsidiary of the Company, as the case may be, owns all shares of capital stock or other securities owned by it in its Subsidiaries, and all Investments owned by it, free and clear of all liens, pledges, security interests, claims or other encumbrances (<u>Liens</u>), and there are no outstanding contractual obligations of the Company or any of its Subsidiaries permitting the repurchase, redemption or other acquisition of any of its interest in any Subsidiary or Investment or requiring the Company or any of its Subsidiaries to provide funds to, make any investment (in the form of a loan, capital contribution or otherwise) in, provide any guarantee with respect to, or assume, endorse or otherwise become responsible for the obligations of, any Subsidiary or Investment. The Company does not own, directly or indirectly, any voting interest in any Person that requires an additional filing by Parent under the Hart-Scott-Rodino Antitrust Improvement Act of 1976, as amended (the <u>HSR Act</u>).

The term (i) <u>Subsidiary</u> means, with respect to the Company, Parent or Merger Sub, as the case may be, any entity, whether incorporated or unincorporated, of which at least a majority of the securities or ownership interests having by their terms voting power to elect a majority of the board of directors or other persons performing similar functions is directly or indirectly owned or controlled by such party or by one or more of its respective Subsidiaries, and (ii) <u>Company Material Adverse Effect</u> means (X) any change or effect on the Company that is reasonably likely to prevent the Company from consummating the Merger and the other transactions contemplated by this Agreement, or (Y) any materially adverse change in, or materially adverse effect on, either individually or in the aggregate with all such other adverse changes in or effects on, the condition (financial or otherwise), results of operations, operations, business, assets (including intangible assets) or liabilities of the Company and its Subsidiaries taken as a whole, excluding, in each case:

(a) changes or effects that are primarily the result of general economic or business conditions, or conditions in financial or securities markets, in the United States;

(b) changes or effects that are primarily the result of factors generally affecting the industries or markets in which the Company operates;

(c) changes or effects that result from the effect of the public announcement or pendency of the transactions contemplated hereby on employees of the Company and its Subsidiaries or the public announcement by Parent of its plans with respect to the business of the Company or any of its Subsidiaries;

#### Table of Contents

(d) changes resulting from or arising out of actions taken pursuant (and/or required by) this Agreement or at the request of Parent, or the failure to take any actions due to restrictions set forth in this Agreement; <u>provided</u>, <u>however</u>, to the extent that the Company reasonably believes that taking any action required by this Agreement or at the request of Parent, or failing to take any action prohibited by this Agreement, could reasonably be expected to result in a Company Material Adverse Effect, only if the Company provides timely prior notification to Parent of such belief, and Parent does not provide timely relief from the provisions of this Agreement or its request, shall the changes or effects resulting from this subsection (d) be deemed to not constitute a Company Material Adverse Effect;

(e) any changes in the price or trading volume of the Company s stock on the Pink Sheets or other over the counter market; <u>provided</u>, <u>however</u>, that the exception in this clause shall not prevent or otherwise affect a determination that any change, effect, circumstance or development underlying such changes has or has not resulted in, or contributed to, a Company Material Adverse Effect, and no such changes shall be used as evidence that some other change, effect, circumstance or development has had or has not had a Company Material Adverse Effect; or

(f) any adverse changes arising from or relating to any change in GAAP or any change in applicable Laws, in each case, proposed, adopted or enacted after the date hereof, or the interpretation or enforcement thereof;

<u>provided</u>, <u>further</u>, that the Company successfully bears the burden of proving that any such change or effect in clause (a), (b) or (f) immediately above does not (i) primarily relate only to (or have the effect of primarily relating only to) the Company and its Subsidiaries, or (ii) disproportionately adversely affect the Company and its Subsidiaries compared to other companies of similar size operating in the industry in which the Company and its Subsidiaries operate. In addition, with respect to clause (a) or (b) above, the Company shall bear the burden of proving that any such change or effect primarily results from the factors identified in such clause.

#### 5.2 Capital Structure.

(a) As of the date of this Agreement, the authorized capital stock of the Company consists of 200,000,000 shares of Company Common Stock and 5,000,000 shares of preferred stock, par value \$0.001 per share (the <u>Preferred Shares</u>). All of the outstanding shares of Company Common Stock have been duly authorized and are validly issued, fully paid and nonassessable. At the close of business on April 20, 2007, 77,637,177 shares of Company Common Stock and no Preferred Shares were issued and outstanding. At the close of business on April 20, 2007, the Company had no shares of Company Common Stock or Preferred Shares reserved for issuance and no other form of equity award had been granted, except that: (i) 12,109,924 shares of Company Common Stock were reserved for issuance by the Company pursuant to outstanding options (a <u>Company Stock Option</u>) under the Company s 1995 Stock Option Plan, as amended (the <u>1995 Plan</u>), 1997 Equity Incentive Plan, as amended (the <u>1997 Plan</u>), 1998 Non-Employee Director s Stock Option Plan (the 1998 Plan ), 1999 Non-Officer Equity Incentive Plan (the 1999 Plan and, together with the 1995 Plan, the 1997 Plan and the 1998 Plan, the <u>Company Option Plans</u>) or granted outside of the Company Option Plans; (ii) 1,812,024 shares of Company Common Stock were reserved for issuance and available for future grants under the Company Option Plans; (iii) 600,371 shares of Company Common Stock were reserved for issuance for future purchase rights under the Company ESP Plans (together with the Company Option Plans, the <u>Company Stock Plans</u>); (iv) 2,000,000 Preferred Shares were reserved for issuance in connection with the Company Rights Agreement (as defined below); and (v) 156,667 shares of Company Common Stock were held by the Company in its treasury. As of the date hereof, the Company has granted pursuant to the 1995 Plan and the 1999 Plan, 692 shares of Company Common Stock in the form of restricted stock, all of which are vested as of the date hereof and included in the number of issued and outstanding shares of Company Common Stock set forth above. Section 5.2(a) of the Company Disclosure Schedule sets forth a true and complete list, as of the close of business on April 20, 2007, of: (i) all Company Stock Plans, indicating for each Company Stock Plan, as of such date, the number of shares of Company Common Stock subject to outstanding Company Stock Options and restricted stock awards under such Company

Stock Plan and the number of shares of Company Common Stock reserved for future issuance under such Company Stock Plan; and (ii) all outstanding Company Stock Options, indicating with respect to each such Company Stock Option the name of the holder of such option, the Company Stock Plan under which it was granted (or if it was granted outside of the Company Option Plans), the number of shares of Company Common Stock subject to such Company Stock Option, the exercise price, the date of grant, and the vesting schedule, including whether (and to what extent) the vesting accelerates in any way

#### Table of Contents

by the execution of this Agreement, the consummation of the Merger or termination of employment or change in position following consummation of the Merger. No form of equity award, including, without limitation, shares of restricted stock or other similar rights is outstanding under the Company Option Plans, except for the Company Stock Options and restricted stock awards set forth on Section 5.2(a) of the Company Disclosure Schedule. There are no outstanding stock purchase rights under the Company ESP Plans. The Company has made available to Parent complete and accurate copies of all Company Stock Plans, and the forms of all stock option agreements and notices of grants or awards evidencing Company Stock Options, and forms of all purchase or participation elections under the Company ESP Plans. As of the date hereof, the Company Common Stock is quoted on the Pink Sheets (the\_Pink Sheets), published by Pink Sheets, LLC.

(b) Each of the outstanding shares of capital stock or other securities of each of the Company s Subsidiaries is duly authorized, validly issued, fully paid and nonassessable.

(c) Except as set forth above in this <u>Section 5.2</u> or in Section 5.2(c) of the Company Disclosure Schedule, and except for the rights (the <u>Rights</u>) issuable pursuant to the Rights Agreement, dated as of February 6, 2001 (the <u>Company</u> <u>Rights Agreement</u>), between the Company and Fleet National Bank, as rights agent, in respect of which no Distribution Date (as defined in the Company Rights Agreement) has occurred, there are no preemptive or other outstanding rights, options, warrants, conversion rights, phantom stock units, restricted stock units, or stock appreciation rights or similar rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or rights of any kind that obligate the Company or any of its Subsidiaries to issue or sell any shares of capital stock or other securities of the Company or any of its Subsidiaries or acquire, any securities of the Company or any of its Subsidiaries are authorized, issued or outstanding. The Company does not have outstanding any bonds, debentures, notes or other obligations (i) the terms of which provide the holders the right to vote with the stockholders of the Company on any matter or (ii) that are convertible into or exercisable for securities having the right to vote with the stockholders of the Company on any matter (any such bonds, debentures, notes or obligations, <u>Voting Debt</u>).

(d) There are no registration rights to which the Company or any of its Subsidiaries is a party or by which it or they are bound with respect to any equity security of any class of the Company. Neither the Company nor any of its Affiliates (as defined below) is a party to or is bound by any agreements or understandings with respect to the voting (including voting trusts and proxies) or sale or transfer (including agreements imposing transfer restrictions) of any shares of capital stock or other equity interests of the Company, except for transfer restrictions under the terms of the Company Stock Options. Except as set forth on Section 5.2(d) of the Company Disclosure Schedule, there are no obligations, contingent or otherwise, of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of Company Common Stock or the capital stock of the Company or any of its Subsidiaries. As used in this Agreement with respect to any party, the term <u>Affiliate</u> means any Person who is an affiliate of that party within the meaning of Rule 405 promulgated under the Securities Act.

(e) Except as set forth on Section 5.2(e) of the Company Disclosure Schedule, (i) all Company Stock Options awarded under the Company Option Plans were duly and lawfully granted and approved in accordance with the requirements of the applicable corporate, Tax Laws, the Securities Act of 1933, as amended (the <u>Securities Act</u>), U.S. state securities Laws, any non-U.S. securities Laws and the terms of the applicable Company Stock Plan; (ii) the Company s minutes, grantee documentation and other equity plan administration records each reflect the proper measurement date of each such Company Stock Option pursuant to the applicable requirements of United States generally accepted accounting principles (<u>GAAP</u>) in effect at the time of each grant; and (iii) all of the Company s financial statements filed with the United States Securities and Exchange Commission (the <u>SEC</u>) have accounted for and reflected in accordance with GAAP in all material respects all awards, modifications, exchanges, or other transactions in connection with the Company Stock Plans. The fair market value of each Company Stock Option on the date of grant

was established in accordance with a valuation methodology set forth under the terms of the applicable Company Stock Plan and meets the requirements of Sections 409A, 422 and 423 of the Code, as, and to the extent, applicable. All purchase rights previously granted under the Company ESP Plans were granted in accordance with all of the requirements of Section 423(b) of the Code. Except as set forth on Section 5.2(e) of the Company Disclosure Schedule, each Company Stock Option was granted with an exercise price per share that was not less than the fair market value per share of the Company Common Stock on the date of

grant. The Company has complied in all material respects with all required income and payroll tax withholding and reporting requirements with respect to the Company Stock Plans and all grants, exercises, issuances and other transactions thereunder.

# 5.3 Corporate Authority; Approval and Fairness.

(a) The Company has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under this Agreement, and to consummate the Merger, subject only to adoption of this Agreement and approval of the Merger by the holders of a majority of the outstanding shares of Company Common Stock entitled to vote thereon (the <u>Shareholder Approval</u>), and the filing of the Certificate of Merger pursuant to Delaware Law. This Agreement is a valid and binding agreement of the Company, enforceable against the Company in accordance with its terms, except as the enforcement may be limited by bankruptcy, insolvency (including all Laws relating to fraudulent transfers), reorganization, moratorium or similar Laws affecting enforcement of creditors rights generally now or hereafter in effect and except as enforcement is subject to general principles of equity (regardless of whether enforcement is considered in a proceeding in equity or at Law).

(b) On or prior to the date hereof, the Company Board, acting unanimously, has (i) determined that this Agreement and the Merger are fair to, and in the best interests of, the Company and the holders of Company Common Stock, (ii) approved this Agreement and declared its advisability in accordance with the provisions of Delaware Law, (iii) resolved to recommend this Agreement and the Merger to the holders of Company Common Stock for adoption and approval in accordance with <u>Section 7.4</u> of this Agreement (the <u>Company Board Recommendation</u>), and (iv) directed that this Agreement and the Merger be submitted to the holders of Company Common Stock for consideration in accordance with this Agreement. The Company Board has received the opinion of its financial advisor, Goldman, Sachs & Co., to the effect that (subject to the assumptions and qualifications set forth in such opinion), as of the date of such opinion, the \$1.80 in cash per share to be received by the holders of the shares of Company Common Stock pursuant to the Agreement is fair from a financial point of view to such holders.

### 5.4 Governmental Filings; No Violations; Certain Contracts, Etc.

(a) Other than (i) the filings, approvals and/or notices pursuant to Section 1.2, (ii) the pre-merger notification requirements under the HSR Act (or similar foreign filings, if applicable), (iii) applicable requirements, if any, of the Securities Act, and the rules and regulations promulgated thereunder and the Securities Exchange Act of 1934, as amended (the <u>Exchange Act</u>), and the rules and regulations promulgated thereunder, including the requirement to file the Proxy Statement with the SEC, (iv) such consents, approvals, orders, authorizations, registrations, declarations and filings as may be required under applicable U.S. state securities Laws, and (v) the notifications, consents and approvals set forth in Section 5.4(a) of the Company Disclosure Schedule (all of such filings, approvals, notices, consents, orders, authorizations, registrations, declarations and notifications described in clauses (i) through (v) above, collectively, the <u>Company Approvals</u>), no notices, reports or other filings are required to be made by the Company with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the Company from, any foreign or domestic governmental or regulatory authority (including self-regulatory authorities), agency, commission, body or other governmental entity, or any quasi-governmental or private body exercising any regulatory, taxing, importing or other governmental or quasi-governmental authority (<u>Governmental Entity</u>), in connection with the execution and delivery of this Agreement by the Company and the consummation by the Company of the Merger and the other transactions contemplated hereby, except those that the failure to make or obtain are not, individually or in the aggregate, reasonably expected to result in a material liability to the Company and its Subsidiaries, taken as a whole.

(b) The execution, delivery and performance of this Agreement by the Company do not, and the consummation by the Company of the Merger and the other transactions contemplated hereby will not, constitute or result in (i) a breach or

violation of, or a default under, the certificate of incorporation or by-laws of the Company or the equivalent governing instruments of any of its Subsidiaries, (ii) a breach or violation of, a termination (or right of termination) or a default under, or the acceleration of any obligations or the creation of a Lien on the assets of the Company or any of its Subsidiaries (with or without notice, lapse of time or both) pursuant to, any agreement, lease, license, contract, note, mortgage, indenture, arrangement or other obligation, whether oral or written (<u>Contracts</u>) binding upon the Company or any of its Subsidiaries, or any Laws or governmental permit or

#### Table of Contents

license to which the Company or any of its Subsidiaries is subject, or (iii) any change in the rights or obligations of any party under any of the Contracts, except, in the case of clause (ii) or (iii) above, for any conflict, breach, violation, termination, default, acceleration, creation or change that has not had, and is not reasonably expected to have, a Company Material Adverse Effect. Section 5.4(b) of the Company Disclosure Schedule sets forth a complete and accurate list of all notices, consents or waivers that are expressly required under the provisions of the Contracts referred to in <u>Section 5.5(a)</u> or Contracts for Third Party Embedded Software or Third Party IP Licenses (other than software subject to open source or similar type license agreements) as a result of the Merger or other transactions contemplated by this Agreement or that are necessary to avoid the other party to any such Contract or Third Party IP License having a right to terminate or claim a breach of any such agreement as a result of the Merger or other transactions contemplated by this Agreement.

### 5.5 Contracts.

(a) The term <u>Company Material Contract</u> means any of the following:

(i) any material contract (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC) with respect to the Company and its Subsidiaries;

(ii) any employment, service or consulting Contract or arrangement with any current or former executive officer of the Company or member of the Company Board, and any employment, service or consulting Contract or arrangement with any other employee of the Company or its Subsidiaries that provides for at least \$100,000 in base compensation, other than those that are terminable by the Company or any of its Subsidiaries on no more than thirty (30) days notice without liability or financial obligation to the Company or any of its Subsidiaries;

(iii) any Contract between the Company or any of its Subsidiaries and any current customer of the Company and its Subsidiaries (A) with respect to which the Company and its Subsidiaries recognized cumulative revenue during the twelve-month period ended December 31, 2006 in excess of one percent (1%) of the Company s consolidated revenue during that period (each such customer, a <u>Major Customer</u>, and each Contract referenced in this Section 5.5(a)(iii)(A), a <u>Major Customer Contract</u>), or (B) that contains any covenant of the Company granting any exclusivity rights or contains most favored customer pricing provisions;

(iv) any Contract between the Company or any of its Subsidiaries and any current customer of the Company and its Subsidiaries that contains any (A) penalties imposed on the Company or any of its Subsidiaries for late delivery of the Company s or any of its Subsidiaries products or breach of other performance obligations by the Company or any of its Subsidiaries, or (B) penalties (other than standard warranty obligations agreed to by the Company in the ordinary course of business) imposed on the Company or any of its Subsidiaries associated with repairs, returns or quality performance of the Company s or any of its Subsidiaries products or services;

(v) any Contract between the Company or any of its Subsidiaries and any supplier of goods, products or components (including software) and/or services with respect to which the Company and its Subsidiaries made cumulative expenditures during the twelve-month period ended December 31, 2006 greater than \$50,000 (each such supplier, a <u>Major Supplier</u>);

(vi) (A) any Contract between the Company or any of its Subsidiaries and any sole source suppliers, or (B) original equipment manufacturer (OEM) Contracts, electronic manufacturing services (EMS) Contracts, original design and manufacturing supply (ODM) Contracts, third party logistics (3PL) Contracts, transportation Contracts, and other contract manufacturing Contracts, or any other Contract that licenses or otherwise authorizes any Person to design, manufacture, reproduce, develop or modify the products, services or technology of the Company and its Subsidiaries (other than agreements allowing internal backup copies to be made by end-user customers in the ordinary course of

business);

(vii) Contracts (A) that contain any take or pay or volume commitment provisions binding the Company or any of its Subsidiaries, or (B) that contain provisions granting any rights of first refusal,

rights of first negotiation or similar rights to any Person other than the Company in a manner which is material to the business of the Company and its Subsidiaries, taken as a whole;

(viii) (A) any Contract containing any covenant limiting in any respect the right of the Company or any of its Subsidiaries to engage in any line of business, or to compete with any Person in any line of business or in the geographic locations in which any such Person may engage in business, or (B) any Contract otherwise prohibiting or limiting the right of the Company or any of its Subsidiaries to (x) make, sell or distribute any products or services or (y) use, transfer, license, distribute or enforce any Intellectual Property rights owned by the Company or any of its Subsidiaries immediately prior to the execution of such Contract;

(ix) any Contract relating to the disposition or acquisition by the Company or any of its Subsidiaries after the date of this Agreement of a material amount of assets not in the ordinary course of business or pursuant to which the Company or any of its Subsidiaries has any material ownership interest in any other Person other than the Company s Subsidiaries (including joint venture, partnership or other similar agreements);

(x) any Contract which provides access to source code to any Person for all or any portion of any product of the Company or Owned Intellectual Property in any circumstance;

(xi) any Contract or other arrangement constituting a direct financial obligation or off-balance sheet arrangement as defined under Item 2.03(c) and (d) in SEC Form 8-K (without regard to its materiality) and any other mortgages, indentures, guarantees, loans or credit agreements, security agreements or other Contracts relating to Indebtedness of the Company or any of its Subsidiaries or extension of credit, other than accounts receivables and payables in the ordinary course of business;

(xii) any settlement agreement entered into by the Company or, to the extent possessed by or available to the Company, by any current or former executive officer within five (5) years prior to the date of this Agreement, other than (A) releases immaterial in nature or amount entered into with former employees or independent contractors of the Company in the ordinary course of business in connection with the routine cessation of such employee s employment or independent contractor s service arrangement with the Company, or (B) settlement agreements with Persons other than Governmental Entities for cash only (which has been paid) that do not exceed \$50,000 as to such settlement;

(xiii) any Contract not described in clause (vi) above under which the Company or any of its Subsidiaries has (A) granted a license or other right to or under any Owned Intellectual Property or a sublicense or other right to or under any Intellectual Property licensed under a Third Party License, in each case to any Person, other than to customers, distributors and other resellers in the ordinary course of business, or (B) assigned any Intellectual Property previously owned by the Company or any of its Subsidiaries and material to the operation of their respective businesses, as applicable, to any Third Party within four (4) years prior to the date of this Agreement;

(xiv) any Contract not otherwise described in this <u>Section 5.5(a)</u> which has aggregate future sums due from the Company or any of its Subsidiaries in excess of 100,000 and is not terminable by the Company or any such Subsidiary (without penalty or payment) on ninety (90) or fewer days notice;

(xv) any Contract (A) with any Affiliate of the Company (other than its Subsidiaries), (B) with investment bankers, financial advisors, attorneys, accountants or other advisors retained by the Company or any of its Subsidiaries involving payments by or to the Company or any of its Subsidiaries of more than \$50,000 on an annual basis during the three (3) years prior to the date of this Agreement, or any such Contract pursuant to which the Company has ongoing obligations, (C) providing for indemnification by the Company or any of its Subsidiaries of any Person, except for any such Contract that is (x) not material to the Company or any of its Subsidiaries and (y) entered into in the ordinary course of business, (D) containing a standstill or similar agreement pursuant to which the Company or

any of its Subsidiaries have agreed not to acquire assets or securities of another Person, or (E) relating to currency hedging or similar transactions.

### Table of Contents

(b) Section 5.5(b) of the Company Disclosure Schedule sets forth a list (arranged in clauses corresponding to the clauses set forth in <u>Section 5.5(a)</u>) of all Company Material Contracts to which the Company or any of its Subsidiaries is a party or bound by as of the date of this Agreement. A complete and accurate copy of each Company Material Contract has been made available to Parent (including all amendments, modifications, extensions, renewals, guarantees or other Contracts with respect thereto).

(c) All Company Material Contracts are valid and binding and in full force and effect, except to the extent they have previously expired in accordance with their terms. Except as set forth on Section 5.5(c) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has violated in any material respect, and, to the knowledge of the Company, no other party to any of the Company Material Contracts has violated in any material respect, any provision of, or committed or failed to perform any act which, with or without notice, lapse of time or both, would constitute a material default under the provisions of such Company Material Contract. Neither the Company nor any of its Subsidiaries has, and, to the knowledge of the Company, no other party to such Contracts has repudiated by oral or written notice to the Company any material provision of any Company Material Contract.

(d) During the last twelve (12) months, none of the Major Customers has terminated or failed to renew or informed the Company of any intention to materially reduce purchases under any of its Major Customer Contracts and neither the Company nor any of its Subsidiaries has received any written notice of termination or such reduced purchases from any of the Major Customers.

(e) Section 5.5(e) of the Company Disclosure Schedule sets forth each Major Supplier and the cumulative expenditures made by the Company and its Subsidiaries during the twelve-month period ended December 31, 2006.

(f) The Company has made available to Parent a copy of each of the standard form Contracts currently in use by the Company or any of its Subsidiaries (including end user, maintenance and reseller standard form Contracts) in connection with their respective businesses.

(g) Section 5.5(g) of the Company Disclosure Schedule sets forth a complete and accurate list of all active vendors, resellers and distributors or similar Persons (including agents) through which the products of the Company and its Subsidiaries were marketed, sold or otherwise distributed during the twelve (12) months preceding the date of this Agreement. Each reseller and distributor agreement of the Company and its Subsidiaries is terminable by the Company or its Subsidiary (without penalty or cost) upon ninety (90) days or less notice.

### 5.6 SEC Filings; Financial Statements; Information Provided.

(a) The Company has filed all registration statements, forms, reports and other documents required to be filed by the Company with the SEC since January 1, 2002. All such registration statements, forms, reports and other documents (including those that the Company files up to the Closing), together with all certifications required pursuant to the Sarbanes-Oxley Act of 2002 and the related rules and regulations promulgated under or pursuant to such act (the <u>Sarbanes-Oxley Act</u>), are referred to as the <u>Company SEC Rep</u>orts. Except to the extent that information contained in any Company SEC Report filed and publicly available prior to the date of this Agreement has been specifically revised or superseded by a later filed Company SEC Report filed prior to the date of this Agreement, the Company SEC Reports (i) were or will be filed on a timely basis (other than the Company s Annual Reports on Form 10-K for the fiscal years ending December 31, 2005 and December 31, 2006, and the Company s Quarterly Reports on Form 10-Q for the fiscal quarters ending September 30, 2005 and March 31, June 30 and September 30, 2006), (ii) at the time filed, complied, or will comply when filed, as to form in all material respects with the applicable requirements of the Securities Act and the Exchange Act, as the case may be, the Sarbanes-Oxley Act and the rules and regulations of the SEC thereunder applicable to such Company SEC Reports, and (iii) did not or will not at the

time they were or are filed contain any untrue statement of a material fact or omit to state a material fact required to be stated in such Company SEC Reports or necessary in order to make the statements in such Company SEC Reports, in the light of the circumstances under which they were made, not misleading. No Subsidiary of the Company is subject to the reporting requirements of Section 13(a) or Section 15(d) of the Exchange Act. The Company has made available to Parent true, correct and complete copies of all correspondence between the SEC, on the one hand, and the Company and any of its Subsidiaries, on the other, since January 1, 2002, including (i) all SEC comment letters and responses to such comment letters by or on behalf of the

A-12

Company, and (ii) any letters, complaints, or other documents from the SEC or any staff or office of the SEC informing the Company of any inquiry, claim or proceeding (formal, informal or otherwise) or request for documents or information, and all written responses thereto by or on behalf of the Company. To the knowledge of the Company, none of the Company SEC Reports is the subject of ongoing SEC review or outstanding SEC comment. There are no off-balance sheet arrangements as defined in Item 2.03(d) of SEC Form 8-K with respect to the Company or any of its Subsidiaries that would be required to be reported or set forth in the Company SEC Reports or any such reports required to be filed in the future.

(b) Except to the extent that information contained in any Company SEC Report filed and publicly available prior to the date of this Agreement has been specifically revised or superseded by a later filed Company SEC Report filed prior to the date of this Agreement, each of the consolidated financial statements (including, in each case, any related notes and schedules) contained or to be contained in or incorporated by reference in the Company SEC Reports at the time filed or to be filed (i) complied or will comply as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto and (ii) were or will be prepared in accordance with GAAP applied on a consistent basis throughout the periods involved (except as may be indicated in the notes to such financial statements or, in the case of unaudited interim financial statements, as permitted by the SEC with respect to Form 10-Q under the Exchange Act). Except to the extent that information contained in any Company SEC Report filed and publicly available prior to the date of this Agreement has been specifically revised or superseded by a later filed Company SEC Report filed prior to the date of this Agreement, each of the consolidated balance sheets (including, in each case, any related notes and schedules) contained or incorporated by reference in the Company SEC Reports at the time filed fairly presented in all material respects the consolidated financial position of the Company and its Subsidiaries as of the dates indicated and each of the consolidated statements of income and of changes in financial position contained or to be contained or incorporated by reference in the Company SEC Reports (including, in each case, any related notes and schedules) fairly presented in all material respects the consolidated results of operations, retained earnings and changes in financial position, as the case may be, of the Company and its Subsidiaries for the periods set forth therein, except that the unaudited interim financial statements were subject to normal and recurring year-end adjustments.

(c) Except as and to the extent set forth on the consolidated balance sheet of the Company and its consolidated Subsidiaries as at December 31, 2006 (including the notes thereto and related management discussion and analysis) included in the Company s Annual Report on Form 10-K for the fiscal year ended December 31, 2006, neither the Company nor any Subsidiary has any liability or obligation of any nature (whether accrued, absolute, contingent or otherwise, and whether or not required to be disclosed), except for liabilities and obligations (i) incurred in connection with the transactions contemplated hereby, (ii) incurred in the ordinary course of business and in a manner consistent with past practice since December 31, 2006, or (iii) except as set forth on Section 5.6(c) of the Company Disclosure Schedule, that have not had, and are not reasonably expected to have, a material adverse impact on the Company and its Subsidiaries, taken as a whole.

(d) The information to be supplied by or on behalf of the Company for inclusion in the proxy statement to be sent to the stockholders of the Company (the <u>Proxy Statement</u>) in connection with the Company Meeting will not, on the date it is first mailed to the stockholders of the Company or at the time of the Company Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated in the Proxy Statement or necessary in order to make the statements in the Proxy Statement, in light of the circumstances under which they are made, not misleading. The Proxy Statement will comply as to form in all material respects with the provisions of the Exchange Act and the rules and regulations promulgated by the SEC thereunder. The representations and warranties contained in this <u>Section 5.6(d)</u> will not apply to statements or omissions included in the Proxy Statement or any other filings made with the SEC based upon information furnished in writing to the Company by Parent or Merger Sub specifically for use in the Proxy Statement.

(e) The Company maintains disclosure controls and procedures and internal control over financial reporting as required under Rule 13a-15(a) promulgated under the Exchange Act. Except as set forth on Section 5.6(e)-1 of the Company Disclosure Schedule, such disclosure controls and procedures and such internal control over financial reporting were effective as of December 31, 2006, and the same are otherwise reasonably designed to comply with the respective definitions of such controls in Rules 13a-15(e) and (f) promulgated under the Exchange Act. The Company has disclosed, based on its most recent evaluation prior to the date of this

A-13

Agreement, to the Company s auditors and the audit committee of the Company Board (i) any significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting which are reasonably likely to adversely affect in any material respect the Company s ability to record, process, summarize and report financial information, and (ii) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting. The Company has made available to Parent a summary of any such disclosure made by management to the Company s auditors and audit committee since January 1, 2004, and Section 5.6(e)-1 of the Company Disclosure Schedule sets forth a summary of all current significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting. Except as set forth on Section 5.6(e)-2 of the Company Disclosure Schedule, since January 1, 2004, no current or former employee of the Company or any of its Subsidiaries has alleged to any of the senior officers of the Company or such Subsidiary that the Company or any such Subsidiaries has engaged in questionable or fraudulent accounting or auditing practices. No attorney representing the Company or any of its Subsidiaries, whether or not employed by the Company or any of its Subsidiaries, has reported evidence of a violation of securities Laws, breach of fiduciary duty or similar violation by the Company or any of its officers, directors, employees or agents to the Company Board or any of its committees or to any director, in his or her capacity as a director, or officer, in his or her capacity as an officer, of the Company or any of its Subsidiaries.

(f) The Company and, to the knowledge of the Company, each of its officers and directors are in compliance with, and have complied, in each case in all material respects with (i) since the enactment of the Sarbanes-Oxley Act, the applicable provisions of the Sarbanes-Oxley Act at the time that such provisions became effective, and (ii) since the date that the Company Common Stock has been quoted on the Pink Sheets, the SEC and National Association of Securities Dealers (the <u>NASD</u>) rules applicable to companies quoted on the Pink Sheets (and since any such date, the Company has not given or been required to give notice to, and has not received notice from, the SEC, the NASD or any other Person, (x) to the effect that the Company is or may be in violation of any of the SEC or NASD rules applicable to companies quoted on the Pink Sheets or regulations that would affect the eligibility of the Company Common Stock from being quoted on the Pink Sheets).

(g) There are no outstanding loans made by the Company or any of its Affiliates to any executive officer (as defined in Rule 3b-7 under the Exchange Act) or director of the Company or any Subsidiary of the Company. Except as permitted by the Exchange Act, including Sections 13(k)(2) and (3), since the enactment of the Sarbanes-Oxley Act, neither the Company nor any of its Affiliates has made, arranged or modified (in any material way) personal loans or an extension of credit to any executive officer (as defined in Rule 3b-7 under the Exchange Act) or director of the Company or any Subsidiary of the Company.

5.7 <u>Absence of Certain Changes</u>. Since December 31, 2006, the Company and its Subsidiaries have conducted their respective businesses only in, and have not engaged in any material transaction other than according to, the ordinary and usual course of such businesses and, since such date, except as set forth on Section 5.7 of the Company Disclosure Schedule, there has not been (a) any change in the financial condition, properties, business or results of operations of the Company and its Subsidiaries or any development, circumstance or occurrence or combination thereof which has had, or could reasonably be expected to have, a Company Material Adverse Effect (including any adverse change with respect to any development, circumstance or occurrence existing on or prior to such date), (b) any material damage, destruction or other casualty loss with respect to any material asset or property owned, leased or otherwise used by the Company or any of its Subsidiaries, whether or not covered by insurance, or (c) any other action or event that would have required the consent of Parent under <u>Section 7.1</u> had such action or event occurred after the date of this Agreement.

## 5.8 Litigation.

(a) Except as set forth on Section 5.8(a)-1 of the Company Disclosure Schedule, there are no (i) civil, criminal or administrative actions, suits, claims, hearings, investigations or proceedings (collectively, <u>Actions</u>) pending or, to the knowledge of the Company, threatened against the Company or any of its Subsidiaries, (ii) judgments, orders or decrees outstanding against the Company or any of its Subsidiaries, or (iii) other facts or circumstances which, to the knowledge of the Company, are reasonably expected to result in any Action against the Company or any of its Subsidiaries. Other than as set forth in Section 5.8(a)-2 of the Company Disclosure Schedule,

A-14

### Table of Contents

there has not been since January 1, 2004, nor are there currently, any internal investigations, or inquiries reasonably expected to lead to a material internal investigation, being conducted by the Company Board (or any of its committees) or any Person at the request of the Company Board concerning any financial, accounting, Tax, conflict of interest, illegal activity, fraudulent or deceptive conduct or other misfeasance or malfeasance issues.

(b) The indemnification obligations of the Company (including advancement of expenses) with respect to any present or former directors, officers or employees of the Company and its Subsidiaries arising out of any past, pending or threatened proceedings or other events that have given rise to or may give rise to any indemnification obligations of the Company pursuant to any agreement, the certificate of incorporation or bylaws, as amended, of the Company, or any statute, are specified in Section 5.8(b) of the Company Disclosure Schedule.

(c) Section 5.8(c)-1 of the Company Disclosure Schedule sets forth a list of (i) all Contracts of the Company and its Subsidiaries relating to Indebtedness, currently outstanding or that could become outstanding in the future, and (ii) the amount of such Indebtedness, including any accrued interest, as of the date of this Agreement. Section 5.8(c)-2 of the Company Disclosure Schedule sets forth, calculated as of the date of this Agreement, the aggregate amount of cash and cash equivalents of the Company and its Subsidiaries less the aggregate amount of Indebtedness of the Company and its Subsidiaries (including, for the avoidance of doubt and without limitation, any penalties, premiums, liquidated damages or similar amounts relating to any Indebtedness that may become due and payable as a result of the execution of this Agreement or the consummation of the Merger or the transactions contemplated by this Agreement) (such difference, the <u>Company Net Cash</u>).

The term <u>Indebtedness</u> means, with respect to any Person, (A) all indebtedness of such Person, whether or not contingent, for borrowed money, (B) all obligations of such Person evidenced by notes, bonds, debentures or other similar instruments, and (C) all Indebtedness of others referred to in clauses (A) and (B) guaranteed, directly or indirectly, in any manner by such Person, or in effect guaranteed directly or indirectly by such Person through a Contract (I) to pay or purchase such Indebtedness or to advance or supply funds for the payment or purchase of such Indebtedness, (II) to purchase, sell or lease (as lessee or lessor) property, or to purchase or sell services, primarily for the purpose of enabling the debtor to make payment of such Indebtedness or to assure the holder of such Indebtedness against loss, (III) to supply funds to or in any other manner invest in the debtor (including any agreement to pay for property or services irrespective of whether such property is received or such services are rendered), or (IV) otherwise to assure a creditor against loss; provided, however, solely with respect to <u>Section 5.8(c)</u> above and <u>Section 8.2(k)</u>, Indebtedness shall not be deemed to include intercompany amounts, capital lease obligations or any expenses or costs incurred by the Company pursuant to <u>Section 7.15</u>.

### 5.9 Employee Benefits.

(a) Section 5.9(a)-1 and Section 5.13(b) of the Company Disclosure Schedule lists all benefit and compensation plans, policies or arrangements, other than commission arrangements, currently maintained or contributed to by the Company, any of its Subsidiaries or any other entity, which together with the Company or any of its Subsidiaries, is treated as a single employer under Section 414 of the Code (an <u>ERISA Affiliate</u>) (or in respect of which the Company, any of its Subsidiaries or any ERISA Affiliate has any outstanding liability) and covering current or former employees, independent contractors, consultants (including outsourcing), temporary employees and current or former directors of the Company, any of its Subsidiaries or any ERISA Affiliate or any ERISA Affiliate, which are employee benefit plans within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended (<u>ERISA</u>), and any other written plan, policy or arrangement (whether or not subject to ERISA) involving direct or indirect compensation, other than commission arrangements, currently maintained by the Company, any of its Subsidiaries or any ERISA Affiliate has any outstanding liability) and covering current or former employees, independent contractors, consultants (including outsourcing), temporary employees and current or former directors of the Company, any of its Subsidiaries or any ERISA Affiliate (or in respect of which the Company, any of its Subsidiaries or any ERISA Affiliate has any outstanding liability) and covering current or former employees, independent contractors, consultants (including outsourcing), temporary employees and current or former employees, independent contractors, consultants (including outsourcing), temporary employees and current or former directors of the Company, any of its Subsidiaries or any ERISA Affiliate has any outstanding liability) and covering current or former employees, independent contractors, consultants (including outsourcing), temporary employees and current or f

including health, dental, vision or life insurance coverage, vacation, loans, fringe benefits, severance benefits, change in control plan or agreements, disability benefits, deferred compensation, bonuses, stock options, stock ownership or purchase, phantom stock, stock appreciation, stock based or other forms of incentive compensation, bonus or post-retirement compensation or benefits (the <u>Company Benefit Plans</u>), other than Company Benefit Plans maintained outside of the United States primarily for the benefit of employees working outside of the United States (such plans are referred to as <u>Company</u>

A-15

#### Table of Contents

<u>Non</u> U.S. Benefit Plans ). Complete and accurate copies of all Company Benefit Plans listed on Section 5.9(a)-1 of the Company Disclosure Schedule, any amendments thereto, all summary plan descriptions, any summary of material modifications thereto, all other documents containing descriptions furnished to participants in a Company Benefit Plan, and any benefits schedule, trust instruments, insurance contracts or other funding vehicle forming a part of any such Company Benefit Plans, the Annual Report (Form 5500 series) and schedules, if any, for the most recent prior three (3) years and opinions of independent accountants to the extent required under applicable Law have been provided or made available to Parent. Section 5.9(a)-2 of the Company Disclosure Schedule identifies each Company Benefit Plan which is a change in control plan or agreement of the Company or any of its Subsidiaries and each employment or retention agreement of the Company or any of its Subsidiaries of each such plan or agreement have been provided to Parent.

(b) Except as set forth on Section 5.9(b) of the Company Disclosure Schedule, all Company Benefit Plans, other than Company Non U.S. Benefit Plans ( Company U.S. Benefit Plans ), have been maintained and administered in all material respects in accordance with ERISA, the Code and other applicable Laws. Each Company U.S. Benefit Plan which is subject to ERISA (the <u>Company ERISA Plans</u>) that is an employee pension benefit plan within the meaning of Section 3(2) of ERISA (a <u>Company Pension Plan</u>) and that is intended to be qualified under Section 401(a) of the Code, has received a current favorable opinion letter or determination letter from the Internal Revenue Service (the <u>IRS</u>), and the Company is not aware of any circumstances likely to result in the loss of the qualification of such Company Pension Plan under Section 401(a) of the Code. There is no voluntary employees beneficiary association within the meaning of Section 501(c)(9) of the Code which provides benefits under a Company U.S. Benefit Plan. Except as set forth on Section 5.9(b) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has engaged in a transaction with respect to any Company ERISA Plan that, assuming the Taxable period of such transaction expired as of the date of this Agreement, could subject the Company or any Subsidiary to a Tax or penalty imposed by either Section 4975 of the Code or Section 502(i) of ERISA in an amount which would be material. Neither the Company nor any of its Subsidiaries has incurred or reasonably expects to incur a material Tax or penalty imposed by Section 4980F of the Code or Section 502 of ERISA or any material liability under Section 4071 of ERISA. Any Company Benefit Plan that is subject to Section 409A of the Code has been operated in good faith compliance with the requirements of Section 409A of the Code (or an available exemption therefrom).

(c) Neither the Company nor any of its Subsidiaries or ERISA Affiliates contributes or ever has contributed to a multiemployer plan within the meaning of Section 4001(a)(3) of ERISA or a plan that has two or more contributing sponsors at least two of whom are not under common control, within the meaning of Section 4063 of ERISA. None of the Company Pension Plans is or ever has been subject to Section 302 of ERISA, Section 412 of the Code or Title IV of ERISA, nor does the Company, any of its Subsidiaries or any ERISA Affiliate have any liability, contingent or otherwise, in respect of any such Company Pension Plan.

(d) Except as set forth on Section 5.9(d) of the Company Disclosure Schedule, all contributions required to be made under each Company Benefit Plan, whether pursuant to applicable Laws or the terms of such Company Benefit Plan, have been timely made and all obligations in respect of each Company Benefit Plan have been properly accrued and reflected in the most recent consolidated balance sheet filed or incorporated by reference in the Company SEC Reports prior to the date of this Agreement.

(e) There is no material litigation pending or, to the knowledge of the Company, threatened, relating to the Company Benefit Plans. Neither the Company nor any of its Subsidiaries has any obligations for retiree health and life benefits under any Company ERISA Plan or collective bargaining agreement. By its terms, the Company or its Subsidiaries may amend or terminate any Company ERISA Plan at any time without incurring any liability thereunder, other than termination fees under service provider contracts with respect to each such Company ERISA Plan, a true and complete copy of each such service provider contract having been provided to Parent, and other than in respect of claims incurred or vested benefits accrued prior to such amendment or termination, and to the knowledge of the

Company, no summary plan description or other written communication distributed generally to participants or employees would prohibit the Company or its Subsidiaries from amending or terminating any such Company ERISA Plan.

(f) There has been no amendment to, announcement by the Company, any of its Subsidiaries or any ERISA Affiliate relating to, or change in employee participation or coverage under, any Company Benefit Plan which would increase materially the expense of maintaining such plan above the level of the expense incurred therefor for the most recent fiscal year. Section 5.9(f) of the Company Disclosure Schedule sets forth a complete and accurate list of all contracts, plans or arrangements obligating the Company or any of its Subsidiaries to pay severance to any current or former directors, employees, independent contractors or consultants (including outsourcing) of the Company or any of its Subsidiaries, except for obligations pursuant to, required by or arising under applicable Law and except for those agreements identified in Section 5.9(a)-2 of the Company Disclosure Schedule. Except pursuant to retention or other agreements set forth in Section 5.9(a)-2 of the Company Disclosure Schedule, neither the execution of this Agreement, stockholder approval of this Agreement nor the consummation of the transactions contemplated hereby (i) entitles any employees of the Company or any of its Subsidiaries to severance pay or any increase in severance pay upon any termination of employment after the date of this Agreement, or (ii) except as specifically contemplated in Section 4.3, accelerates the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increases the amount payable or results in any other material obligation pursuant to, any of the Company Benefit Plans or any Company Non-U.S. Benefit Plan. Neither the Company nor any of its Subsidiaries has entered into any contract, agreement, plan or arrangement covering any employee or former employee or independent contractor that, individually or collectively, could give rise to the payment by the Company or any of its Subsidiaries of any amount that would not be deductible by reason of Code Section 280G or would give rise to a payment that could subject the recipient to excise tax imposed by Code Section 4999.

(g) All Company Non-U.S. Benefit Plans have been maintained and administered in all material respects in accordance with applicable local Law, and have received all necessary rulings or determinations as to the qualification (to the extent such concept or a comparable concept exists in the relevant jurisdiction) of such Company Non-U.S. Benefit Plans from the appropriate Governmental Entity. All Company Non-U.S. Benefit Plans, and all governmental plans, funds or programs to which the Company or any of its Subsidiaries contributes on behalf of any of their employees, are listed on Section 5.9(g) of the Company Disclosure Schedule. There is no pending or, to the knowledge of the Company, threatened, litigation relating to the Company Non-U.S. Benefit Plans (except for individuals claims for benefits payable in the normal operation of such Company Non-U.S. Benefit Plans) that has resulted in, or is reasonably expected to result in, a material expense in respect of the Company or any of its Subsidiaries.

(h) All material contributions required to be made under each Company Non-U.S. Benefit Plan, whether pursuant to applicable Laws or the terms of such Company Non-U.S. Benefit Plan, have been timely made and all obligations in respect of each Company Non-U.S. Benefit Plan have been properly accrued and reflected in the most recent consolidated balance sheet filed or incorporated by reference in the Company SEC Reports prior to the date of this Agreement. The Company and its Subsidiaries have no material unfunded liabilities with respect to any such Company Non-U.S. Benefit Plan.

5.10 <u>Compliance with Laws; Permits</u>. Except as set forth on Section 5.10 of the Company Disclosure Schedule, the businesses of each of the Company and its Subsidiaries have been, and are being, conducted in compliance with all applicable federal, state, local, municipal, foreign or other laws, statutes, constitutions, principles of common law, resolutions, ordinances, codes, edicts, rules, regulations, judgments, orders, rulings, injunctions, decrees, directives, arbitration awards, agency requirements, authorizations, opinions, licenses and permits of all Governmental Entities (collectively, <u>Laws</u>) applicable to the Company or its Subsidiaries, except for violations or possible violations that (i) have not had, and are not reasonably expected to have, a Company Material Adverse Effect and (ii) have not resulted, and are not reasonably likely to result in, the imposition of a criminal fine, penalty or sanction against the Company, any of its Subsidiaries, or any of their respective directors or officers. No (i) material investigation or review (for which the Company or one of its Subsidiaries has received notice) or (ii) other investigation or review (for

which the Company or one of its Subsidiaries has received written notice) by any Governmental Entity with respect to the Company or any of its Subsidiaries is pending or, to the knowledge of the Company, threatened. The Company and its Subsidiaries each have all governmental permits, licenses, franchises, variances, exemptions, orders and other governmental authorizations, consents and approvals necessary to conduct its business as presently conducted (each, a <u>Company Permit</u>) except those the absence of

which have not had, and are not reasonably expected to have, a Company Material Adverse Effect. No Company Permit will cease to be effective as a result of the execution of this Agreement or the consummation of the transactions contemplated by this Agreement.

### 5.11 Environmental Matters.

(a) Except for such matters that are not reasonably expected to have a Company Material Adverse Effect: (i) the Company and its Subsidiaries have complied with all applicable Environmental Laws (as defined below) during the previous five (5) years; (ii) no property currently owned, leased or operated by the Company or any of its Subsidiaries (including soils, groundwater, surface water, buildings or other structures) is contaminated with any Hazardous Substance (as defined below) that requires, or is reasonably expected to require, investigation, monitoring, contribution or other financial responsibility and/or remediation by the Company or any of its Subsidiaries under applicable Environmental Laws; (iii) no property formerly owned or operated by the Company or any of its Subsidiaries was contaminated with any Hazardous Substance during or prior to such period of ownership or operation that requires, or is reasonably expected to require, investigation, monitoring, contribution or other financial responsibility and/or remediation by the Company or any of its Subsidiaries under applicable Environmental Laws; (iv) to the Company s knowledge, neither the Company nor any of its Subsidiaries is subject to liability for any Hazardous Substance disposal or contamination on any property owned by any Third Party; (v) neither the Company nor any of its Subsidiaries has caused or could be held liable for any release or threat of release of any Hazardous Substance; (vi) neither the Company nor any of its Subsidiaries has received any notice, demand, letter, claim or request for information alleging that the Company or any of its Subsidiaries may be in violation of or subject to liability under any Environmental Law; (vii) neither the Company nor any of its Subsidiaries is subject to any order, decree, injunction or other arrangement with any Governmental Entity or any indemnity or other agreement with any Third Party pursuant to which it has assumed any liability or obligation under any Environmental Law; (viii) there are no other existing circumstances or conditions (including plans for modification or expansion which are the subject of an approved capital authorization request) involving the Company s or any of its Subsidiaries owned or leased properties or operations that are reasonably likely to result in any claim, liability, investigation, cost or restriction on the Company s or any of its Subsidiaries ownership, use or transfer of any property pursuant to any Environmental Laws; and (ix) the Company has delivered or made available to Parent copies of all Material Environmental Reports (as defined below), studies, assessments, soil or groundwater sampling data and other material environmental information in its possession relating to the Company or its Subsidiaries or their respective current and former properties or operations which were prepared within the last five (5) years.

The term (x) <u>Material Environmental Reports</u> means any reports generated by any third party consultants or experts, including any due diligence reports prepared under the ASTM standards and any reports submitted to any Governmental Entity within the last five (5) years, (y) <u>Environmental Law</u> means any applicable Law relating to: (A) the protection, investigation or restoration of the environment, health, safety, or natural resources, (B) the handling, use, presence, disposal, release or threatened release of any Hazardous Substance, (C) noise, odor, indoor air, worker safety and health, wetlands, pollution or contamination, or any injury or threat of injury to Persons or property relating to any Hazardous Substance, or (D) the labeling, packaging, takeback or recycling of products or the manufacturing of products, and (z) <u>Hazardous Substance</u> means any substance that is listed, classified or regulated pursuant to any Environmental Law, including any petroleum product or by-product, asbestos-containing material, lead, polychlorinated biphenyls, radioactive material or radon.

(b) Except as set forth on Section 5.11 of the Company Disclosure Schedule, the products of the Company or any of its Subsidiaries sold or otherwise made available in the European Union market comply in all material respects with the Restrictions on the Use of Certain Hazardous Substances in Electrical and Electronic Equipment (2002/95/EC) Directive, and the Waste Electrical and Electronic Equipment (2002/96/EC) Directive, to the extent such directives and/or any legislation enacted or implemented thereunder by applicable European Union member nations are

applicable to such products.

### 5.12 <u>Taxes</u>.

(a) The Company and each of its Subsidiaries (a) have prepared in good faith and duly and timely filed (taking into account any extension of time within which to file) all income Tax Returns and other material Tax

Returns (as defined below) required to be filed by any of them and all such filed Tax Returns are complete and accurate in all material respects, (b) have paid or accrued for all Taxes that are required to be paid as shown in such Tax Returns or that the Company or any of its Subsidiaries are obligated to withhold from amounts owing to any employee, creditor or other Person, except with respect to matters contested in good faith, and (c) have not waived any statute of limitations with respect to any material Taxes that has continuing effect or agreed to any extension of time with respect to a Tax assessment or deficiency that has continuing effect. There are not pending or, to the knowledge of the Company, threatened, any audits, examinations, investigations or similar proceedings in respect of Taxes or Tax matters. The Company has made available to Parent correct and complete copies of the federal income Tax Returns filed by the Company and its Subsidiaries for each of their respective Taxable years ending in 2005, 2004 and 2003. Neither the Company nor any of its Subsidiaries has any liability with respect to income, franchise or similar Taxes in excess of the amounts accrued with respect thereto that are reflected in the financial statements included in the Company SEC Reports filed on or prior to the date of this Agreement, except for any liability with respect to such Taxes that has been incurred in the ordinary course of business since the date of filing of such Company SEC Reports. None of the Company or any of its Subsidiaries has any liability for Taxes of any Person other than members of the tax consolidated or combined group of which the Company is or was the common parent. None of the Company or any of its Subsidiaries was the distributing corporation or the controlled corporation in a distribution intended to qualify under Section 355(a) of the Code. Neither the Company nor any of its Subsidiaries has engaged in any transaction that the IRS has determined to be a listed transaction for purposes of § 1.6011-4(b)(2). With respect to any year for which the applicable statute of limitations is still open, none of the Company or any of its Subsidiaries has (i) engaged in a transaction of which it made disclosure to any Tax authority to avoid penalties, or (ii) participated in a tax amnesty or similar program offered by any Tax authority to avoid the assessment of penalties or other additions to Tax.

(b) The term (i) <u>Tax</u> (including, with correlative meaning, the terms <u>Taxes</u> and <u>Taxable</u>) means all federal, state, local and foreign income, profits, franchise, gross receipts, environmental, customs duty, capital stock, severances, stamp, payroll, sales, employment, unemployment, disability, use, property, withholding, excise, production, value added, occupancy and other taxes, duties or assessments of any nature whatsoever, together with all interest, penalties and additions imposed with respect to such amounts and any interest in respect of such penalties and additions, and (ii) <u>Tax Return</u> means all returns and reports (including elections, declarations, disclosures, schedules, estimates and information returns) required to be supplied to a Tax authority relating to Taxes.

## 5.13 Employees: Independent Contractors.

(a) The Company has provided to Parent a list of all employees of the Company and its Subsidiaries as of the date of this Agreement, and will provide at Closing an updated list of such employees as of immediately before the Effective Time (<u>Employees</u>), along with the position, date of hire and the annual rate of compensation of each such person (including salary or, with respect to Employees compensated on an hourly or per diem basis, the hourly or per diem rate of compensation and estimated or target annual incentive compensation), promised increases in compensation or increases contemplated by the Company, promised promotions or promotions contemplated by the Company, accrued but unused sick and vacation leave, and service credited for purposes of vesting and eligibility to participate under any Company Benefit Plans or Company Non-U.S. Benefit Plans, and has identified any Employees who are on a Company-approved leave of absence and the type of such approved leave. Except as set forth on Section 5.13(a) of the Company Disclosure Schedule, each such Employee has entered into a confidentiality and assignment of inventions agreement with the Company or a Subsidiary of the Company in the form set forth in Section 5.13(a) of the Company Disclosure Schedule.

(b) Section 5.13(b) of the Company Disclosure Schedule sets forth a list of all independent contractors performing services or under contract to perform future services for the Company or any of its Subsidiaries immediately before the Effective Time, and the Company has provided to Parent a copy of all contracts applicable to such independent

## Table of Contents

contractors. To the knowledge of the Company, the Company and its Subsidiaries have properly classified all such independent contractors under applicable Law.

#### Table of Contents

(c) To the knowledge of the Company, no Employee identified on Section 5.13(c) of the Company Disclosure Schedule under the heading Key Employee (Key Employee) has any plans to terminate employment with the Company or any of its Subsidiaries.

(d) Except as set forth on Section 5.13(d) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries is a party to or bound by any collective bargaining agreement, works council or representative of any employee group, or otherwise required to bargain with any union, works council or representative of any employee group, nor has the Company or any Subsidiary experienced within the last twenty-four (24) months any strikes or other industrial actions, grievances, claims of unfair labor practices, or other collective bargaining disputes or trade disputes. No organizational effort has been made or, to the knowledge of the Company, threatened by or on behalf of any labor union (which includes any application or request for recognition) within the last twenty-four (24) months with respect to any employees of the Company or any of its Subsidiaries. There is no union, works council or representative of any employee group that must be notified, consulted or with which negotiations need to be conducted in connection with the transactions contemplated by this Agreement.

(e) To the knowledge of the Company, neither the Company nor any of its Subsidiaries has committed any unfair labor practice, and each of the Company and its Subsidiaries has complied in all material respects with applicable Laws, including foreign Laws, their own respective policies, including handbooks, work rules, or internal regulations, within the last twenty-four (24) months relating to employment or employment practices or termination of employment, including those relating to wages and hours, including overtime, rest and meal periods, discrimination in employment opportunity, benefits, workers compensation, and collective bargaining, including any applicable foreign national collective bargaining agreement. To the knowledge of the Company, except as set forth on Section 5.13(e) of the Company Disclosure Schedule, there is no pending or threatened charge or complaint against the Company or any of its Subsidiaries involving any employment matter, including any charge or complaint before the National Labor Relations Board, the Equal Employment Opportunity Commission, or any comparable state, local, or foreign agency. To the knowledge of the Company classified as exempt or non-exempt in accordance with applicable Laws.

(f) The Company and its Subsidiaries are in compliance with the Fair Credit Reporting Act and state and foreign counterparts.

(g) The Company has distributed the California Department of Fair Employment and Housing pamphlet Sexual Harassment is Forbidden by Law (DFEH-185) or otherwise made a one-time dissemination of the DFEH mandated information to all former and current California employees.

(h) Except as set forth on Section 5.13(h) of the Company Disclosure Schedule, neither the Company, any director or officer of the Company, nor, to the knowledge of the Company, any immediate family members of any director or officer, owns directly or indirectly, individually or collectively, any interest in any corporation, company, partnership, entity or organization which is in a business similar or competitive to the businesses of the Company and its Subsidiaries or which has any existing undisclosed contractual relationship with the Company or any of its Subsidiaries, other than the ownership as a passive investor (i.e., where the Company, director, officer, or family member is not involved in any way in the management of the business) of less than one percent of the securities of a publicly traded corporation or mutual fund.

(i) The Company and its Subsidiaries have properly paid all wages and salaries and employment Taxes (including social security Taxes and other payroll Taxes and including any share owed by the employer and any share that the Company and its Subsidiaries were required to withhold from the compensation paid to Employees) and are not liable

for any penalties or arrears, except for any arrears that would exist in the ordinary course of business and would be in compliance in all material respects with applicable Law.

(j) To the knowledge of the Company, all Employees are authorized and have appropriate documentation to work in the countries in which they are assigned and the Company and its Subsidiaries are in compliance with all applicable immigration Laws.

(k) To the knowledge of the Company, except as set forth on Section 5.13(k) of the Company Disclosure Schedule, no former or current employee, consultant, or contractor of the Company or any of its Subsidiaries is in

A-20

violation of any agreement with the Company or any of its Subsidiaries relating to inventions, competition, solicitation or confidentiality.

(1) The Company and its Subsidiaries have not experienced a layoff or plant closing within the last twelve (12) months that is reasonably likely to give rise to liability under the Worker Adjustment and Retraining Notification Act or any similar state, local or foreign law or regulation.

(m) The Company and its Subsidiaries have no affirmative action obligations under applicable law.

5.14 <u>Insurance</u>. The Company and its Subsidiaries maintain insurance coverage reasonably adequate for the operation of the business of the Company and its Subsidiaries (taking into account the cost and availability of such insurance). Except as set forth on Section 5.14 of the Company Disclosure Schedule, since January 1, 2004, no insurer of the Company or any of its Subsidiaries has (a) cancelled or invalidated any insurance policy of the Company or any of its Subsidiaries or (b) refused any coverage or rejected any material claim under any such insurance policy, and each such insurance policy is in full force and effect and all premiums due with respect to all such insurance policies have been paid.

### 5.15 Intellectual Property.

(a) Section 5.15(a) of the Company Disclosure Schedule sets forth, for the Owned Intellectual Property (as defined below), a correct and complete list of all (i) issued Patents (as defined below) and filed and pending applications for Patents, (ii) registered Trademarks (as defined below) and Trademarks for which registrations have been applied for, (iii) domain name registrations, and (iv) registered Copyrights (as defined below) and Copyrights for which registrations have been applied for, indicating for each of the foregoing (i)-(iv), the applicable jurisdiction, registration number (or application number) and date issued (or date filed). The Company and its Subsidiaries exclusively own, free and clear of all Liens (except for Permitted Liens), all right, title and interest in the Owned Intellectual Property.

(b) All Trademarks, Patents and Copyrights listed in Section 5.15(a) of the Company Disclosure Schedule (i) are currently in compliance in all material respects with all applicable legal requirements (including, as applicable, application, registration and maintenance requirements, such as the timely post-registration filing of affidavits of use and incontestability and renewal applications with respect to Trademarks, and the payment of filing, examination and annuity and maintenance fees and proof of working or use with respect to Patents), (ii) are, to the knowledge of the Company, valid and enforceable and (iii) are not subject to any maintenance fees or actions falling due within ninety (90) days after the Closing Date. No Trademark listed in Section 5.15(a) of the Company Disclosure Schedule is currently involved in any opposition or cancellation proceeding and, to the knowledge of the Company, no such action has been threatened with respect to any of those Trademarks. No Patent owned by the Company is currently involved in any interference, reissue, re-examination or opposition proceeding and, to the knowledge of the Company, no such action has been threatened with respect to any such Patent.

(c) Section 5.15(c)-1 of the Company Disclosure Schedule sets forth a complete and accurate list of any and all Contracts or other written agreements (excluding license agreements for off-the-shelf software applications programs having a price of less than \$10,000 per copy, seat, CPU or named user) pursuant to which the Company or any of its Subsidiaries has been granted or otherwise receives from a Third Party any right to use or distribute any Software (as defined below) (other than the Third Party Embedded Software (as defined below) and any Software licensed pursuant to a Limited License (as defined below)), indicating for each such Contract and agreement the title, the parties and the date executed. Section 5.15(c)-2 of the Company Disclosure Schedule sets forth a complete and accurate list of all Third Party Software that is contained or embedded in, and necessary for the operation and use of, any commercially available products of the Company (<u>Third Party Embedded Software</u> and, together with the Contracts and agreements listed in Section 5.15(c)-1 of the Company Disclosure Schedule, the <u>Third Party Software License</u>). The Company

has identified in Section 5.15(c)-1 and Section 5.15(c)-2 of the Company Disclosure Schedule, for each Contract listed thereon, all royalties, honoraria or other fees (if any) that will become due or payable thereunder within five (5) months after the date of this Agreement.

(d) Except for the Third Party Software Licenses, Section 5.15(d) of the Company Disclosure Schedule sets forth a complete and accurate list of any and all Contracts or other written arrangements pursuant to which the Company or any of its Subsidiaries has been granted or otherwise receives any right to use, exercise or practice any

### Table of Contents

right under any Intellectual Property (as defined below) of a Third Party, indicating for each such Contract and written arrangement the title, the parties and the date executed (the <u>Third Party IP Licenses</u> and, together with the Third Party Software Licenses, the <u>Third Party Licenses</u>). To the knowledge of the Company, the Company and its Subsidiaries have valid and enforceable rights to use all of the Intellectual Property covered by the Third Party Licenses. The Company has identified in Section 5.15(d) of the Company Disclosure Schedule, for each Contract listed thereon, all royalties, honoraria or other fees (if any) that will become due or payable thereunder within five (5) months after the date of this Agreement. No royalties, honoraria or other fees are past due and owing by the Company or any of its Subsidiaries under the Third Party Licenses.

(e) Except as set forth on Section 5.15(e)-1 of the Company Disclosure Schedule, the Owned Intellectual Property and the Intellectual Property covered by the Third Party Licenses constitute all of the Intellectual Property used in and, to the knowledge of the Company, necessary for the operation of the Company s business as currently conducted. The Company and its Subsidiaries have taken all reasonable steps to protect the Owned Intellectual Property, including reasonable steps to prevent and abate any infringement or misappropriation of the Owned Intellectual Property. Except as set forth on Section 5.15(e)-2 of the Company bisclosure Schedule, to the knowledge of the Company, no Third Party has challenged in writing to the Company nor any of its Subsidiaries have licensed or otherwise authorized any Third Party to make, have made, sell, copy, distribute, modify, reverse engineer, or prepare derivatives of any Owned Intellectual Property (other than Copyrights in Company Software, which is addressed in the last sentence of Section 5.15(h) below), except pursuant to a written agreement (including via electronic means).

(f) Except as set forth on Section 5.15(f)-1 of the Company Disclosure Schedule, to the knowledge of the Company, the conduct of the Company s business as currently conducted and as it is intended to be conducted with respect to the development of the products and platforms set forth on Section 5.15(f)-2 of the Company Disclosure Schedule does not infringe upon any Intellectual Property rights of any Third Party. Except as set forth on Section 5.15(f)-3 of the Company Disclosure Schedule, no Third Party has notified the Company or any of the Company s Subsidiaries in writing that (i) any of such Third Party s Intellectual Property rights are infringed by the Company or any of its Subsidiaries, or (ii) the Company or any of its Subsidiaries, as applicable, to be non-infringing, and neither the Company nor any of its Subsidiaries has received any written offer to license (or any other form of written notice of) any of such Third Party s Intellectual Property rights.

(g) To the knowledge of the Company, no Third Party is misappropriating, infringing, diluting or violating any Owned Intellectual Property. Except as set forth on Section 5.15(g) of the Company Disclosure Schedule, no such claims have been brought or threatened against any Third Party by or on behalf of the Company or any of its Subsidiaries.

(h) Section 5.15(h) of the Company Disclosure Schedule contains a complete and accurate list of all Software that is owned by the Company or any of its Subsidiaries and sold, licensed, leased or otherwise distributed by the Company or any of its Subsidiaries or authorized resellers to end user customers of the Company s or its Subsidiaries products or services (the <u>Company Software</u>). The Company Software was developed either by (i) employees of the Company or its Subsidiaries within the scope of their employment who have executed the confidentiality and assignment of inventions agreement set forth in Section 5.13 of the Company Disclosure Schedule, or (ii) independent contractors who have assigned their rights to the Company or one of its Subsidiaries pursuant to enforceable written agreements. Neither the Company nor any of its Subsidiaries have licensed or otherwise authorized any Third Party to copy, distribute, modify, decompile, or prepare derivatives of any Company Software except pursuant to a written license agreement or other written arrangement.

(i) Except as set forth on Section 5.15(i)-1 of the Company Disclosure Schedule, all material Trademarks of the Company and its Subsidiaries within the Owned Intellectual Property and currently used in the operation of the

business of the Company or any of its Subsidiaries have been in continuous use by the Company or a Subsidiary of the Company, as applicable, since the date of their initial use in commerce. Except as set forth on Section 5.15(i)-2 of the Company Disclosure Schedule, to the knowledge of the Company, there has been no prior use of any registered Trademarks owned by the Company or any of its Subsidiaries or other action taken by any Third Party

that would confer upon such Third Party superior rights in such Trademarks. The Company has taken reasonable steps to prevent infringement of the Trademarks referenced in the first sentence of this subsection (i).

(j) The Copyrights within the Owned Intellectual Property have been solely (i) created by (A) employees of the Company and its Subsidiaries within the scope of their employment who have executed the confidentiality and assignment of inventions agreement set forth in Section 5.13(a) of the Company Disclosure Schedule, or (B) independent contractors who have assigned their rights to the Company pursuant to enforceable written agreements, or (ii) acquired pursuant to an enforceable written assignment from the original author(s) or subsequent assignees. To the knowledge of the Company, the works covered by such Copyrights were not copies of, nor derived from, any work for which the Company or any of its Subsidiaries does not own the Copyrights. To the knowledge of the Company, no other Person has any claim to authorship or ownership of any part of any of the Copyrights within the Owned Intellectual Property.

(k) The Patents within the Owned Intellectual Property relate solely to inventions (i) created by (A) employees of the Company and its Subsidiaries within the scope of their employment who have executed the confidentiality and assignment of inventions agreement set forth in Section 5.13(a) of the Company Disclosure Schedule, or (B) independent contractors who have assigned their rights to the Company pursuant to enforceable written agreements, or (ii) acquired pursuant to an enforceable written assignment from the original inventor(s) or subsequent assignees (a complete and accurate list of all such agreements referenced in subsection (i)(B) and (ii) above is set forth on Section 5.15(k) of the Company Disclosure Schedule). The inventions covered by such Patents were not copies of, nor derived from, any invention for which the Company or any of its Subsidiaries does not own the Patent, and no other Person has any claim to inventorship or ownership of any part thereof.

(1) The Company and its Subsidiaries have taken reasonable steps to protect their respective rights in material confidential information and trade secrets owned by them or disclosed to them by a Third Party and used in connection with the conduct of the Company s business. Without limiting the foregoing, the Company and its Subsidiaries have enforced a policy of requiring each employee, consultant and contractor to execute proprietary information, invention assignment and confidentiality agreements, as appropriate, substantially consistent with the Company s standard forms (complete and current copies of which have been delivered or made available to Parent). Except under valid and binding confidentiality obligations, there has been no material disclosure by the Company or any of its Subsidiaries to a Third Party of any confidential information or trade secrets used in connection with the conduct of the Company s business.

(m) The Company and its Subsidiaries have valid registrations for each of the domain names set forth in Section 5.15(a) of the Company Disclosure Schedule. The registration of each such domain name is free and clear of all Liens (except for Permitted Liens) and is in full force and effect. The Company has paid all fees required to maintain each such registration. To the knowledge of the Company, none of the Company s registrations or uses of the domain names has been disturbed or placed on hold and neither the Company nor any of its Subsidiaries has received written notice of any claim asserted against the Company or any of its Subsidiaries adverse to its rights to such domain names.

(n) All Company Software is free from any material defect or programming or documentation error, including major bugs, logic errors or failures of such Software to operate in all material respects as described in the related documentation, and substantially conforms to the specifications of such Software. To the knowledge of the Company, all Software licensed from any Third Party is free from any material defect or programming or documentation error, including major bugs, logic errors or failures of such Software to operate in all material respects as described in the related documentation, and substantially conforms to the specifications of such Software. With respect to Company Software, the applications can be compiled from the associated source code in accordance with the means currently employed by the Company. Except for any components of the source code licensed in from Third Parties, the

Company has actual and sole possession of the complete source code of the Company Software. Except as set forth on Section 5.15(n) of the Company Disclosure Schedule, other than the Company s or any of its Subsidiaries delivery of the Company source code to third party escrow agents or their disclosure of such source code to Third Parties as part of a software development kit made available by the Company or any of its Subsidiaries in the ordinary course of business, no event has occurred, and to the knowledge of the Company no circumstance or condition exists, that (with or without notice or lapse of time) will, or could

reasonably be expected to, result in the disclosure or delivery to any Third Party of the source code for the Company Software. The Company Software (as used or distributed by the Company or its Subsidiaries) does not contain any back door, time bomb, Trojan horse, worm, drop dead device, virus (as these terms are commonly used in the computer software industry), or other Software routines or hardware components intentionally designed to permit unauthorized access, to disrupt, disable or erase software, hardware or data, or to perform any other similar type of unauthorized activities.

(o) Except as set forth in Section 5.15(o) of the Company Disclosure Schedule, none of the Company Software or any Owned Intellectual Property are, in whole or in part, subject to the provision of any open source or other similar type of license agreement or distribution model that (i) requires the distribution or making available of the source code for the Company Software to the general public, (ii) prohibits or limits the Company or any of its Subsidiaries from charging a fee or receiving consideration in connection with sublicensing or distributing any Company Software, (iii) except as specifically permitted by Law, grants any right to any Third Party (other than the Company and its Subsidiaries) or otherwise allows any such Third Party to decompile, disassemble or otherwise reverse-engineer any Company Software, or (iv) requires the licensing of any Company Software to the general public for the purpose of permitting others to make derivative works of the Company Software (any such open source or other type of license agreement or distribution model described in clause (i), (ii), (iii) or (iv) above, a Limited License ). By way of clarification, but not limitation, the term Limited License includes (A) GNU s General Public License (GPL) or Lesser/Library GPL (LGPL), (B) the Artistic License (e.g., PERL), (C) the Mozilla Public License, (D) the Netscape Public License, (E) the Sun Community Source License (SCSL), and (F) the Sun Industry Standards License (SISL). To the knowledge of the Company, none of the Company Software incorporates, or is distributed with, any Software that is subject to a Limited License, nor does any Company Software constitute a derivative work of, dynamically link with or otherwise interact with any such Software.

(p) Except as set forth on Section 5.15(p) of the Company Disclosure Schedule, no government funding, facilities of a university, college, or other educational institution or research center was used in the creation or development of the Owned Intellectual Property or Company Software. To the knowledge of the Company, no current or former employee, consultant or independent contractor who was directly involved in, or who contributed directly to, the creation or development of any Owned Intellectual Property or Company Software has performed services for any Governmental Entity, a university, college, or other educational institution, or a research center, during a period of time during which such employee, consultant or independent contractor was also performing services used in the creation or development of the Owned Intellectual Property or Company Software. Neither the Company nor any of its Subsidiaries are party to any contract, license or agreement with any Governmental Entity that grants to such Governmental Entity any right or license with respect to the Owned Intellectual Property or Company Software, other than as granted in the ordinary course of business pursuant to a non-exclusive license to any Company Software.

## (q) For the purposes of this Agreement:

(i) The term <u>Copyrights</u> means (A) any rights in original works of authorship fixed in any tangible medium of expression as set forth in the United States Copyright Act, 17 U.S.C. §101 et. seq., and any rights in mask works, registered and unregistered, as defined in 17 U.S.C. §901, (B) all registrations and applications to register the foregoing anywhere in the world, (C) all foreign counterparts and analogous rights anywhere in the world, and (D) all rights in and to any of the foregoing;

(ii) The term <u>Intellectual Property</u> means any and all (A) Copyrights, Trademarks, and Patents and all rights to obtain and rights to apply for Patents, and to register Trademarks and Copyrights, (B) Software, ideas, innovations, inventions (whether or not patentable, reduced to practice, or the subject of an application for Patent), know-how and show-how, trade secrets, works of authorship, and confidential technical and non-technical information, (C) moral rights and author s rights, (D) all other industrial, proprietary and intellectual property related rights anywhere in the

world, and all renewals and extensions of any of the foregoing, regardless of whether or not such rights have been registered with the appropriate authorities in such jurisdictions in accordance with the relevant legislation, (E) copies and tangible embodiments of all the foregoing, in whatever form or medium, (F) all rights in and to any of the

### Table of Contents

foregoing, including the right to sue, recover, and retain damages, costs, and attorneys fees for past and present infringement or misappropriation of any of the foregoing;

(iii) The term <u>Owned Intellectual Property</u> means Intellectual Property currently owned by or subject to an obligation to be assigned to the Company and its Subsidiaries;

(iv) The term <u>Paten</u>ts means (A) all classes and types of patents (including national and multinational statutory invention registrations, utility models, petty patents, design patents and industrial designs) and the inventions covered thereby and any enhancements or improvements thereto (including the exclusive right to use, make, have made, sell, offer to sell and import the inventions), (B) invention disclosures, provisional patent applications, patent applications, continuations-in-part, divisionals or substitutes of the original applications upon which the any of foregoing patent rights are based, (C) any reexaminations, reissues, renewals or extensions of any of the foregoing, (D) foreign counterparts (including national and multinational) of any of the foregoing, and (E) all rights in and to any of the foregoing;

(v) The term <u>Software</u> means all computer programs and systems, whether embodied in software, firmware or otherwise, including, software compilations, software implementations of algorithms, software tool sets, compilers, and software models and methodologies (regardless of the stage of development or completion), and all related documentation, and including any and all forms in which any of the foregoing is embodied (whether in source code, object code, executable code or human readable form); and

(vi) The term <u>Trademarks</u> means (A) all classes and types of trademarks, service marks, logos, trade dress and trade names, Web addresses and domain names, and other indicia of commercial source or origin (whether registered, common law, statutory or otherwise), (B) registrations and pending applications to register any of the foregoing including any intent to use applications, supplemental registrations and any renewals or extensions, (C) goodwill associated with any of the foregoing, (D) foreign counterparts of any of the foregoing anywhere in the world, and (E) all rights in and to any of the foregoing.

### 5.16 Owned and Leased Properties.

(a) Neither the Company nor any Subsidiary of the Company currently owns or has ever owned any real property.

(b) Section 5.16(b) of the Company Disclosure Schedule sets forth a complete and accurate list as of the date of this Agreement of all real property leased or subleased by the Company or any of its Subsidiaries with space in excess of 5,000 square feet (the <u>Leased Real Property</u>), together with a true and complete list of all leases (including the parties thereto, date and address of the real property covered by the leases), lease guaranties, subleases, licenses, easements, and any other agreements for the leasing, use or occupancy of, or otherwise granting a right in or relating to the Leased Real Property with space in excess of 5,000 square feet, entered into by the Company or any of its Subsidiaries, including all amendments, terminations and modifications (each, a <u>Company Lease</u>). The Company has made available to Parent complete and accurate copies of all Company Leases. The Company or one of its Subsidiaries has a valid and existing leasehold estate in and the right to quiet enjoyment of the Leased Real Property for the full term, subject to the terms of the Company Leases, to any Permitted Liens (as defined below) with respect thereto and to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors rights and remedies generally, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

(c) With respect to each Company Lease: (i) the Company Lease as modified or amended is legal, valid, binding, enforceable by the Company or any of its Subsidiaries which is a party thereto (the <u>Tenant</u>), and in full force and effect; (ii) except as set forth on Section 5.16(c)(ii) of the Company Disclosure Schedule, the Company Lease will

continue to be legal, valid, binding, enforceable, and in full force and effect on substantially the same terms following the consummation of the Merger, and the landlord will not be entitled to terminate such Company Lease upon the Merger; (iii) the Tenant has not received or given any notice of any material default or event that, with notice or lapse of time or both, would constitute a default by the Tenant under the Company Lease for which such Company Lease could be terminated and, to the knowledge of the Company, no other party is in material

default thereof and no party to the Company Lease has exercised any termination rights with respect thereto as a result of an event of default; (iv) neither the Tenant nor, to the knowledge of the Company, any other party has repudiated any material provision of any Company Lease; (v) neither Tenant nor, to the knowledge of the Company, any other party to the Company Lease, is engaged in any material dispute, oral agreement or forbearance program with respect to the Company Lease which could have a material adverse effect on the rights or obligations of the Tenant under the Company Lease; (vi) except for Permitted Liens, or as set forth on Section 5.16(c)(vi) of the Company Disclosure Schedule, the Tenant has not subleased, assigned, transferred, conveyed, mortgaged, granted a deed of trust or encumbered its leasehold interest in the Leased Real Property subject to the Company Lease; (vii) the monthly rent and all other charges due and payable by the Tenant under such Company Lease have been paid in full through the respective dates such amounts are due thereunder; (viii) to the extent the Tenant is responsible therefor under the Company Lease, all facilities leased under the Company Lease have received all approvals of Governmental Entities (including licenses and permits) required in connection with the operation and use thereof by the Tenant and have been operated and maintained by the Tenant in accordance with applicable laws, rules and regulations, except for any such approvals, laws, rules or regulations, which if not obtained, or if not operated and maintained in accordance with which, would not materially and adversely affect the present use by the Tenant of the Leased Real Property; (ix) all facilities leased under the Company Lease are supplied with utilities and other services necessary for the operation of said facilities as used by the Tenant; (x) except as set forth on Section 5.16(c)(x) of the Company Disclosure Schedule, there are no parties (other than the Company and its Subsidiaries) in possession of such Leased Real Property; (xi) the Tenant has not received written notice of, nor does the Company have knowledge of, any pending or threatened condemnation proceedings, lawsuits or administrative actions relating to Leased Real Property subject to the Company Lease; (xii) neither the Company nor any of its Subsidiaries owes any brokerage commissions with respect to any such Leased Real Property; and (xiii) all real estate Taxes and assessments owed by the Tenant which are due and payable with respect to the Leased Real Property have been paid prior to the delinquency thereof.

(d) The Company has not received any notice of any special Tax assessment affecting any real property which the Company or any of its Subsidiaries is responsible for paying and, to the knowledge of the Company, no such assessments are pending or threatened. The zoning of each parcel of Leased Real Property permits the presently existing improvements thereon and the continuation of the business presently being conducted on such parcel. Neither the Company nor any of its Subsidiaries subleases any Leased Real Property to any Third Party other than to the Company and its Subsidiaries.

(e) Except as set forth on Section 5.16(e) of the Company Disclosure Schedule, the Company and each of its Subsidiaries, as applicable, has good and marketable title to, or valid leasehold interests in, all of its material tangible assets and properties, including the Leased Real Property, except for (i) such tangible assets and properties which are disposed of or, with respect to the Leased Real Property, the leasehold interests in which are terminated or expire, in the ordinary course of business, (ii) Permitted Liens and (iii) any other defects in title, easements, restrictive covenants and other encumbrances of any nature that have not had, and are not reasonably expected to have, a Company Material Adverse Effect. All such material tangible assets and properties, other than assets and properties in which the Company or any of its Subsidiaries has a leasehold interest, are free and clear of all Liens, except for (A) Liens for Taxes which are not yet due and payable or delinquent or that are being contested in good faith by appropriate proceedings, (B) Liens for assessments and other governmental charges or Liens of landlords, carriers, warehousemen, mechanics and repairmen incurred in the ordinary course of business, in each case for sums not yet due and payable or due but not delinquent or being contested in good faith by appropriate proceedings, (C) Liens incurred in the ordinary course of business in connection with workers compensation, unemployment insurance and other types of social security or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, performance and return of money bonds and similar obligations, (D) Liens that do not materially interfere with the conduct of the Company s business and do not materially adversely affect the present use or value of the Company s assets, (E) zoning, building and other land use and environmental regulations by any Governmental Entities which are not currently violated or with respect to which the violation, if any, does not

materially interfere with the conduct of the Company s business and does not materially adversely affect the present use by the Company and its Subsidiaries of the Leased Real Property, (F) such other imperfections or irregularities in title, charges, easements, survey exceptions, leases, subleases and other occupancy agreements, reciprocal easement agreements, restrictions and other encumbrances on title that do not

#### Table of Contents

materially interfere with the conduct of the Company s business and do not materially adversely affect the present use by the Company and its Subsidiaries of the Leased Real Property, (G) as to the Leased Real Property, Liens affecting the lessor which have not been created by the Company or any of its Subsidiaries or caused by the actions of the Company or any of its Subsidiaries, (H) Liens relating to any debt or liabilities, including any contingent liabilities, that are reflected on the most recent consolidated balance sheet of the Company and its Subsidiaries, (I) matters which an accurate survey would disclose, provided such matters do not interfere with the present use or occupancy of the property subject thereto or affected thereby, (J) such other exceptions to or imperfections in title, charges, easements, covenants, conditions, restrictions and encumbrances which do not materially interfere with the present use of any property subject thereto or affected thereby and (K) Liens consented to in writing pursuant to <u>Section 7.1</u> by Parent (such Liens set forth in clauses (A) through (K) constituting, <u>Permitted Liens</u>).

5.17 <u>Government Contracts</u>. With respect to each Contract between the Company or any of its Subsidiaries, on the one hand, and any Governmental Entity, on the other hand, and each outstanding bid, quotation or proposal by the Company or any of its Subsidiaries (each, a <u>Bid</u>) that if accepted or awarded could lead to a Contract between the Company or any of its Subsidiaries, on the one hand, and any Governmental Entity, on the other hand (each such Contract or Bid, a <u>Company Government Contract</u>), and each Contract between the Company or any of its Subsidiaries, on the one hand, and any prime contractor or upper-tier subcontractor, on the other hand, relating to a Contract between the Company or any of its Subsidiaries to a Contract between the Company or any of its Subsidiaries, on the other hand, relating to a Contract between the Company or any of its Subsidiaries, on the other hand, relating to a Contract between the Company or any of its Subsidiaries, on the other hand, relating to a Contract between the Company or any of its Subsidiaries, on the other hand, relating to a Contract between the Company or any of its Subsidiaries, on the one hand, and a prime contractor or upper-tier subcontractor, on the other hand, relating to a Contract between such Person and any Governmental Entity between such Person and any Governmental Entity (each such Contract or Bid, a <u>Company Government Subcontract</u>):

(a) Each such Company Government Contract or Company Government Subcontract (other than Bids) (i) is set forth on Section 5.17(a) of the Company Disclosure Schedule, and (ii) was, to the knowledge of the Company, legally awarded, and, unless fully performed in accordance with its terms, is binding on the parties thereto and in full force and effect, except any failure to be legally awarded or in full force and effect that, individually or in the aggregate, is not reasonably likely to result in a material liability to the Company and its Subsidiaries, taken as a whole.

(b) There is no material action, suit, claim or proceeding pending or, to the knowledge of the Company, threatened, in connection with any Company Government Contract or Company Government Subcontract, against the Company or any of its Subsidiaries alleging fraud or under the United States False Claims Act, the United States Procurement Integrity Act or the United States Truth in Negotiations Act. Neither the Company, any Subsidiary of the Company nor any cost incurred by the Company or any of its Subsidiaries pertaining to a Company Government Contract or Company Government Subcontract is the subject of any audit or, to the knowledge of the Company, investigation or has been disallowed by any Governmental Entity, except any investigation, audit or disallowance (i) that, individually or in the aggregate, is not reasonably likely to result in a material liability to the Company and its Subsidiaries taken as a whole or (ii) which commenced more than three (3) years before the date of this Agreement and is closed.

(c) The Company and its Subsidiaries have complied in all material respects with all requirements of the Company Government Contracts or Company Government Subcontracts and any material Law relating to the safeguarding of, and access to, classified information. The Company is not aware of any facts that are reasonably likely to give rise to the revocation of any security clearance of the Company, any of its Subsidiaries or any employee of the Company or any of its Subsidiaries.

5.18 <u>Import and Export Control Laws</u>. The Company and each of its Subsidiaries has at all times as to which the applicable statute of limitations has not yet expired, conducted its import and export transactions in accordance in all material respects with all applicable U.S. import, export and re-export controls, including the United States Export Administration Act and Regulations and Foreign Assets Control Regulations, and all other applicable import/export controls in other countries in which the Company and its Subsidiaries conduct material business. Without limiting the

foregoing:

(a) The Company and each of its Subsidiaries has obtained, and is in compliance in all material respects with, all material export licenses, license exceptions and other consents, notices, waivers, approvals, orders, authorizations, registrations, declarations, classifications and filings with any Governmental Entity required for

#### Table of Contents

(i) the export and re-export of products, services, Software and technologies and (ii) releases of technologies and Software to foreign nationals located in the United States and abroad (<u>Export Approvals</u>);

(b) There are no pending or, to the knowledge of the Company, threatened claims against the Company or any of its Subsidiaries with respect to such Export Approvals;

(c) To the knowledge of the Company, there are no actions, conditions or circumstances pertaining to the Company s or any of its Subsidiaries import or export transactions that may give rise to any future claims;

(d) Except as set forth on Section 5.18(d) of the Company Disclosure Schedule, no Export Approvals for the transfer of export licenses to Parent or the Surviving Corporation are required, or if any such Export Approvals are required, they can be obtained expeditiously without material cost;

(e) None of the Company, its Subsidiaries or any of their respective Affiliates is a party to any Contract or bid with, or has conducted business with (directly or, to the knowledge of the Company, indirectly), a Person located in, or otherwise has any operations in, or sales to, Cuba, Myanmar (Burma), Iran, Iraq, North Korea, Libya, Rwanda, Syria or Sudan;

(f) Since January 1, 2004, neither the Company nor any of its Subsidiaries has received written notice to the effect that a Governmental Entity claimed or alleged that the Company or any of its Subsidiaries was not in compliance in any material respect with any applicable Laws relating to the export of goods and services to any foreign jurisdiction against which the United States or the United Nations maintains sanctions or export controls, including applicable regulations of the United States Department of Commerce and the United States Department of State; and

(g) None of the Company, its Subsidiaries or any of their respective Affiliates has made any voluntary disclosures to, or has been subject to any fines, penalties or sanctions from, any Governmental Entity regarding any past import or export control violations.

5.19 <u>Foreign Corrupt Practices Act</u>. Neither the Company nor any of its Subsidiaries (including any of their officers, directors, agents, distributors, employees or other Person associated with or acting on their behalf) has, directly or indirectly, taken any action which would cause it to be in material violation of the Foreign Corrupt Practices Act of 1977, as amended, or any rules or regulations thereunder or any similar anti-corruption or anti-bribery Laws applicable to the Company or any of its Subsidiaries in any jurisdiction other than the United States (collectively, the <u>FCPA</u>), or, to the knowledge of the Company, used any corporate funds for unlawful contributions, gifts,

entertainment or other unlawful expenses relating to political activity, made, offered or authorized any unlawful payment to foreign or domestic government officials or employees, whether directly or indirectly, or made, offered or authorized any unlawful bribe, rebate, payoff, influence payment, kickback or other similar unlawful payment, whether directly or indirectly, except for any of the foregoing which is no longer subject to potential claims of violation as a result of the expiration of the applicable statute of limitations. The Company has established reasonable internal controls and procedures intended to ensure compliance with the FCPA and has made available to Parent copies of any such written controls and procedures.

5.20 <u>Consent Decrees</u>. Section 5.20 of the Company Disclosure Schedule sets forth a list of all material consent decrees to which the Company or any of its Subsidiaries is subject and any material voluntary agreements with any Governmental Entity resulting from any Actions by any Governmental Entity that impose any continuing duties on the Company, including any additional reporting or monitoring requirements.

### 5.21 Product Liability and Recalls.

(a) Neither the Company nor any of its Subsidiaries has any material liability (and, to the knowledge of the Company, there is no reasonable basis for any present or future action, suit, proceeding, hearing, investigation, charge, complaint, claim, or demand against any of them giving rise to any material liability) arising out of any injury to individuals or property as a result of the license or use of any product of the Company or any of its Subsidiaries.

(b) There are no pending internal investigations, material external investigations for which the Company has received notice, other external investigations for which the Company has received written notice, or voluntary

#### Table of Contents

or involuntary recalls, of any product of the Company or any of its Subsidiaries nor, to the knowledge of the Company, has it received any notifications from any Third Party or Governmental Entity that might give rise to any potential investigation, or the recall, of any product of the Company or any of its Subsidiaries. Each product that is sold or licensed by the Company or any of its Subsidiaries is designed and manufactured, and functions or operates, in all material respects in accordance with such product s design or specifications, and in accordance with applicable product safety or regulatory requirements.

5.22 <u>Takeover Statutes</u>. The Company Board has taken all actions so that the restrictions contained in Section 203 of Delaware Law applicable to a business combination (as defined in such Section 203) will not apply to Parent during the pendency of this Agreement, including the execution, delivery or performance of this Agreement and the consummation of the Merger and the other transactions contemplated hereby. No other fair price , moratorium , control share acquisition , interested shareholder , business combination or other similar anti-takeover statute or regulation of any jurisdiction (each, including the business combination provisions of Section 203 of Delaware Law, a <u>Takeover Statute</u> ) is applicable to the Merger or any of the other transactions contemplated hereby.

5.23 <u>Change of Control</u>. The Merger and the other transactions contemplated hereby will not constitute a change of control under, require the consent from or the giving of notice to a Person pursuant to, permit a Person to terminate or accelerate vesting or repurchase rights or create any other detriment to the Company or any Subsidiary of the Company under the terms, conditions or provisions of, any Company Material Contract or Company Lease to which the Company or any Subsidiary of the Company is a party or by which any of them or any of their properties or assets may be bound.

5.24 <u>Vote Required</u>. The approval of this Agreement by the Company Board constitutes approval of this Agreement and the Merger for purposes of Section 203 of Delaware Law and represents the only action necessary to ensure that Section 203 of Delaware Law does not and will not apply to the execution and delivery of this Agreement or the consummation of the Merger. The only vote of the holders of any class or series of capital stock or other securities of the Company necessary to adopt this Agreement or consummate the other transactions contemplated hereby is the Shareholder Approval.

5.25 <u>Company Rights Plan</u>. The Company has amended, and the Company and the Company Board have taken all necessary action to amend, the Company Rights Agreement to render the Rights issuable pursuant to the Company Rights Agreement inapplicable to the execution and delivery of this Agreement and consummation of the Merger and ensure that none of the execution or delivery of this Agreement or the consummation of the Merger will result in (a) the occurrence of the flip-in event described under Section 11(a)(ii) of the Company Rights Agreement, (b) the occurrence of the flip-over event described under Section 13(a) of the Company Rights Agreement, or (c) the Rights becoming evidenced by, and transferable pursuant to, certificates separate from the certificates representing the shares of Company Common Stock. The Company and the Company Board have taken all actions necessary to ensure that the Rights shall expire immediately prior to the Effective Time, without the payment of any money or other consideration.

5.26 <u>Brokers and Finders</u>. Except for the Company s engagement of Goldman, Sachs & Co. as its financial advisor, neither the Company nor any of its officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finder s fees in connection with the Merger or the other transactions contemplated in this Agreement. The Company has made available to Parent a complete and accurate copy of all agreements pursuant to which Goldman, Sachs & Co. is entitled to any fees or expenses in connection with any of the transactions contemplated by this Agreement.

ARTICLE VI

### REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Parent and Merger Sub represent and warrant to the Company that the statements contained in this <u>Article VI</u> are true and correct.

#### Table of Contents

6.1 <u>Organization, Good Standing and Qualification</u>. Each of Parent and Merger Sub is a corporation duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation, has all requisite corporate power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted, and is duly qualified to do business and, where applicable as a legal concept, is in good standing as a foreign corporation in each jurisdiction where the ownership or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in good standing, or to have such power or authority, when taken together with all other such failures, has not had, and is not reasonably expected to have, a Parent Material Adverse Effect. The term <u>Parent Material Adverse Effect</u> means any change or effect that materially and adversely affects the ability of Parent or Merger Sub to consummate, or materially delays the consummation of, the Merger and the other transactions contemplated by this Agreement.

### 6.2 Authority; No Conflict; Required Filings and Consents.

(a) Each of Parent and Merger Sub has all requisite corporate power and authority to enter into this Agreement and to consummate the transactions contemplated by this Agreement. The execution and delivery of this Agreement and the consummation of the transactions contemplated by this Agreement by Parent and Merger Sub have been duly authorized by all necessary corporate action on the part of each of Parent and Merger Sub. This Agreement has been duly executed and delivered by each of Parent and Merger Sub and constitutes the valid and binding obligation of each of Parent and Merger Sub, enforceable against each of them in accordance with its terms, except as the enforcement may be limited by bankruptcy, insolvency (including all Laws relating to fraudulent transfers), reorganization, moratorium or similar Laws affecting enforcement of creditors rights generally now or hereafter in effect and except as enforcement is subject to general principles of equity (regardless of whether enforcement is considered in a proceeding in equity or at Law).

(b) The execution and delivery of this Agreement by each of Parent and Merger Sub do not, and the consummation by Parent and Merger Sub of the transactions contemplated by this Agreement will not, (i) conflict with, or result in any violation or breach of, any provision of the certificate of incorporation or by-laws of Parent or Merger Sub, (ii) conflict with, or result in any violation or breach of, or constitute (with or without notice or lapse of time, or both) a default (or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any material benefit) under, require a consent or waiver under, constitute a change in control under, require the payment of a penalty under or result in the imposition of any Lien on Parent s or Merger Sub s assets under, any of the terms, conditions or provisions of any Contract to which Parent or Merger Sub is a party or by which any of them or any of their properties or assets may be bound, or (iii) subject to compliance with the requirements specified in clauses (i) and (ii) of Section 6.2(c), conflict with or violate any permit, concession, franchise, license or Law applicable to Parent or Merger Sub or any of its or their respective properties or assets, except in the case of clauses (ii) and (iii) of this Section 6.2(b) for any such conflicts, violations, breaches, defaults, terminations, cancellations, accelerations, losses, penalties or Liens, and for any consents or waivers not obtained, that, individually or in the aggregate, are not reasonably expected to have a Parent Material Adverse Effect.

(c) Other than (i) the filings, approvals and/or notices pursuant to <u>Section 1.2</u>, (ii) filings, approvals and/or notices under the HSR Act (or similar foreign filings), the Securities Act and the rules and regulations promulgated thereunder and the Exchange Act and the rules and regulations promulgated thereunder, (iii) filings, approvals and/or notices required to be made with or obtained from the New York Stock Exchange and (iv) such consents, approvals, orders, authorizations, registrations, declarations and filings as may be required under applicable U.S. state securities Laws, no notices, reports or other filings are required to be made by Parent with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by Parent from, any Governmental Entity in connection with the execution and delivery of this Agreement by Parent and Merger Sub and the consummation by Parent and Merger Sub of the Merger and the other transactions contemplated hereby, except those that the failure to make or

obtain could not, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect.

(d) No vote of the holders of any class or series of Parent s capital stock or other securities is necessary for the consummation by Parent of the transactions contemplated by this Agreement.

#### Table of Contents

6.3 <u>Information Provided</u>. The information concerning Parent and Merger Sub that is supplied by or on behalf of Parent expressly for inclusion in the Proxy Statement to be sent to the stockholders of the Company in connection with the Company Meeting will not, on the date it is first mailed to the stockholders of the Company or at the time of the Company Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated in the Proxy Statement or necessary in order to make the statements in the Proxy Statement, in light of the circumstances under which they are made, not misleading.

6.4 <u>Operations of Merger Sub</u>. Merger Sub was formed solely for the purpose of engaging in the transactions contemplated by this Agreement, has engaged in no other business activities and has conducted its operations only as contemplated by this Agreement.

6.5 <u>Brokers and Finders</u>. Neither Parent nor Merger Sub has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finder s fees for which the Company would have liability in connection with the Merger or the other transactions contemplated in this Agreement.

6.6 <u>Financing</u>. Parent has immediately available funds to pay, in cash, the total Merger Consideration and necessary to perform Parent s and Merger Sub s other obligations under this Agreement (including all fees and expenses payable by Parent and/or Merger Sub in connection with the transactions contemplated by this Agreement).

### ARTICLE VII

### **COVENANTS**

7.1 <u>Interim Operations</u>. The Company covenants and agrees as to itself and its Subsidiaries that, after the date of this Agreement and prior to the Effective Time (unless Parent shall otherwise consent in writing, which consent will not unreasonably be withheld or delayed, and except as otherwise expressly set forth in or contemplated by this Agreement or Section 7.1 of the Company Disclosure Schedule):

(a) the business of it and its Subsidiaries will be conducted in the ordinary and usual course and, to the extent consistent therewith, it and its Subsidiaries shall use their respective commercially reasonable efforts to preserve its business organization intact and maintain its existing relations and goodwill with customers, suppliers, distributors, strategic partners, creditors, lessors, employees and business associates;

(b) it shall not (i) issue, sell, pledge, dispose of or encumber any capital stock owned by it in any of its Subsidiaries, (ii) amend its certificate of incorporation or by-laws (or its equivalent governing instruments), (iii) split, combine or reclassify its outstanding shares of capital stock, (iv) declare, set aside or pay any dividend payable in cash, stock or property in respect of any capital stock other than dividends from its direct or indirect wholly-owned Subsidiaries, or (v) purchase, redeem or otherwise acquire, except for the acquisition of shares of Company Common Stock from holders of Company Stock Options in full or partial payment of the exercise price payable by such holder upon exercise of Company Stock Options to the extent required or permitted under the terms of such Company Stock Options, or permit any of its Subsidiaries to purchase or otherwise acquire, any shares of its capital stock or any securities convertible into or exchangeable or exercisable for any shares of its capital stock;

(c) neither it nor any of its Subsidiaries shall (i) authorize, issue, sell, pledge, dispose of or encumber any shares of, or securities convertible into or exchangeable or exercisable for, or options, warrants, calls, commitments or rights of any kind (including any rights or poison pill agreement) to acquire, any shares of its capital stock of any class, or any Voting Debt or any other property or assets (other than shares of Company Common Stock and associated rights issuable pursuant to options and other stock-based awards outstanding on the date of this Agreement under the

Company Stock Plans), or (ii) transfer, lease, license, guarantee, sell, mortgage, pledge, dispose of, abandon, cancel, surrender or allow to lapse or expire or encumber any material property or material assets (including capital stock of any of its Subsidiaries) or business, other than the sale of inventory in the ordinary course of business;

(d) neither it nor any of its Subsidiaries shall restructure, recapitalize, reorganize or completely or partially liquidate or adopt a plan of complete or partial liquidation or otherwise enter into any agreement or

arrangement imposing material changes or restrictions on the operation of its assets or businesses or adopt resolutions providing for or authorizing any of the foregoing;

(e) neither it nor any of its Subsidiaries shall acquire (i) by merging or consolidating with, or by purchasing all or a substantial portion of the assets of or any stock of, or by any other manner, any business or any corporation, partnership, joint venture, limited liability company, association or other business organization or division thereof, or (ii) any assets that are material, individually or in the aggregate, to the Company and any of its Subsidiaries, taken as a whole, except purchases of inventory and raw materials in the ordinary course of business;

(f) neither it nor any of its Subsidiaries shall (i) incur any Indebtedness or guarantee any such Indebtedness of another Person, (ii) issue, sell or amend any debt securities or warrants or other rights to acquire any debt securities of the Company or any of its Subsidiaries, guarantee any debt securities of another Person, enter into any keep well or other agreement to maintain any financial statement condition of another Person or enter into any arrangement having the economic effect of any of the foregoing, (iii) other than accounts receivable in the ordinary course of business, make any loans, advances (other than routine advances to employees of the Company and its Subsidiaries in the ordinary course of business) or capital contributions to, or investment in, any other Person, other than the Company or any of its direct or indirect wholly owned Subsidiaries, or (iv) other than in the ordinary course of business, enter into any hedging agreement or other financial agreement or arrangement designed to protect the Company or its Subsidiaries against fluctuations in commodities prices or exchange rates;

(g) neither it nor any of its Subsidiaries shall make any capital expenditures or other expenditures with respect to property, plant or equipment in excess of \$200,000 in the aggregate for the Company and its Subsidiaries, taken as a whole, other than as set forth in the Company s budget for capital expenditures previously made available to Parent;

(h) neither it nor any of its Subsidiaries shall make any material changes in accounting methods, principles or practices, except insofar as may have been required by a change in GAAP or applicable Law or, except as so required, change any assumption underlying, or method of calculating, any bad debt, contingency or other reserve;

(i) neither it nor any of its Subsidiaries shall, except in the ordinary course of business consistent with past practice, enter into, renew, modify, amend, terminate, waive, delay the exercise of, release or assign any material rights or claims under, any Company Material Contract or Company Lease; <u>provided</u>, that, nothing in this Agreement permits the Company or any of its Subsidiaries to (i) enter into any Contract of the type specified in Sections 5.5(a)(iii)(B), 5.5(a)(vii) or 5.5(a)(viii) to the extent such Contract would survive after the Effective Time or modify or amend in a manner adverse to the Company or any of its Subsidiaries any existing Contract of the type specified in Section 7.2(a), enter into, renew, modify, amend, terminate, waive, delay the exercise of, or release or assign any material rights or claims under, any confidentiality, standstill or similar agreement to which the Company or any of its Subsidiaries is bound by or subject;

(j) neither it nor any of its Subsidiaries shall, except as required to comply with applicable Law, any express provision of this Agreement, or agreements, plans or arrangements existing on the date of this Agreement and disclosed on Section 5.9(a)-1 of the Company Disclosure Schedule, (i) take any action with respect to, adopt, enter into, terminate or amend any employment, severance, change in control, retirement, retention, welfare, incentive or similar agreement, arrangement or benefit plan for the benefit or welfare of any current or former director, officer, employee or consultant or any collective bargaining agreement, (ii) increase in any respect the compensation or fringe benefits of, or pay any bonus to, any director, officer, employee or consultant, (iii) amend or accelerate the payment, right to payment or vesting of any compensation or benefits, including any outstanding options or restricted stock awards, (iv) pay any benefit not provided for as of the date of this Agreement under any Company Benefit Plan, (v) grant any awards under any bonus, incentive, performance or other compensation plan or arrangement or benefit plan, including

the grant of stock options, stock appreciation rights, stock based or stock related awards, performance units or restricted stock, or the removal of existing restrictions in any benefit plans or agreements or awards made thereunder or (vi) take any action to fund or in any other way secure the payment of compensation or benefits under any Company Benefit Plan;

#### Table of Contents

(k) neither it nor any of its Subsidiaries shall initiate, settle or compromise any litigation, claim, grievance, charge or proceeding involving any Intellectual Property or any matters reported as Legal Proceedings in any Company SEC Reports, or any other material litigation, claim, grievance, charge or proceeding (other than in connection with the enforcement of the Company s rights under this Agreement);

(1) neither it nor any of its Subsidiaries shall make or rescind any material Tax election, amend any material Tax Return or permit any insurance policy naming it as a beneficiary or loss-payable payee to be cancelled or terminated, in each case except in a manner consistent with past practice or as required by applicable Law;

(m) neither it nor any of its Subsidiaries shall authorize any of, or commit, resolve or agree, in writing or otherwise, to take, any of the foregoing actions; and

(n) it will and will cause its Subsidiaries to: (i) timely file all material Tax Returns required to be filed by the Company or such Subsidiary, as the case may be, and, except as required by applicable Law, prepare such Tax Returns in a manner consistent with past practice, and (ii) timely pay or accrue all material Taxes due and payable by the Company and such Subsidiary, respectively, except with respect to matters contested in good faith;

provided, however, that nothing contained in this Agreement will give to Parent, directly or indirectly, rights to control or direct the operations of the Company prior to Closing. Prior to Closing, the Company shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision of its and its Subsidiaries operations.

#### 7.2 No Solicitation.

(a) The Company shall not, nor shall it permit or authorize any of its Subsidiaries or any officer, director, employee, accountant, counsel, financial advisor, agent or other representative of the Company or any of its Subsidiaries (collectively, the <u>Company Representatives</u>) to, on its or any of its Subsidiaries behalf, directly or indirectly, (i) solicit, initiate, facilitate, respond to or encourage, including by way of furnishing non-public information, any inquiries regarding or relating to, or the submission of, any Takeover Proposal (as defined below), (ii) participate in any discussions or negotiations, furnish to any Person any information or data relating to the Company or its Subsidiaries or take any other action, in each such case regarding or to facilitate the making of any proposal that constitutes, or may reasonably be expected to lead to, any Takeover Proposal, (iii) enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or other similar agreement or commitment with respect to any Takeover Proposal (an <u>Alternative Acquisition Agreement</u>) or agree to, approve, endorse or resolve to recommend or approve any Takeover Proposal, except in each case as otherwise specifically provided in <u>Section 7.2(b)</u>, (iv) grant any waiver or release under any standstill or similar agreement by any Third Party who has made a Takeover Proposal, (v) take any action to exempt any Third Party from the restrictions on

business combinations contained in Section 203 of Delaware Law or otherwise cause such restrictions not to apply, or (vi) authorize or direct any Company Representative to take any such action; provided, however, that nothing contained in this Section 7.2(a) or any other provision of this Agreement prohibits the Company or the Company Board from (A) taking and disclosing to the Company s stockholders a position required by Rules 14d-9 and 14e-2 or Item 1012(a) of Regulation M-A promulgated under the Exchange Act, (B) making such disclosure to the Company s stockholders as, in the good faith judgment of the Company Board, after receiving advice from its outside counsel, is required under applicable Law in order to comply with its fiduciary duties, or (C) notifying any Third Party solely of the existence of, and restrictions under, the provisions of this Section 7.2, provided that the Company may not, except as permitted by Section 7.2(b), withdraw or modify, or propose to withdraw or modify, its approval or recommendation of this Agreement or the transactions contemplated hereby, including the Merger, or approve or

recommend, or propose to approve or recommend, any Takeover Proposal, or enter into any Alternative Acquisition Agreement. Upon execution of this Agreement, the Company shall, and it shall cause the Company Representatives and its Subsidiaries to, immediately terminate any existing activities, discussions, solicitations or negotiations with any Third Party conducted previously with respect to any Takeover Proposal. Notwithstanding any of the foregoing restrictions set forth in <u>Section 7.1</u> or this <u>Section 7.2(a)</u>, nothing in this Agreement prevents the Company or the Company Board from furnishing (or causing to be furnished), prior to, but not after, the time the vote is taken with respect to adoption of this Agreement and approval of the Merger at the Company Meeting, information concerning its business, properties or assets, which

A-33

information is not of greater scope, area or detail than was provided to Parent, to any Third Party pursuant to a confidentiality agreement with terms and conditions substantially similar to those of the Confidentiality Agreement, and may negotiate and participate in discussions and negotiations with such Third Party who has made a bona fide, written Takeover Proposal, but only if: (w) such Takeover Proposal was made after the date of this Agreement (it being understood that such a Takeover Proposal made after the date of this Agreement by a Third Party who has made a Takeover Proposal prior to the date of this Agreement is considered a new Takeover Proposal made after the date of this Agreement); (x) none of the Company, its Subsidiaries and the Company Representatives has solicited, initiated, or knowingly facilitated or encouraged any Takeover Proposal, or otherwise directly or indirectly violated this Section 7.2 (other than unintentional breaches that (1) have not directly or indirectly resulted in the making of such Takeover Proposal and (2) otherwise have had only an immaterial impact on Parent s rights under this Section 7.2); (y) such Third Party has submitted a Takeover Proposal that the Company Board has determined (after consulting with outside legal counsel) either (i) constitutes a Superior Proposal (as defined below) or (ii) is more favorable to the Company s stockholders from a financial point of view than the Merger and is reasonably likely to lead to a Superior Proposal; and (z) the Company Board determines in good faith, after receiving advice from its outside counsel, that such action is required to discharge the Company Board s fiduciary duties to the Company s stockholders under applicable Law. The Company shall not release or permit the release of any Third Party from, or waive or permit the waiver of any provision of, any confidentiality, standstill or similar agreement to which the Company is a party or under which the Company has any rights. The Company shall promptly (and in any event within one (1) Business Day) notify Parent telephonically and in writing of the existence of any proposal, discussion, negotiation or inquiry received by the Company that is or could reasonably be expected to constitute or lead to a Takeover Proposal, and the Company shall promptly communicate in writing to Parent the terms and conditions of any such proposal, discussion, negotiation or inquiry which it may receive, and provide a copy of any written proposal and the identity of the Third Party making the same. The Company shall inform Parent within twenty-four (24) hours after any change to the material terms of any such Takeover Proposal. Within twenty-four (24) hours after any determination by the Company Board that a Takeover Proposal constitutes a Superior Proposal, the Company shall deliver to Parent and Merger Sub a written notice advising them of such determination, specifying the terms and conditions of such Superior Proposal and the identity of the Third Party making such Superior Proposal, and providing Parent and Merger Sub with a copy of the Superior Proposal.

(b) Neither the Company Board nor any committee thereof shall (i) withdraw or modify, or propose to withdraw or modify, in a manner adverse to Parent or Merger Sub, the Company Board Recommendation, (ii) approve or recommend, or propose to approve or recommend, any Takeover Proposal or (iii) enter into any Alternative Acquisition Agreement (other than a confidentiality agreement expressly permitted by and in accordance with Section 7.2(a)). Notwithstanding the foregoing, prior to, but not after, the time the vote is taken with respect to the adoption of this Agreement and approval of the Merger at the Company Meeting, the Company Board may make a change in the Company Board Recommendation in a manner adverse to Parent or Merger Sub (including, for such purpose, a withdrawal of such Company Board Recommendation) (a <u>Change in Company Recommendation</u>) and/or approve or recommend a Superior Proposal (and, in connection therewith, take such action as shall be necessary to exempt such Superior Proposal from any Takeover Statute and the Company Rights Agreement), and the Company may enter into an Alternative Acquisition Agreement with respect to a Superior Proposal in connection with the termination of this Agreement, in each case if (A) the Company has received a Superior Proposal which is pending at the time the Company determines to take such action, (B) the Company Board has determined in good faith, after receiving advice from its outside counsel, that such action is required to discharge the Company Board s fiduciary duties to the Company s stockholders under applicable Law and (C) at least three (3) Business Days have passed following Parent s receipt of an Adverse Recommendation Notice (as defined below), and Parent does not make an offer within such three (3) Business Day period that is at least as favorable to the Company s stockholders as the Superior Proposal, as concluded by the Company Board in its good faith judgment, after consultation with its financial advisors and receiving advice from its outside counsel (it being agreed that the Company Board shall convene a meeting to consider any such offer by Parent promptly following receipt of such offer and that the Company Board

shall not withhold, withdraw or modify the Company Board Recommendation until the earlier of the receipt of Parent s revised offer or three (3) Business Days after receipt by Parent of the Adverse Recommendation Notice).

(c) For purposes of this Agreement:

(i) The term <u>Adverse Recommendation Notice</u> means a written notice from the Company advising Parent that the Company Board has received a Superior Proposal which it intends to accept or recommend or advising Parent that it intends to make a Change in Company Recommendation, specifying the material terms and conditions of such Superior Proposal and the other information required by <u>Section 7.2(a)</u>; provided, however, any material amendment to the financial terms or other material terms of such Superior Proposal shall require a new Adverse Recommendation Notice and a new three (3) Business Day period.

(ii) The term <u>Competing Transaction</u> means any transaction, other than the transactions contemplated by this Agreement, to acquire beneficial ownership (as defined under Rule 13(d) of the Exchange Act) of (A) assets that constitute fifteen percent (15%) or more of the consolidated revenues, net income or assets of the Company and its Subsidiaries or (B) fifteen percent (15%) or more (in number or voting power) of any class of equity securities or other capital stock of the Company or any of its Subsidiaries, in any such case pursuant to any transaction or series of transactions, including (I) a merger, consolidation, share exchange, or other business combination (including any so-called merger-of-equals and whether or not the Company is the entity surviving any such transaction) involving the Company or any of its Subsidiaries, (II) a sale, issuance, exchange, transfer or other disposition of shares of capital stock of the Company or any of its Subsidiaries or (IV) a tender offer, exchange offer or similar transaction with respect to either the Company or any of its Subsidiaries, including any single or multi-step transaction or series of related transactions, which is structured to permit such Third Party or another Third Party to acquire beneficial ownership of assets that constitute fifteen percent (15%) or more of the consolidated revenues, net income or assets of the Company or any of its Subsidiaries, or fifteen percent (15%) or more of the consolidated revenues, net income or assets of the Company or any of its Subsidiaries, or fifteen percent (15%) or more of the equity interest in either the Company or any of its Subsidiaries, or more of the consolidated revenues, net income or assets of the Company and its Subsidiaries, or fifteen percent (15%) or more of the equity interest in either the Company or any of its Subsidiaries.

(iii) The term <u>Superior Propos</u>al means an unsolicited written proposal or offer (whether a Takeover Proposal or otherwise) by a Third Party to acquire (whether by way of merger, acquisition or otherwise), directly or indirectly, greater than fifty percent (50%) of the shares of Company Common Stock then outstanding (or the effect of which would be that the stockholders of the Company beneficially own less than fifty percent (50%) of the voting power of the combined or ongoing entity), or to acquire all or substantially all of the assets of the Company, and (A) otherwise on terms which the Company Board determines in good faith (after consultation with its financial advisors), and taking into account all relevant terms and conditions of the proposal or offer that it deems relevant (including all legal, financial, regulatory and other aspects, including any financing condition and time to consummation), to be more favorable to the Company s stockholders from a financial point of view than the Merger, and (B) which, in the good faith reasonable judgment of the Company Board, is reasonably likely to be consummated.

(iv) The term <u>Takeover Propos</u>al means any inquiry, proposal, offer or indication of interest (including any inquiry, proposal, offer or indication of interest to its stockholders), whether in writing or otherwise, from a Third Party that constitutes, or could reasonably be expected to lead to, a Competing Transaction.

(v) The term <u>Third Party</u> means any Person or group other than Parent, Merger Sub or any Affiliate of Parent.

7.3 <u>Proxy Statement</u>. As promptly as practicable after the execution of this Agreement, the Company, in cooperation with Parent, shall prepare and file with the SEC the Proxy Statement. Prior to filing the Proxy Statement or any other filing with the SEC or any other Governmental Entity, the Company shall provide Parent with reasonable opportunity to review and comment on each such filing in advance of its filing with the SEC or other applicable Governmental Entity and the Company shall consider and act in good faith with respect to the incorporation of any changes in such filings reasonably proposed by Parent. The Company shall respond to any comments of the SEC or its staff and shall

cause the Proxy Statement to be mailed to its stockholders at the earliest practicable time after the resolution of any such comments. The Company shall notify Parent promptly upon the receipt of any comments from the SEC or its staff or any other government officials and of any request by the

SEC or its staff or any other government officials for amendments or supplements to the Proxy Statement and shall supply Parent with copies of all correspondence between the Company or any of its representatives, on the one hand, and the SEC, its staff or any other government officials, on the other hand, with respect to the Proxy Statement. The Company shall use its commercially reasonable efforts to cause all documents that it is responsible for filing with the SEC or other regulatory authorities under this <u>Section 7.3</u> to comply in all material respects with all applicable requirements of Law and the rules and regulations promulgated thereunder. If at any time prior to the Effective Time, any event occurs, or any information relating to the Company, Parent, Merger Sub or any of their respective Affiliates, officers or directors should be discovered by the Company, Parent or Merger Sub, which is required to be set forth in an amendment or supplement to the Proxy Statement or any other filing with any Governmental Entity, so that the Proxy Statement or such other filing shall not contain any untrue statement of a material fact or omit to state any material fact required to be stated in the Proxy Statement or necessary in order to make the statements in the Proxy Statement, in light of the circumstances under which they are made, not misleading, the party that discovers such information shall promptly notify the other parties, and cooperate in filing with the SEC or its staff or any other government.

7.4 Company Meeting. The Company, acting through the Company Board, shall take all actions in accordance with applicable Law, its certificate of incorporation and by-laws and SEC rules to promptly and duly call, give notice of, convene and hold as promptly as practicable a meeting of the holders of shares of Company Common Stock (the <u>Company Meeting</u>) for the purpose of considering and voting upon the approval of this Agreement and the Merger by the holders of a majority of the outstanding shares of Company Common Stock entitled to vote on such matters. Unless this Agreement is terminated in accordance with Article IX, the obligation of the Company to convene and hold the Company Meeting shall not be limited or otherwise affected by a Change in Company Recommendation or by the commencement, public proposal, public disclosure or communication to the Company of any Takeover Proposal. Except as otherwise provided in Section 7.2, (i) the Company Board shall recommend to the stockholders of the Company the adoption of this Agreement and approval of the Merger and include such recommendation in the Proxy Statement and (ii) the Company Board shall not withhold, withdraw, qualify or modify, or publicly propose or resolve to withhold, withdraw, qualify or modify, in a manner adverse to Parent, the recommendation of the Company Board that the Company s stockholders vote in favor of the adoption of this Agreement and approval of the Merger. Unless such recommendation shall have been withdrawn or modified in accordance with Section 7.2 (but without affecting in any manner the Company s obligations pursuant to Section 7.3), the Company shall use commercially reasonable efforts to solicit from its stockholders proxies in favor of the approval of this Agreement and the Merger and shall use commercially reasonable efforts to take all other action necessary or advisable to secure the vote or consent of the stockholders of the Company required by Delaware Law. Notwithstanding anything to the contrary contained in this Agreement, the Company, after consultation with Parent, may adjourn the Company Meeting, but only to the extent necessary to ensure that any required supplement or amendment to the Proxy Statement is provided to the Company s stockholders or, if as of the time for which the Company Meeting is originally scheduled (as set forth in the Proxy Statement) there are insufficient shares of Company Common Stock represented (either in person or by proxy) to constitute a quorum necessary to conduct the business of the Company Meeting.

### 7.5 Filings; Other Actions; Notification.

(a) Subject to the terms of this Agreement, including <u>Section 7.2</u> and <u>Section 7.5(b)</u>, the Company and Parent shall each use commercially reasonable efforts to (i) take or omit to take, or cause to be taken or omitted, all actions, and do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective the transactions contemplated hereby as promptly as practicable, (ii) as promptly as practicable, obtain from any Governmental Entity or any other Third Party any consents, licenses, permits, waivers, approvals, authorizations, or orders required to be obtained or made by the Company or Parent or any of their respective Subsidiaries, or otherwise reasonably requested by Parent, in connection with the authorization, execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, (iii) as

promptly as practicable, and on a mutually agreed date, make all necessary filings, and thereafter make any other required submissions, with respect to this Agreement and the Merger required under (A) the Exchange Act and any other applicable federal or state securities Laws, (B) any Antitrust Law (as defined below) and any related governmental request thereunder, and (C) any other applicable

Law, and (iv) execute or deliver any additional instruments necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement. The Company and Parent shall consult and cooperate with each other, and consider in good faith the views of one another, in connection with the obtaining of all such consents, licenses, permits, waivers, approvals, authorizations or orders, including (i) keeping the other apprised of the status of matters relating to the completion of the transactions contemplated hereby, (ii) providing copies of written notices or other communications received by such party or any of its respective Subsidiaries with respect to the transactions contemplated hereby, (iii) subject to applicable Laws relating to the sharing of information and with the right to withhold confidential information, providing copies of any proposed filings to be made with, or written materials submitted to, any Third Party and/or any Governmental Entity in connection with the transactions contemplated hereby (including any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any party hereto) and (iv) if requested, accepting reasonable additions, deletions or changes suggested in connection therewith. The Company and Parent shall each use its commercially reasonable efforts to furnish to each other all information required for any application or other filing to be made pursuant to the rules and regulations of any applicable Law (including all information required to be included in the Proxy Statement) in connection with the transactions contemplated by this Agreement. In connection with the foregoing, each party shall (i) promptly notify the other party in writing of any communication received by that party or its Affiliates from any Governmental Entity, and subject to applicable Laws, provide the other party with a copy of any such written communication (or an oral or written summary of any oral communication), and (ii) not participate in any substantive meeting or discussion with any Governmental Entity in respect of any filing, investigation or inquiry concerning the transactions contemplated by this Agreement unless, where practicable, it consults with the other party in advance, and to the extent permitted by such Governmental Entity and where practicable, gives the other party the opportunity to attend and participate thereat. For the avoidance of doubt, Parent and the Company agree that nothing contained in this <u>Section 7.5(a)</u> modifies or affects their respective rights and responsibilities under <u>Section 7.5(b)</u>.

(b) Subject to the terms of this Agreement, each of Parent and the Company shall, and shall cause each of their respective Subsidiaries to, cooperate and use commercially reasonable efforts to obtain any government clearances or approvals required for Closing under the HSR Act, the Sherman Act, as amended, the Clayton Act, as amended, the Federal Trade Commission Act, as amended, and any other federal, state or foreign Law, regulation or decree designed to prohibit, restrict or regulate actions for the purpose or effect of monopolization or restraint of trade, as the case may be (collectively <u>Antitrust Laws</u>), to respond to requests of any Governmental Entity for information under any Antitrust Law, and to contest and resist any action, including any legislative, administrative or judicial action, and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order (whether temporary, preliminary or permanent) that restricts, prevents or prohibits the consummation of the Merger or any other transactions contemplated by this Agreement under any Antitrust Law. Notwithstanding the foregoing or anything to the contrary in this Agreement, neither Parent nor any of its Affiliates are required to (i) proffer to, or agree to, sell, license, lease, transfer or otherwise encumber (or consent to any sale, license, lease, transfer or other encumbrance or agreement to sell, license, lease, transfer or otherwise encumber by the Company), before or after the Effective Time, any assets, businesses or interest in any assets or businesses of Parent, the Company or any of their respective Affiliates, (ii) proffer to, or agree to, hold separate, before or after the Effective Time, any assets, businesses or interest in any assets or businesses of Parent, the Company or any of their respective Affiliates, (iii) agree to any other changes (including through a licensing arrangement) or restriction on, or other impairment of, Parent s ability to own or operate the operations of any such assets or businesses or Parent s ability to vote, transfer, receive dividends with respect to or otherwise exercise ownership rights with respect to the stock of the Surviving Corporation, or (iv) take any other action under this Section 7.5 if any applicable Governmental Entity seeks a preliminary injunction or restraining order to enjoin consummation of the Merger. The Company and Parent shall each request early termination of the waiting period with respect to the Merger under the HSR Act.

(c) During the period commencing on the date of this Agreement and ending at the Effective Time or such earlier date as this Agreement may be terminated in accordance with its terms, Parent shall give prompt notice to the Company,

and the Company shall give prompt notice to Parent, of (i) the occurrence, or failure to occur, of any event, which occurrence or failure to occur is reasonably likely to cause the conditions in <u>Section 8.2(a)</u> or <u>Section 8.3(a)</u>, as the case may be, not to be satisfied, (ii) any material failure of Parent and Merger Sub or the Company, as the case may be, or of any officer, director, employee or agent, to comply with or satisfy any covenant,

condition or agreement to be complied with or satisfied by it under this Agreement, (iii) any actions, suits, claims, investigations or proceedings commenced or threatened in writing against, relating to or involving or otherwise affecting such party or any of its Subsidiaries that relate to the consummation of the Merger, or (iv) any offers received by the Company and its Subsidiaries to settle or compromise any litigation, claim, grievance, charge or proceeding involving Intellectual Property or any other material litigation, claim, grievance, charge or proceeding. Notwithstanding the above, the delivery of any notice pursuant to this <u>Section 7.5</u> does not limit or otherwise affect the remedies available hereunder to the party receiving such notice or the conditions to such party s obligation to consummate the Merger.

7.6 Access. Subject to applicable Law relating to the sharing of information, upon reasonable notice, and except as may otherwise be required by applicable Law, the Company shall (and shall cause its Subsidiaries to) afford Parent and its representatives reasonable access, during normal business hours throughout the period prior to the Effective Time, to its properties, books, contracts and records and, during such period, shall (and shall cause its Subsidiaries to) furnish promptly to Parent all information concerning its business, properties and personnel as may reasonably be requested, including information relating to Contracts with Governmental Entities, insurance, pending litigation or claims, employee and employment matters, and information regarding Company membership in standards organizations; provided, however, that no investigation pursuant to this Section 7.6 affects or modifies any representation or warranty made by the Company; and provided, further, that the foregoing do not require the Company (a) to permit any inspection, or to disclose any information, that in the reasonable judgment of the Company would result in the disclosure of any trade secrets of any Person or violate any of its obligations with respect to confidentiality if the Company shall have used its commercially reasonable efforts to obtain the consent of such Person to such inspection or disclosure or (b) to disclose any privileged information of the Company or any of its Subsidiaries. All requests for information made pursuant to this Section 7.6 must be directed to an executive officer of the Company or such Person as may be designated by such executive officer. All information that is made available pursuant to this <u>Section 7.6</u> is governed by the terms of the Confidentiality Agreement.

7.7 Notice of Certain Matters. Without limiting the Company s obligations under Section 7.5(c), each party shall notify the other in writing promptly after learning of any of the following: (i) any notice or other communication from any Third Party alleging that the consent of such Third Party is or may be required in connection with the Merger; (ii) any notice or other communication from any Governmental Entity in connection with the Merger; (iii) any action, suit, arbitration, mediation, proceeding, claim or investigation by or before any Governmental Entity or arbitrator initiated by or against it or any of its Subsidiaries, or known by it or any of its Subsidiaries to be threatened against it or any of its Subsidiaries or any of their respective directors, officers, employees or stockholders in their capacity as such, or of any oral or written correspondence from any Third Party asserting or implying a claim against it or with respect to any of its assets or properties (including Intellectual Property) that is, or is reasonably likely to be, material to it and its Subsidiaries, taken as a whole; (iv) any change, occurrence or event which is reasonably likely to cause any of the conditions to closing set forth in Article VIII not to be satisfied; (v) any claim, or any written inquiry by any Taxing authority, regarding a material deficiency to pay Taxes payable by the Company; or (vi) any event that occurs after the date of this Agreement that, had it occurred prior to the date of this Agreement, would have constituted an exception to the representation set forth in Section 5.6(e). Each party shall use commercially reasonable efforts to give to the other party prompt notice of any representation or warranty made by such party contained in this Agreement known to have become untrue or inaccurate in any material respect, or any known failure by such party to comply with or satisfy in any material respect any covenant, condition or agreement to be complied with or satisfied by it under this Agreement; provided, however, that no such notification shall affect the representations, warranties, covenants or agreements of the parties herein or affect the satisfaction or non-satisfaction of any conditions to the obligations of the parties under this Agreement or otherwise limit or affect the remedies available hereunder to Parent or the Company.

7.8 <u>Removal of Company Common Stock From Pink Sheets</u>. If immediately prior to the Effective Time, the Company Common Stock is quoted on the Pink Sheets, the Company shall give timely notice of removal of the Company Common Stock from quotation on the Pink Sheets to the Market Integrity Department of The Nasdaq Stock Market, and shall use its commercially reasonable efforts to cause the shares of Company Common Stock to cease to be quoted on the Pink Sheets effective as of the Effective Time and, if applicable, to report timely such event on Form 8-K.

#### Table of Contents

7.9 <u>Publicity</u>. The initial press release regarding the Merger will be a joint press release and thereafter the Company and Parent each shall consult with each other before issuing, and provide each other the opportunity to review, comment upon and concur with and use reasonable efforts to agree on, any press releases or other public announcements with respect to the Merger and the other transactions contemplated by this Agreement and prior to making any filings with any Third Party and/or any Governmental Entity with respect thereto, except as may be required by Law. Notwithstanding anything to the contrary contained in this <u>Section 7.9</u>, either party may respond to questions from stockholders or inquiries from financial analysts and media representatives in a manner that is consistent with then-existing public disclosures.

7.10 <u>Company and Parent Benefit Plans</u>. Prior to the Effective Time, if requested by Parent in writing, the Company shall take, in accordance with the requirements of applicable Law, all actions necessary to terminate any or all Company Benefit Plans (other than employment agreements, consulting agreements and the Company Benefit Plans set forth in Section 7.10 of the Company Disclosure Schedule) effective immediately prior to the Effective Time. Prior to the Effective Time, if requested by Parent in writing, to the extent permitted by applicable Law and the terms of the applicable plan or arrangement, the Company shall cause to be amended the Company Benefit Plans to the extent necessary to provide that no employees of Parent and its Subsidiaries shall commence to participate therein following the Effective Time unless the Surviving Corporation or such Subsidiary explicitly authorizes such participation.

7.11 Loans to Company Employees, Officers and Directors. Prior to the Effective Time, the Company and its Subsidiaries shall take all action necessary so that all loans (other than loans under the Company s 401(k) plan, travel advances, payroll advances and other advances made in the ordinary course of business, which in each case do not exceed \$1,000) by the Company or any of its Subsidiaries to any of their employees, officers or directors will no longer be outstanding as of the Effective Time.

#### 7.12 Indemnification; Directors and Officers Insurance.

(a) From the Effective Time through the sixth (6th) anniversary of the date on which the Effective Time occurs, Parent shall cause the Surviving Corporation (and any successor entity to the Surviving Corporation) to indemnify and hold harmless each person who is now, who has been prior to the date hereof, or who becomes prior to the Effective Time, a director or officer of the Company or any of its Subsidiaries (the <u>Indemnified Parties</u>), against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, including attorneys fees and disbursements (collectively, <u>Costs</u>), arising out of or incurred in connection with any claim, action, suit, proceeding, demand or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the Indemnified Party is or was an officer or director of the Company or any of its Subsidiaries, whether asserted or claimed prior to, at or after the Effective Time, to the fullest extent permitted under Delaware Law for officers and directors of Delaware corporations. Each Indemnified Party will be entitled to advancement of expenses incurred in the defense of any such claim, action, suit, proceeding or investigation from the Surviving Corporation (or successor entity) within ten (10) Business Days of receipt by Parent or the Surviving Corporation from the Indemnified Party of a request therefor; provided, however, that any Third Party to whom expenses are advanced provides an undertaking to repay such advances if it is ultimately determined that such Third Party is not entitled to indemnification.

(b) Subject to the next sentence, the Surviving Corporation (or successor entity) shall purchase, and Parent shall cause the Surviving Corporation (or successor entity) to purchase, at no expense to the Indemnified Parties, from the Effective Time, a six (6) year tail under the current policies of the directors and officers liability insurance maintained by the Company with respect to matters existing or occurring at or prior to the Effective Time (including the transactions contemplated by this Agreement), so long as the premium therefor would not be in excess of the percentage, as set forth in Section 7.12(b) of the Company Disclosure Schedule, of the annual premium paid by the Company in its most recent fiscal year, which premium is set forth in Section 7.12(b) of the Company Disclosure

Schedule (such percentage of the annual premium, the <u>Maximum Premium</u>). If the premium for such tail coverage exceeds the Maximum Premium, the Surviving Corporation shall obtain, and Parent shall cause the Surviving Corporation to obtain, the longest tail coverage as can be obtained for a premium not in excess of the Maximum Premium.

#### Table of Contents

(c) If Parent fails to comply with its obligations under this <u>Section 7.12</u>, and, in order to enforce an Indemnified Party s rights under this <u>Section 7.12</u>, an Indemnified Party commences a suit that results in a judgment against Parent that Parent breached its obligations under this <u>Section 7.12</u>, Parent shall pay to the Indemnified Party its reasonable costs and expenses (including reasonable attorneys fees and expenses) incurred in connection with such suit after delivery to Parent of reasonable documentation evidencing such costs and expenses.

(d) The provisions of this <u>Section 7.12</u> are intended to be in addition to the rights otherwise available to the current officers and directors of the Company by law, charter, statute, by-law or agreement, and operate for the benefit of, and are enforceable by, each of the Indemnified Parties, their heirs and their representatives in accordance with <u>Section 10.8</u>.

7.13 <u>Takeover Statute</u>. If any Takeover Statute is or may become applicable to the Merger or the other transactions contemplated by this Agreement, the Company and the Company Board shall grant such approvals and take such actions as are necessary so that such transactions may be consummated as promptly as practicable on the terms contemplated by this Agreement or by the Merger and otherwise act to eliminate the effects of such statute or regulation on such transactions.

7.14 <u>Section 16 Matters</u>. Prior to the Effective Time, the Company shall take all such steps as may be required (to the extent permitted under applicable Law) to cause any dispositions of Company Common Stock (including derivative securities with respect to Company Common Stock) resulting from the transactions contemplated by Article I by each individual who is subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to the Company to be exempt under Rule 16b-3 promulgated under the Exchange Act.

7.15 Litigation Insurance Policy. The Company agrees to cooperate as reasonably requested by Parent with respect to Parent s assessment as to whether it is desirable to obtain additional insurance coverage relating to certain litigation matters set forth in Section 5.8 of the Company Disclosure Schedule (the Litigation Insurance Policy ), including providing all reasonably requested information to agents and representatives of Parent. The Company acknowledges and agrees that such cooperation will also include, if requested by Parent not less than ten (10) Business Days prior to Closing, the Company using its reasonable best efforts to apply for, and purchase immediately prior to or concurrent with Closing, the Litigation Insurance Policy, such that the purchase will be reflected on the financial statements of the Company prior to Closing. The Company and its Subsidiaries shall be the named insureds on the Litigation Insurance Policy authorized by Parent pursuant to this Section 7.15 in the event that the Agreement is subsequently terminated by Parent pursuant to <u>Section 9.2</u> or <u>Section 9.4</u>; provided, however, the Company agrees to terminate the Litigation Insurance Policy at the request of Parent and return to Parent any premium reimbursement from the insurer.

#### 7.16 Tax Returns.

(a) As soon as practicable after the date hereof, the Company will provide to Parent the Company s apportionment schedule for 2005 and similar available apportionment information for such other years as Parent may reasonably request (the <u>Apportionment Information</u>). Any Apportionment Information for the years ending December 31, 2002 and thereafter will be certified by the Chief Financial Officer of the Company to the effect that, to the best of his knowledge, the Apportionment Information is true, correct and complete. Any Apportionment Information for any other year, to the extent available, will be certified by the Chief Financial Officer of the Company to the effect that, to the best of his knowledge, it is true, correct, and, to the extent available, complete. The Apportionment Information for each relevant year will (i) for years ending December 31, 2002 and thereafter indicate all U.S. jurisdictions into

which the Company and its Subsidiaries have sales or in which they own property or employ individuals, (ii) for other years, to the extent available, indicate all U.S. jurisdictions into which the Company and its Subsidiaries have sales or in which they own property or employ individuals, (iii) for any year, to the extent available, indicate any non-U.S. jurisdiction in which the Company and its Subsidiaries have sales or in which they own property or employ individuals, (iii) for any year, to the extent available, indicate any non-U.S. jurisdiction in which the Company and its Subsidiaries have sales or in which they own property or employ individuals, and (iv) indicate any jurisdiction in which the Company or any Subsidiary have filed income Tax Returns. The Company will cooperate with Parent as necessary, including by providing access to appropriate personnel to discuss the Apportionment Information and any additional information reasonably requested by Parent, in order to assist Parent s review of the Apportionment Information. The Company

A-40

will also supplement the provision of any Apportionment Information provided to Parent to the extent that, without such supplement, the Chief Financial Officer could not provide the required certificate under Section 7.16(b). Any provision of supplemental Apportionment Information by the Company to Parent will be via email to the individuals designated for such purpose in Section 10.6. Within twenty (20) days of receipt of the Chief Financial Officer s certificate with respect to the initial Apportionment Information, and five (5) days of the receipt of any supplemental Apportionment Information, Parent will notify the Company in writing of any income Tax Return that it reasonably believes is delinquent but not disclosed on Section 5.12 of the Company Disclosure Schedule (each such written notice is referred to as a <u>Section 7.16(a) Notice</u>) and, if reasonably requested by the Company, Parent will provide in reasonable detail the legal basis for that belief for each such Tax Return. If the Company receives a Section 7.16(a) Notice within thirty (30) days of the Outside Date (as determined without regard to this sentence), and the failure to file each delinquent Tax Return set forth in a Section 7.16(a) Notice received within thirty (30) days of the Outside Date is the sole condition under this Agreement to Closing that has not been satisfied as of the Outside Date (as determined without regard to this sentence), the Outside Date will be extended such that the Outside Date will be thirty (30) days after the date the final Section 7.16(a) Notice is received by the Company. The parties acknowledge and agree that the filing of each delinquent Tax Return set forth in a Section 7.16(a) Notice by the Company or one of its Subsidiaries, as applicable, will be sufficient to satisfy the closing condition set forth in clause (B) of Section 8.2(m)(i). For purposes of this Section 7.16, the Company and Parent agree that a Tax Return will not be considered to be delinquent if (i) the relevant statute of limitations (after giving effect to any tolling, waiver, extension, or mitigation thereof ) has expired; (ii) the Company or its Subsidiary has substantial authority within the meaning of Code section 6662(d)(2)(B)(i) (or a similar standard which avoids the imposition of penalties under applicable Law) not to file the Tax Return; or (iii) the parties otherwise agree that the Tax Return need not be filed.

(b) At least five (5) days prior to Closing, the Company will provide to Parent a certificate signed on behalf of the Company by the Chief Financial Officer of the Company to the effect that, to the best of his knowledge, (1) the Apportionment Information (as supplemented) provided pursuant to Section 7.16(a) for years ending on December 31, 2002 and thereafter remains true, correct and complete and the Apportionment Information (as supplemented) provided pursuant to Section 7.16(a) for any other year remains true, correct, and, to the extent available, complete and (2) the conditions set forth in Section 8.2(m) have been satisfied.

### ARTICLE VIII

### **CONDITIONS**

8.1 <u>Conditions to Each Party</u> <u>s</u> <u>Obligation to Effect the Merger</u>. The respective obligation of each party to effect the Merger is subject to the satisfaction or waiver at or prior to the Effective Time of each of the following conditions:

(a) <u>Stockholder Approval</u>. This Agreement must have been adopted and the Merger approved at the Company Meeting, at which a quorum is present, by the affirmative vote of the holders of at least a majority of the outstanding shares of Company Common Stock on the record date for the Company Meeting entitled to vote thereon.

(b) <u>Antitrust Laws</u>. The waiting period applicable to the consummation of the Merger under the HSR Act must have expired or been terminated, any filings required to be made under any other applicable Antitrust Laws have been made, and any approvals required to be obtained under any other applicable Antitrust Laws must have been obtained or any required waiting periods must have expired or been terminated.

(c) <u>No Injunctions</u>. No Governmental Entity of competent jurisdiction has enacted, issued, promulgated, enforced or entered any order, executive order, stay, decree, judgment or injunction (preliminary or permanent) or statute, rule or regulation (each, an <u>Order</u>) which is in effect and which has the effect of making the Merger illegal or otherwise prohibiting consummation of the Merger or the other transactions contemplated by this Agreement.

#### Table of Contents

8.2 <u>Conditions to Obligations of Parent and Merger Sub</u>. The obligations of Parent and Merger Sub to effect the Merger are also subject to the satisfaction or waiver by Parent at or prior to the Effective Time of the following conditions:

(a) <u>Representations and Warranties</u>. The representations and warranties of the Company set forth in this Agreement must be true and correct (without giving effect to any limitation as to materiality or Company Material Adverse Effect or any similar limitation set forth therein) as of the date of this Agreement and as of the Closing Date as though made on and as of such date and time (except to the extent that any such representation and warranty speaks of an earlier date, in which case such representation and warranty must be true and correct as of such earlier date), except for such inaccuracies (other than with respect to <u>Sections 5.2</u>, 5.3 and 5.22 which must be true and correct in all material respects) that, individually or in the aggregate, have not had, and could not reasonably be expected to have, a Company Material Adverse Effect (it being understood that, for purposes of determining the accuracy of such representations and warranties, any update of or modification to the Company Disclosure Schedule purported to have been made after the execution of this Agreement will be disregarded). Parent must have received at the Closing a certificate signed on behalf of the Company by the Chief Executive Officer and the Chief Financial Officer of the Company to the effect that each such executive officer has read this <u>Section 8.2(a)</u> and the conditions set forth in this <u>Section 8.2(a)</u> have been satisfied.

(b) <u>Performance of Obligations of the Company</u>. The Company must have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Parent must have received a certificate signed on behalf of the Company by the Chief Executive Officer and the Chief Financial Officer of the Company to such effect.

(c) <u>No Restraints</u>. No action, investigation, proceeding or litigation may be instituted, commenced, pending or threatened:

(i) in which a Governmental Entity is (A) challenging or seeking to restrain or prohibit the consummation of the Merger or any of the other transactions contemplated by this Agreement or (B) seeking to (x) prohibit or impair Parent s ability to own or operate any of the businesses and assets of the Company or its Subsidiaries from and after the Effective Time or any of the businesses or assets of Parent or its Subsidiaries (including through any divestiture, licensing, lease or hold separate arrangement) or (y) prohibit or limit in any material respect Parent s ability to vote, transfer, receive dividends with respect to or otherwise exercise ownership rights with respect to the stock of the Surviving Corporation (any such restraint, prohibition, impairment, limitation or result described in clause (A) and (B) above, a <u>Burdensome Condition</u>); or

(ii) which may reasonably be expected to result in the imposition of (A) criminal sanctions on the Company or any of its Subsidiaries or (B) material penalties or fines to a Governmental Entity, or restitution to a Third Party, in each case, resulting from the (x) conviction (including as a result of the entry of a guilty plea, a consent judgment or a plea of nolo contendere) of the Company or any of its Subsidiaries of a crime or (y) settlement with a Governmental Entity for the purpose of closing an investigation, being imposed on Parent or the Surviving Corporation or any of their respective Affiliates.

No Governmental Entity of competent jurisdiction may have enacted, issued, promulgated, enforced or entered any Order imposing a Burdensome Condition.

(d) <u>Governmental Approvals</u>. Other than the filing pursuant to <u>Section 1.2</u>, all other authorizations, consents, orders or approvals of, or declarations, notices or filings with, or expirations of waiting periods imposed by, any Governmental Entity in connection with the Merger and the consummation of the other transactions contemplated

hereby by the Company, Parent and Merger Sub must have been made or obtained (as the case may be) except those that the failure to make or obtain could not reasonably be expected to have a Company Material Adverse Effect or a material adverse effect on Parent or to provide a reasonable basis to conclude that the parties hereto or any of their Affiliates would be subject to risk of criminal sanctions or any of their representatives would be subject to the risk of criminal or material civil sanctions.

(e) Required Consents.

#### Table of Contents

(i) The consents, approvals or waivers indicated on Section  $\underline{8.2(e)(i)}$  of the Company Disclosure Schedule must have been obtained or made, as applicable.

(ii) Those contracts on <u>Section 8.2(e)(ii)</u> of the Company Disclosure Schedule must remain in full force and effect as of the Effective Time.

(f) <u>No Company Material Adverse Effect</u>. Since the date of this Agreement, there must not have occurred any change, event, circumstance or development that has had, or could reasonably be expected to have, a Company Material Adverse Effect.

(g) <u>Sarbanes-Oxley Certifications</u>. Neither the principal executive officer nor the principal financial officer of the Company will have failed to provide the necessary certifications in the form required under Section 302 and Section 906 of the Sarbanes-Oxley Act on any Company SEC Reports.

(h) <u>Company SEC Reports; Restatements</u>. The Company will have filed all Company SEC Reports required to be filed by it with the SEC on or prior to the Closing Date. To the extent that the Company has publicly announced or otherwise disclosed to Parent that information contained in any Company SEC Report filed with the SEC prior to the date of this Agreement will be amended or superseded by a later filed Company SEC Report (a <u>Restatement</u>), the Company will have completed and filed such Restatement with respect to any and all such Company SEC Reports on or prior to the Closing Date. Any Restatement of a Company SEC Report filed with the SEC after the date of this Agreement will not have contained any information materially and adversely different from information contained in the Company SEC Report amended or superseded by such Restatement.

(i) <u>Intellectual Property Litigation</u>. Since the date of this Agreement, no Person may have instituted or threatened any Action that challenges the validity and ownership of the Owned Intellectual Property, other than such Actions the outcome of which is not reasonably expected to result in a Company Material Adverse Effect.

(j) <u>SEC Investigation</u>. The Staff of the SEC shall not have recommended that any charges or enforcement action be brought against the Company, its Subsidiaries or any of its current officers or directors in connection with the SEC Investigation, and the SEC shall not have authorized any such recommendation or issued a Wells Notice to the Company or any of its current officers or directors in connection with the SEC Investigation. There shall not be pending any action, suit, proceeding, hearing, arbitration, or investigation, in each case by any Governmental Entity (distinct from the SEC Investigation), of or against any director, officer or Key Employee of the Company or any of its Subsidiaries relating to the matters in the SEC Investigation. The term <u>SEC Investigation</u> means the matters relating to or arising under the formal investigation commenced by the SEC in December 2005 in connection with the Company is accounting review of certain customer transactions, which investigation was publicly disclosed by the Company in its Current Report on Form 8-K filed on March 2, 2006.

(k) <u>Company Net Cash</u>. The Company Net Cash must equal or exceed the Target Net Cash Amount as of the Closing Date, giving effect to any payments (other than payments of Expenses) required to be made by the Company or its Subsidiaries on or prior to the Effective Time in accordance with past payment practice. The term <u>Target Net Cash Amount</u> means \$15 million, if the Closing Date occurs on or prior to July 31, 2007, and \$13 million, if the Closing Date occurs after July 31, 2007, reduced in each case by the aggregate amount of all reasonable Expenses (whether incurred before or after the date hereof) paid after the date hereof.

(l) Key Employees.

(i) Each of the employees of the Company or one of its Subsidiaries set forth on Section 8.2(1) of the Company Disclosure Schedule as Group 1 Employees as of the date hereof have entered into a retention agreement with Parent, and seventy five percent (75%) of such Group 1 Employees (in accordance with the parameters set forth on Section 8.2(1)(i) of the Company Disclosure Schedule) will still be on-the-job and performing their usual and customary duties for the Company or one of its Subsidiaries immediately before the Effective Time.

(ii) At least eighty percent (80%) of the employees of the Company or one of its Subsidiaries set forth on Section 8.2(1) of the Company Disclosure Schedule as Group 2 Employees will still be on-the-job and performing their usual and customary duties for the Company or one of its Subsidiaries immediately before the Effective Time.

A-43

#### (m) Tax Returns.

(i) The Company has filed each U.S. federal income Tax Return for the affiliated group of corporations for which the Company is the common parent for the years ending on or before the date of this Agreement that is delinquent as of the date of this Agreement and provided Parent with a copy of such Tax Returns. Either the Company or one of its Subsidiaries, as applicable, has filed (A) each other income Tax Return which has been disclosed on Section 5.12 of the Company Disclosure Schedule as being delinquent, and (B) each other income Tax Return, if any, that Parent and the Company agree is delinquent after the date of this Agreement pursuant to <u>Section 7.16</u>, and the Company has provided Parent with a copy of such Tax Returns. Parent will have received the certificate specified in <u>Section 7.16(b)</u>.

(ii) For purposes of clauses (A) and (B) of the second sentence of <u>Section 8.2(m)(i)</u>, the Profit Tax Returns for Terayon Hong Kong Limited will not be considered to be delinquent as of the date of this Agreement.

8.3 <u>Conditions to Obligation of the Company</u>. The obligation of the Company to effect the Merger is also subject to the satisfaction or waiver by the Company at or prior to the Effective Time of the following conditions:

(a) <u>Representations and Warranties</u>. The representations and warranties of Parent set forth in this Agreement must be true and correct (without giving effect to any limitation as to materiality or Parent Material Adverse Effect or any similar limitation set forth therein) as of the date of this Agreement and as of the Closing Date as though made on and as of such date and time (except to the extent that any such representation and warranty speaks of an earlier date, in which case such representation and warranty must be true and correct as of such earlier date), except for such inaccuracies that, individually or in the aggregate, have not had, and could not reasonably be expected to have, a Parent Material Adverse Effect. The Company must have received at the Closing a certificate signed on behalf of Parent by an executive officer of Parent to the effect that such executive officer has read this <u>Section 8.3(a)</u> and the conditions set forth in this <u>Section 8.3(a)</u> have been satisfied.

(b) <u>Performance of Obligations of Parent and Merger Sub</u>. Each of Parent and Merger Sub must have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and the Company must have received a certificate signed on behalf of Parent and Merger Sub by an executive officer of Parent to such effect.

# ARTICLE IX

#### **TERMINATION**

9.1 <u>Termination by Mutual Consent</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the approval by stockholders of the Company referred to in <u>Section 8.1(a)</u>, by mutual written consent of the Company by action of the Company Board and Parent.

9.2 <u>Termination by Either Parent or the Company</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the approval by stockholders of the Company referred to in <u>Section 8.1(a)</u>, by Parent or the Company by action of the Company Board and by written notice if:

(a) the Merger is not consummated by September 21, 2007 (as such date may be extended by mutual written consent of Parent and the Company) (the <u>Outside Date</u>);

(b) the approval of the Company s stockholders required by <u>Section 8.1(a)</u> is not obtained at the Company Meeting (after giving effect to any adjournment or postponement thereof if a vote on the approval of this Agreement and the Merger is taken at such Company Meeting or adjournment or postponement thereof); or

(c) any Order permanently restraining, enjoining or otherwise prohibiting consummation of the Merger becomes final and non-appealable, whether before or after the Company Meeting (provided such party used commercially reasonable efforts to have such Order lifted);

A-44

#### Table of Contents

<u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to clause (a) or (b) above will not be available to any party that has breached or failed to perform in any material respect its obligations under this Agreement in any manner that has been the principal cause of or primarily resulted in the failure of the Merger to be consummated; <u>provided</u>, <u>further</u>, that, prior to or upon any termination by the Company pursuant to clause (b) above, the Company must have paid to Parent any Termination Fee then due and payable under <u>Section 9.5</u> under the terms specified in <u>Section 9.5</u>.

9.3 <u>Termination by the Company</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the approval by stockholders of the Company referred to in <u>Section 8.1(a)</u>, by the Company by action of the Company Board:

(a) if prior to, but not after, the time the vote is taken with respect to the adoption of this Agreement at the Company Meeting, (i) the Company Board, pursuant to and in compliance with Section 7.2, has approved or recommended to the stockholders of the Company any Superior Proposal and (ii) prior to or upon termination pursuant to this Section 9.3(a), the Company has paid to Parent the Termination Fee then due and payable under Section 9.5; provided, however, that (A) prior to such termination pursuant to this <u>Section 9.3(a)</u>, the Company notified Parent in writing promptly of its intention to terminate this Agreement and to enter into a binding Alternative Acquisition Agreement concerning a Superior Proposal promptly following the Waiting Period (as defined below), attaching the most current version of such agreement (or, to the extent no such agreement is contemplated to be entered into by the Company in connection with such Superior Proposal, a description of all material terms and conditions of such Superior Proposal), and (B) Parent did not make, within three (3) Business Days after its receipt of such written notification (the <u>Waiting Period</u>), an offer that the Company Board determined, in good faith after consultation with its financial advisor, is at least as favorable to the stockholders of the Company from a financial point of view as such Superior Proposal (it being understood that (1) the Company shall not enter into any such binding agreement prior to or during the Waiting Period, (2) the Company shall keep Parent reasonably informed at all times during the Waiting Period of the status and material terms and conditions (including any amendment thereto) of such Superior Proposal and provide copies of all draft Alternative Acquisition Agreements relating to such Superior Proposal (and any executed confidentiality agreement entered into in the circumstances referred to in